SU(4) Flavorful Portals

George N. Wojcik and Thomas G. Rizzo[†]

SLAC National Accelerator Laboratory 2575 Sand Hill Rd., Menlo Park, CA, 94025 USA

Abstract

Dark Matter (DM) and its potential interactions with the Standard Model (SM) continue to present a rich framework for model building. In the case of thermal DM of a mass between a few MeV and a few GeV, a compelling and much-explored framework is that of a dark photon/vector portal, which posits a new U(1) "dark photon" that undergoes small kinetic mixing (KM) with the SM hypercharge. This mixing can be mediated at the one-loop level by portal matter (PM) fields which are charged under both the dark U(1) and the SM gauge group. From requirements for a lack of gauge anomalies and lifetime constraints from early-universe cosmology, it is favored that fermionic portal matter take the form of vector-like copies of SM particles, albeit non-trivially charged under the dark U(1), which have delicate cancellations of dark U(1) charges and SM hypercharges in order to yield a finite and calculable KM. The distinctive particle content of PM models then presents an intriguing framework with which one may postulate on the UV completions of a simplified model with the dark photon portal, including those which may embed the dark U(1) in a larger group structure. We construct a model in which the dark U(1) is extended to $SU(4)_F \times U(1)_F$, incorporating a local SU(3) flavor symmetry with PM appearing as a vector-like "fourth generation" to supplement the three generations of the SM. To ensure finite contributions to KM, the SM gauge group is arranged into Pati-Salam multiplets. The new extended dark gauge group presents a variety of interesting experimental signatures, including non-trivial consequences of the flavor symmetry being unified with the dark sector.

[†]gwojci03,rizzo@slac.stanford.edu

1 Introduction

In spite of composing roughly 80% of the matter in the universe, the precise identity of dark matter (DM) remains undetermined. Models to produce the appropriate abundance of DM, however, strongly suggest that it is subject to interactions other than gravity, and null results (thus far) in searches for historically favored DM candidates such as axions [1,2] and WIMPs [3] have led to a proliferation of new ideas on the identity, production mechanisms, and experimental signatures of DM [4, 5]. In this work, we will be exploring one of the more recently popular DM models, that of a kinetic mixing/vector portal [6,7]. In these setups, interaction between the Standard Model (SM) fields and the DM is mediated by a massive vector dark photon, A_D , associated with a broken dark gauge group $U(1)_D$. The DM particle is uncharged under the SM gauge group, but possesses non-zero $U(1)_D$ charge, while the SM fields are just the opposite: Charged under the SM group, but not under $U(1)_D$. Small couplings between the dark photon and the SM fields occur due to kinetic mixing (KM) between $U(1)_D$ and $U(1)_Y$, which after electroweak symmetry breaking ultimately give every SM particle a coupling to A_D of $e \epsilon Q$, where e is the electromagnetic coupling constant, Q is the electric charge of a given particle, and ϵ is a small value which parameterizes the strength of the kinetic mixing. While the the simplest realizations of these models, in which the only parameters are the DM and dark photon masses (as well as potentially the mass of a scalar associated with $U(1)_D$ breaking), the $U(1)_D$ gauge coupling, and the KM parameter ϵ have a rich phenomenology, they leave little framework for addressing a number of questions: What is the origin of the small KM parameter ϵ ? Is $U(1)_D$ the only dark sector gauge group, or is it embedded in some larger gauge symmetry? Could the dark sector gauge forces be connected in some way to other ongoing questions in particle physics?

In the region of parameter space where DM and the dark photon have masses ranging between $\sim a$ few MeV to $\sim a$ few GeV, the favored value of ϵ generally lies in the range of $\epsilon \sim 10^{-(3-5)}$ [8,9], which can in turn suggests that the KM can arise from a simple one-loop vacuum polarization-like diagram featuring "portal matter" fields charged under both $U(1)_Y$ and $U(1)_D$ [7]. The kinetic mixing parameter then has the dependence

$$\epsilon \propto \sum_{i} Q_{Y_i} Q_{D_i} \log\left(\frac{m_i^2}{\mu^2}\right),\tag{1}$$

where the sum over i denotes a sum over all fermions (a nearly identical expression arising from complex scalar loops also occurs, differing only in the omitted proportionality constant in front of the sum), Q_{Y_i} denotes the hypercharge of particle i, Q_{D_i} denotes its $U(1)_D$ charge, and μ is a renormalization scale. The sum in Eq.(1) is finite and calculable, then, when sum $\sum_i Q_{D_i} Q_{Y_i} = 0$. In two previous works by one of our authors [10, 11], henceforth referred to as I and II, respectively, the theory and phenomenology of these portal matter fields was explored. In I, it was argued that fermionic portal matter fields must be vector-like in order to avoid constraints due to gauge anomalies and precision electroweak measurements. Additionally, in order to ensure that the fields are unstable (and hence conform to cosmological constraints), they should only appear in the same representations under the SM gauge group as SM particles. In short, portal matter fermions should be vector-like copies of SM fields [12], a possibility rarely explored in the literature [13]. The discussion in I limited itself to the simplest possible constructions, in which a pair of such vector-like fermions, with opposite dark charges, generate finite KM through a small mass splitting between them. Although this setup provided for a rich phenomenology, the inclusion of portal matter fields which satisfied the condition $\sum_i Q_{D_i} Q_{Y_i} = 0$ was ultimately ad hoc; in spite of the critical importance that this cancellation had to the model, it did not happen "naturally". Furthermore, the $U(1)_D$ itself was still minimal – the potentially new effects arising from a non-minimal set of dark gauge symmetries in which $U(1)_D$ could be embedded, including the possibility that the SM fields were non-trivially charged under some part of the extended gauge group orthogonal to $U(1)_D$, was not explored. II began the process of addressing both of these questions. First, it was noted that the required cancellation to render Eq.(1) finite would naturally occur when the portal matter was placed in the vector-like representation $\mathbf{5} + \overline{\mathbf{5}}$ of SU(5), which would then be broken down to the SM gauge group. Inspired, then, by E_6 theories in which the 27 of E_6 breaks down to a $(\overline{5}, 2) + (1, 2) + (5, 1) + (10, 1)$ of $SU(5) \times SU(2)$ (allowing for the complete SM to be contained in a $\overline{5} + 10$, with a vector-like set of portal fields in $\overline{\mathbf{5}} + \mathbf{5}$), II then developed a model based on a gauge group $SU(5) \times SU(2)_I \times U(1)_{Y_I}$. This setup exhibited a number of intriguing properties, including the emergence of new heavy gauge bosons associated with the dark sector but coupled to the SM and non-standard dark photon couplings emerging from mass mixing with SM gauge bosons– the dark group $U(1)_D$ only emerges as a combination of $SU(2)_I$ and $U(1)_{Y_I}$ generators. The model in II also exhibited potentially interesting flavor physics behavior, including flavor-changing neutral currents and flavor-dependent couplings of new gauge bosons to SM fermions. However, the setup as written had comparatively little underlying theoretical structure for the shape and magnitudes that these effects might take.

We wish now to expand on the work of **II**, and in particular the flavor physics questions it raised, by considering the following: If the extended gauge group containing $U(1)_D$ can include groups under which the SM particles are non-trivial representations, might we create a model in which the dark photon gauge symmetry, $U(1)_D$, might be unified, either partially or completely, with some sort of flavor symmetry? With that goal in mind, we develop a specific model: First, we extend the SM to a Pati-Salam symmetry $SU(4)_c \times SU(2)_L \times SU(2)_R$ [14], which, as a semisimple group, we shall prove in Section 2.1 will guarantee that KM from contributions of the form of Eq.(1) will remain finite and calculable. For our "dark sector", in which the dark photon $U(1)_D$ shall be embedded, we choose the $SU(4)_F \times U(1)_F$ group. We shall see that this group is large enough to contain both a $U(1)_D$ under which the SM fields can remain uncharged, and an embedded $SU(3)_F$ group describing quark flavor. Multiplets containing the SM fields then are placed in fundamental (and antifundamental) representations of the $SU(4)_F$ group, with the three SM generations forming triplets (and antitriplets) under $SU(3)_F$ and portal matter fields representing a "fourth generation" that are singlets under $SU(3)_F$. From this outline, we create a phenomenologically realistic theory in which an $SU(3)_F$ flavor symmetry is partially unified with a vector portal for DM.

Our paper is laid out as follows: In Section 2, we discuss the field content of the model and the masses of any new exotic particles we require. In addition to explicitly computing the masses and mixings of all phenomenologically relevant new fermions and gauge bosons, we comment qualitatively on several important aspects of the scalar sector, since a rigorous treatment of this sector of the model is complex enough to lie beyond the scope of the present work. In Section 3, we establish several important phenomenological tools required for our later analysis, in particular, explicitly determining the couplings of the new gauge bosons, as well as the modifications of couplings of SM gauge bosons (and the Higgs), to all model fermions, and determining what a "natural" range for our model's parameters might be. In this Section, we also explicitly compute the magnitude of the KM effect, *i.e.*, ϵ in the model, and comment briefly on how it constrains our parameter space. In Section 4, we compute the phenomenologically significant flavor-changing neutral currents appearing in the model, and compare them with existing experimental constraints. In Section 5, we offer a brief survey of the phenomenology of the new fields that may be produced at collider experiments, in particular focusing on how our model here differs from the results seen in I and II. Finally, Section 6 summarizes our results and discusses possible avenues with which this work might be expounded upon.

2 Model Setup

2.1 Fermion Content

To begin our construction, it is useful to make several assumptions. First, to ensure that our fermions will always produce a finite and calculable kinetic mixing between SM hypercharge and any new U(1)'s appearing in our model, we will restrict our efforts to theories where the gauge group can be written $\mathcal{G} \times \mathcal{G}_P$, where \mathcal{G}_P is an arbitrary gauge group containing a dark U(1) symmetry, and \mathcal{G} is some *semisimple* group containing the SM. Our inclusion of the condition that \mathcal{G} is semisimple now guarantees that, once \mathcal{G} is broken down to a group which includes U(1) factors, any kinetic mixing between these U(1) factors and any U(1) contained in \mathcal{G}_P generated by one-loop contributions of the form of Eq.(1) will be finite. We can prove this straightforwardly: Take a field in the representation $\mathcal{R}_1 \times \mathcal{R}_2$ of $\mathcal{G} \times \mathcal{G}_P$. Then, consider an arbitrary U(1) embedded in \mathcal{G} , the generator for which we'll call T_Y , and an arbitrary U(1) embedded in \mathcal{G}_P , the generator for which we'll call T_P . Then, the contribution of $\mathcal{R}_1 \times \mathcal{R}_2$ to KM will be finite and calculable as long as the trace $Tr[T_YT_P] = 0$, where the generators T_Y and T_P are given for the representation $\mathcal{R}_1 \times \mathcal{R}_2$. However, since this representation is a direct product, this trace is simply $Tr[T_Y] \times Tr[T_P]$. For a semisimple group, the trace $Tr[T_Y]$ is always zero for any U(1) generator and any representation. As as a result, $Tr[T_YT_P]$ will always be zero, and the contribution of any representation under $\mathcal{G} \times \mathcal{G}_P$ to KM will always be finite and calculable, as long as \mathcal{G} is semisimple.¹

For our purposes, we shall select (as mentioned in the Introduction) the Pati-Salam gauge group $SU(4)_c \times SU(2)_L \times SU(2)_R$ as our semisimple group \mathcal{G} [14]. To ensure that any new exotic fermions occur in representations that will quickly decay into SM fields, we then assume that all fermions in our model occur in the traditional Pati-Salam multiplets (4, 2, 1) and $(\overline{4}, 1, 2)$, or their conjugate representations.² As our work above for a general \mathcal{G} would suggest, it is straightforward to see that each of these multiplets has $Tr[Y_{SM}] = 0$, and hence, if we only introduce new matter in these representations, then regardless of its transformation properties under \mathcal{G}_P , the new fermions will invariably produce a finite and calculable kinetic mixing between the SM hypercharge and any new $U(1) \subset \mathcal{G}_P$. Dark matter, if we wish it to be fermionic, can then be introduced as a Pati-Salam singlet (or collection of such singlets) or as the SM singlet component of a (4, 1, 2) multiplet, with some non-trivial charge under $U(1)_D$. For the sake of brevity, we shall employ some further simplifying notation: We shall denote the groups $SU(N)_A \times SU(M)_B \times ... \times U(1)_C$ by $N_A M_B ... 1_C$, so that, for example, $\mathcal{G}_{PS} = 4_c 2_L 2_R$, and the SM gauge group $SU(3)_c \times SU(2)_L \times U(1)_Y$ is simply written $3_c 2_L 1_Y$.

Our choice of \mathcal{G}_P is then further restricted by our requirement that the model incorporate three generations of the usual SM fermion fields, all of which must be uncharged under $U(1)_D$. A simple method of accomplishing this is to assume that \mathcal{G}_P contains an SU(3) flavor symmetry, $SU(3)_F$, the generators of which are orthogonal to the generators of the $U(1)_D$ symmetry. That is, we can decompose $\mathcal{G}_P \to SU(3)_F \times \mathcal{G}'_P$, with $\mathcal{G}'_P \supset U(1)_D$. Then, if we can find a model which gives the complete set of SM fermions in fundamental or antifundamental representations of $SU(3)_F$ with zero charge under $U(1)_D$, we'll have recreated all three generations of the SM. The sole remaining requirement is that all other additional fermions in the model be vector-like once \mathcal{G}'_P is broken down to $U(1)_D$, leaving the SM as the only 'light' chiral matter content.

Precisely this sort of construction can be arranged if we take, e.g., $\mathcal{G}_P = SU(4)_F \times U(1)_F = 4_F 1_F$, and write the matter content of our theory in terms of left-handed multiplets (not to be confused with the $SU(2)_L$ gauge group) of the form $4_c 2_L 2_R 4_F 1_F$ as

$$(\mathbf{4}, \mathbf{2}, \mathbf{1}, \mathbf{4}, -1/4) + (\bar{\mathbf{4}}, \mathbf{1}, \mathbf{2}, \bar{\mathbf{4}}, +1/4) + (\bar{\mathbf{4}}, \mathbf{2}, \mathbf{1}, \mathbf{1}, +1) + (\mathbf{4}, \mathbf{1}, \mathbf{2}, \mathbf{1}, -1).$$
(2)

By breaking $4_F 1_F \to 3_F 1_{F'} 1_F \to 3_F 1_D$, we can then decompose this field content into

$$(4, 2, 1, 3, 0) + (\overline{4}, 1, 2, \overline{3}, 0) + (4, 2, 1, 1, -1) + (\overline{4}, 2, 1, 1, +1) + (4, 1, 2, 1, -1) + (\overline{4}, 1, 2, 1, +1), \quad (3)$$

where multiplets are now labelled according to $4_c 2_L 2_R 3_F 1_D$. As is readily apparent from Eq.(3), all of our criteria are satisfied: The SM is contained in a chiral $SU(3)_F$ triplet and an antitriplet, which only gain mass at the scale of electroweak symmetry breaking, while all additional fields, which might serve as portal matter to contribute to kinetic mixing, are vector-like and therefore able to obtain far larger masses. Inspection of Eq.(2) also shows this model to be anomaly-free.

In order to effect realistic flavor mixing within this setup, we finally introduce additional vector-like multiplets to act as seesaw partners to the SM fermions, analogous to the treatment of the charged leptons and quarks in the model construction of [15]. Specifically, we find that adding a vector-like multiplet in

¹There are two caveats worth mentioning here: First, this works equally well assuming \mathcal{G}_P , rather than \mathcal{G} , is semisimple, however, given the large number of extensions of the SM gauge group to a semisimple one (Pati-Salam, trinification, and all GUT models), we have elected here to restrict our attention to the scenario where \mathcal{G} is semisimple. Second, we note that in the high-energy theory, the absence of an independent U(1) factor in \mathcal{G} (since U(1) is not simple) means that KM arising from vacuum polarization diagrams as described in the Introduction are forbidden by gauge invariance. However, we note that these terms still arise via effective operators featuring the insertion of scalar vev's when these gauge symmetries are broken– we shall argue in Section 3.3 that this does not vitiate our KM calculation there, at least for the purposes of evaluating the magnitude of the mixing.

 $^{^{2}}$ This is not strictly necessary, since there exist other Pati-Salam representations that contain only fields with SM-like quantum numbers. However, we shall stick to the familiar canonical choices for simplicity.

the $(4, 1, 2, 4, -1/4) + (\overline{4}, 1, 2, \overline{4}, +1/4)$ representation to the matter content of Eq.(2) allows us, with appropriate selections for the model's scalar content, to exhibit the observed mass hierarchy for quarks and charged leptons, as well as the SM CKM matrix. Our final matter content can then be written as

$$(4, 2, 1, 4, -1/4) + (\overline{4}, 1, 2, \overline{4}, +1/4) + (\overline{4}, 2, 1, 1, +1) + (4, 1, 2, 1, -1)$$
(4)
+ (4, 1, 2, 4, -1/4) + ($\overline{4}, 1, 2, \overline{4}, +1/4$).

For convenience, we have listed this matter content, labelled by Pati-Salam and SM multiplets, in Table 1. The addition of the new vector-like states in Eq.(4), however, raises two issues that warrant some discussion. First, we note that these extra states vitiate any discrete left-right symmetry that previously existed in the model, and which is frequently assumed in Pati-Salam constructions. Our selection here is ultimately one of convenience: The additional vector-like fields represent the *minimal* additions we need to effect the observed SM flavor structure, at least via the method outlined in Section 2.3. In principle, nothing prevents us from adding a corresponding vector-like $SU(2)_L$ multiplet to restore the left-right discrete symmetry. In practice, however, the addition of these states significantly complicates the numerical structure of the model, so for our purposes here we simply include the extra $SU(2)_R$ vectorlike multiplet. If we wish to maintain a left-right discrete symmetry, we can assume that the particles of the corresponding $SU(2)_L$ vector-like multiplet are significantly more massive than those of the $SU(2)_R$ multiplet, and therefore do not have a significant numerical effect on the theory at low energy. We also note that even if we were to include an $SU(2)_L$ multiplet with similar particle masses to those of its $SU(2)_R$ counterpart, the effect on low-energy phenomenology would be minimal; we shall see in Section 2.3 that the natural masses of the particles in the extra vector-like multiplets are so high that only one of these vector-like particles, acting as a seesaw partner to the top quark, will be accessible at the LHC or, indeed, any likely future planned colliders. We therefore can determine that the effect of adding an extra vector-like multiplet to the theory to restore left-right symmetry would, even if the masses of the particles in the multiplet were comparable, likely only result in the introduction of a single additional particle, a vector-like $SU(2)_L$ doublet quark, at accessible energies. Given the substantial increase in numerical complexity coupled with relatively limited phenomenological impact, we leave an exploration of the strict enforcement of discrete left-right symmetry in this model to future work.

We also note that the fermion content of Eq.(4) pushes us over the constraint for asymptotically free QCD: If the Pati-Salam $SU(4)_c$ symmetry is broken down to the SM QCD at a higher scale than any of these fermions acquire mass, then the theory has 18 flavors, while QCD only remains asymptotically free for 16 or fewer flavors. We might note that as long as we don't introduce any further colored fermions (as we would need to in order to, for example, enforce a discrete left-right symmetry), the matter content of Eq.4 should result in a confined $SU(4)_c$ theory (even accounting for the existence of an $SU(4)_c$ adjoint scalar to break this symmetry). However, given the fact that we shall assume the Pati-Salam symmetry, including $SU(4)_c$, will be broken at an exceedingly high scale $\sim O(10^{13})$ TeV (following setups such as [16–18]), it is quite likely irresponsible to assume that the $SU(3)_c$ theory won't reach its Landau pole before $SU(4)_c$ symmetry is restored and asymptotic freedom regained. It has been conjectured, however, that it's not unreasonable to abandon asymptotic freedom at some intermediate scales, such as the TeV-scale, in exchange for asymptotic safety, where a UV interacting fixed point is reinstated by physics at some high energy before the Planck scale [19]. As our discussion in this paper is limited to physics well below the Planck scale, we won't conjecture as to how this asymptotic freedom is restored, and only note that such a restoration may be possible.

2.2 Scalar Content

In order to break the gauge symmetry in our model from its original GUT-scale $4_c 2_L 2_R 4_F 1_F$ to the SM gauge group, and then down to $3_c 1_{\rm em}$, as well as generate the appropriate spectrum for SM fermions, we must posit the existence of a significant number of scalar fields. For simplicity, we shall assume that the Pati-Salam symmetries $SU(4)_c$ and $SU(2)_R$ are broken at some exceedingly high scale (or scales). We shall see that this is not in fact an unreasonable assumption – if we were to assume a discrete left-right symmetry, for example, renormalization group running of the $SU(2)_L$ and $SU(2)_R$ in a minimal

Fermion	$4_c 2_L 2_R 4_F 1_F$ Multiplet	SM Rep	$SU(3)_c$	Y/2	$SU(3)_F$	$U(1)_D$
		q_L	3	+1/6	3	0
π	(A 9 1 4 1 1 4)	l_L	1	-1/2	3	0
Ψ_a	(4, 2, 1, 4, -1/4)	Q_L	3	+1/6	1	-1
		L_L	1	-1/2	1	-1
Ψ.	$(\overline{4} \ 2 \ 1 \ 1 \ \pm 1)$	Q_R^c	$\overline{3}$	-1/6	1	+1
¥ b	(4, 2, 1, 1, 1)	L_R^c	1	+1/2	1	+1
		d_R^c	$\overline{3}$	+1/3	3	0
		u_R^c	$\overline{3}$	-2/3	3	0
		e_R^c	1	+1	$\overline{3}$	0
Ψ.	$(\overline{4} \ 1 \ 2 \ \overline{4} \ +1/4)$	$ u_R^c $	1	0	$\overline{3}$	0
± c		$(D_1)^c_R$	$\overline{3}$	+1/3	1	+1
		$(U_1)^c_R$	$\overline{3}$	-2/3	1	+1
		$(E_1)^c_R$	1	+1	1	+1
		$(N_1)^c_R$	1	0	1	+1
		$(D_1)_L$	3	-1/3	1	-1
Ψ.	$(4 \ 1 \ 2 \ 1 \ -1)$	$(U_1)_L$	3	+2/3	1	-1
Ψd	(4 , 1 , 2 , 1 , -1)	$(E_1)_L$	1	-1	1	-1
		$(N_1)_L$	1	0	1	-1
		$(D_2)_L$	3	-1/3	3	0
		$(U_2)_L$	3	+2/3	3	0
		$(E_2)_L$	1	-1	3	0
Ψ	$(A \ 1 \ 2 \ A \ -1/4)$	$(N_2)_L$	1	0	3	0
Ψ_e	(4, 1, 2, 4, -1/4)	$(D_3)_L$	3	-1/3	1	-1
		$(U_3)_L$	3	+2/3	1	-1
	$\begin{pmatrix} (C_3)_L \\ (E_3)_L \end{pmatrix} = \begin{bmatrix} 3 & -1 \\ 1 & -1 \end{bmatrix}$		1	-1		
		$(N_3)_L$	1	0	1	-1
		$(D_2)^c_R$	3	+1/3	3	0
		$(U_2)_R^c$	$\overline{3}$	-2/3	$\overline{3}$	0
<u>.</u>		$(E_2)_R^c$	1	+1	$\overline{3}$	0
	$(\overline{A} \ 1 \ 2 \ \overline{A} \ 1 \ 1 \ 2 \ \overline{A} \ 1 \$	$(N_2)_R^c$	1	0	$\overline{3}$	0
± <i>f</i>	$(\mathbf{T}, \mathbf{I}, \mathbf{\Delta}, \mathbf{T}, \mathbf{T})$	$(D_3)^c_R$	$\overline{3}$	+1/3	1	+1
		$(U_3)_R^c$	$\overline{3}$	-2/3	1	+1
		$(E_3)_R^c$	1	+1	1	+1
		$(N_3)^c_R$	1	0	1	+1

Table 1: The fermion content of the model, grouped by their representation under $4_c 2_L 2_R 4_F 1_F$

Pati-Salam model suggests symmetry breaking around ~ 10^{13-14} GeV [16–18], while the highest scales we shall require for breaking the $SU(4)_F \times U(1)_F$ symmetry and reproducing the fermion spectra and mixings will not exceed $O(10^{8-9} \text{ GeV})$. Assuming the Pati-Salam symmetry is broken, then, we are now left with the need for scalars which break the $SU(4)_F \times U(1)_F$ symmetry down to a dark U(1), which we shall call $U(1)_D$, which is then broken entirely at some low scale at roughly O(0.1 - 1 GeV), as well as some scalar content to perform the role of the SM Higgs, breaking $2_L 1_Y \rightarrow 1_{\text{em}}$. To accomplish these tasks, we posit 5 Higgs scalars, given in the $4_c 2_L 2_R 4_F 1_F$ representations as

$$\Phi_A \sim (\mathbf{1}, \mathbf{1}, \mathbf{1}, \mathbf{15}, 0),
\Phi_B \sim (\mathbf{1}, \mathbf{1}, \mathbf{3}, \mathbf{15}, 0),
\Phi_S \sim (\mathbf{1}, \mathbf{1}, \mathbf{1}, \mathbf{1}, 0),
\Phi_P \sim (\mathbf{1}, \mathbf{1}, \mathbf{1}, \mathbf{4}, +3/4),
H \sim (\mathbf{1}, \mathbf{2}, \mathbf{2}, \mathbf{1}, 0).$$
(5)

For convenience, we have listed these scalars, along with rough orders of magnitude for their vacuum expectation values necessary to achieve the observed SM quark mass spectrum (which we shall derive in Section 2.3), in Table 2. The scalar content that we have selected here can be easily deduced from phenomenological considerations: The scalars Φ_A , Φ_B , Φ_S , and H are all immediately required for a treatment of the quark masses analogous to that of [15], generalized to treat our larger model. To adapt the construction of that work to the present scenario, we require that the $SU(3)_F$ adjoints in that work are promoted to $SU(4)_F$ adjoints, that one of the two $SU(4)_F$ adjoints also be a triplet of $SU(2)_R$ in order to effect the large discrepancy between up-like and down-like quark masses (and the non-trivial CKM matrix) without relying on renormalization group evolution, and that the scalar H containing the SM Higgs field is promoted to a bidoublet under $SU(2)_L \times SU(2)_R$. It is useful to note that the scalar content of Eq.(5) does not contain any scalars that are charged under both $U(1)_Y$ and $U(1)_F$. Loop-level kinetic mixing between these two U(1) symmetries, then, such as that mediated by the fermion content in our model, does NOT occur in the scalar sector.

					-
Scalar	$SU(2)_L$	$SU(2)_R$	$SU(4)_F$	$U(1)_F$	$\langle \Phi angle$
Φ_A	1	1	15	0	$ \begin{pmatrix} \langle A \rangle & \vec{\gamma}_A \\ \vec{\gamma}_A^{\dagger} & -Tr[\langle A \rangle] \end{pmatrix} \sim \begin{pmatrix} 10^3 - 10^9 \text{ GeV} & 0.1 - 1 \text{ GeV} \\ 0.1 - 1 \text{ GeV} & 10^9 \text{ GeV} \end{pmatrix} $
Φ_B	1	3	15	0	$ \frac{\sigma_3}{2} \otimes \begin{pmatrix} \langle B \rangle & \vec{\gamma}_B \\ \vec{\gamma}_B^{\dagger} & -Tr[\langle B \rangle] \end{pmatrix} \sim \begin{pmatrix} 10^3 - 10^9 \text{ GeV} & 0.1 - 1 \text{ GeV} \\ 0.1 - 1 \text{ GeV} & 10^9 \text{ GeV} \end{pmatrix} $
Φ_P	1	1	4	+3/4	$(\vec{\gamma}_P, v_P) \sim (0.1 - 1 \text{ GeV}, \ 10^3 - 10^4 \text{ GeV})$
Φ_S	1	1	1	0	$v_S \sim 10^3 - 10^4 \text{ GeV}$
Н	2	2	1	0	$rac{v}{\sqrt{2}} egin{pmatrix} \coseta & 0 \\ 0 & \sineta \end{pmatrix}$

Table 2: The scalars introduced to break the $4_F 1_F$ flavor symmetry and provide masses to the fermions. Here, $\langle A \rangle$ and $\langle B \rangle$ are 3×3 Hermitian matrices, $\vec{\gamma}_{A,B,P}$ are 3-component complex vectors, and v_P , v_S , and v_{EW} are simply real vevs, with $v \approx 246$ GeV being the SM Higgs vev. In the definition of Φ_B 's vev, σ_3 denotes the third Pauli matrix.

It is helpful here to use the labelling of Table 2 to illustrate the rough pattern of symmetry breaking from the scale where the Pati-Salam group is broken down to the SM group $3_c 2_L 1_Y$ down to the scale where the dark photon gauge symmetry $U(1)_D$ is broken – or in other words, how the $SU(4)_F \times U(1)_F$ symmetry is broken.

$$SU(4)_F \times U(1)_F \xrightarrow{\langle A,B \rangle} U(1)'_F \times U(1)_F \xrightarrow{v_P} U(1)_D \xrightarrow{\vec{\gamma}_{A,B,P}}$$
 Nothing. (6)

In Eq.(6), we have simply listed how the various vev terms from Table 2 break the $SU(4)_F \times U(1)_F$ symmetry down, arranged by the rough scale at which each breaking occurs. Consulting Table 2, we can see that $SU(4)_F$ is broken by $\langle A \rangle$ and $\langle B \rangle$ down to $U(1)'_F$ at a series of scales spanning between $\sim 10^9~{\rm GeV}$ and $10^3~{\rm GeV}$ (in practice, we shall that this symmetry breaking usually completes at a substantially higher scale than 10^3 GeV), while v_P then breaks $U(1)'_F \times U(1)_F$ down to $U(1)_D$ at a scale of roughly 10^{3-4} GeV, and finally the small vev's $\gamma_{A,B,P}$ break $U(1)_D$ entirely at a scale of $\sim 0.1-1$ GeV. Notably, the vacuum expectation values of certain components of the scalar fields are exceedingly large (we remind the reader that even the largest scales here, $\sim 10^9$ GeV, are still well below what we can anticipate for the breaking of the symmetries $SU(4)_c$ and $SU(2)_R$, which as noted before we assume to be ~ 10^{13-14} GeV). As a result, we can anticipate (and, in Sections 2.3 and 2.4, prove) that much of the new physics which shall arise in this model appears at scales far in excess of what can be directly probed in the foreseeable future. The task left to us is therefore identifying the elements of our model construction that yield new physics at scales that we do expect to be probed in the near future. In the scalar sector itself, given the large number of multiplets and the resulting high degree of complexity for the scalar potential, performing this task in detail is far beyond the scope of the present work. We can, however, briefly comment on some of the scalars that will likely be relevant at accessible energies.

First, we note that if the left-right symmetry breaking scale is quite high (as we have already assumed), the SM Higgs field is well-approximated as the sole element of the bidoublet H that achieves a non-zero

vev, so that

$$H = \begin{pmatrix} \phi_1^0 & \phi_2^+ \\ \phi_1^- & -\phi_2^{0*} \end{pmatrix}, \quad \langle H \rangle = \frac{v}{\sqrt{2}} \begin{pmatrix} \cos \beta & 0 \\ 0 & \sin \beta \end{pmatrix}, \quad h_{SM} \approx \phi_1^0 \cos \beta + \phi_2^0 \sin \beta, \tag{7}$$

where h_{SM} denotes the real scalar identified with the SM Higgs boson, and $v \approx 246$ GeV is the SM Higgs vev. In immediate analogy with conventional left-right symmetric models, additional physical scalars associated with the bidoublet H will all have masses at approximately the scale of $SU(2)_R$ breaking [20–22], and hence given our assumptions be unobservable at present or immediately foreseeable experiments. Finally, we note that to ensure perturbativity for the Yukawa couplings of the heavy scalars arising from the bidoublet, again in analogy with well-known left-right symmetric model building principles, we must also require that $\tan \beta \leq 0.8$ [21].³

Apart from the SM Higgs, there is only one other scalar that we shall consider here. Specifically, given the exceptionally low scale at which the symmetry $U(1)_D$ is broken in comparison to the other scales of the theory, we anticipate that some combination of the scalars described in this section will emerge as a physical 'dark Higgs' boson of mass ~ $O(|\vec{\gamma}_{P,A,B}|)$, as the physical counterpart to the combination of Goldstone bosons which become the longitudinal component of A_D - such a scalar appears in, for example, the discussions of I and II. Because the precise structure of the scalar sector is too complex to be able to probe in detail in this work, we instead shall simply deduce the contribution of this scalar to relevant processes (in particular, in Sections 5.1 and 5.2 when discussing the decay processes of heavy exotic fermions) via the Equivalence Theorem [23] where applicable, and comment briefly upon the potential effect of these scalars where such a quantitative treatment fails.

2.3 Fermion Spectrum

In order to determine the structure of the model's scalar vevs, we turn to the CKM mixing matrix and the spectrum of SM quark masses for guidance. In order to recreate the observed flavor structure of the SM, we present a construction based heavily on that of [15]. In the unbroken theory, we shall write the Yukawa sector of the model as

$$\mathcal{L}_{Y} = -y_{H} \cos \alpha \Psi_{f}^{T} i \sigma_{2} H \Psi_{a} - y_{H} \sin \alpha \Psi_{f}^{T} i \sigma_{2} \tilde{H} \Psi_{a} - y_{P1} \Psi_{b}^{T} i \sigma_{2} \Phi_{P}^{\dagger} \Psi_{a} - y_{P2} \Psi_{c}^{T} i \sigma_{2} \Phi_{P} \Psi_{d}$$

$$- y_{A} \Psi_{f}^{T} i \sigma_{2} \Phi_{A} \Psi_{e} - y_{B} \Psi_{f}^{T} i \sigma_{2} \Phi_{B} \Psi_{e} - M \Psi_{f}^{T} i \sigma_{2} \Psi_{e} - y_{S} \Psi_{c}^{T} i \sigma_{2} \Phi_{S} \Psi_{e} + h.c.$$

$$(8)$$

Here, $H \equiv \sigma_2 H^* \sigma_2$, σ_2 is the second Pauli matrix, α is an arbitrary real angle between $-\pi$ and π , and M is an arbitrary mass term not forbidden by any symmetries. For simplicity, we assume that all Yukawa coupling parameters here are real, as would be the case in a model with no explicit CP violation. The expression in Eq.(8) is not, a priori, the most general set of Yukawa couplings that can be written with the scalar and fermion content of Tables 1 and 2; however, any additional terms can be easily eliminated by imposing a discrete Z_2 symmetry. Specifically, we outline our parity assignments for the fermions and scalars in Table 3.

Z_2 Parity	Fermions	Scalars
+	$\Psi_a, \Psi_b, \Psi_e, \Psi_f$	$H, \Phi_P, \Phi_A, \Phi_B$
-	Ψ_c, Ψ_d	Φ_S

Table 3:	The parity	of the	model's	fermion	and so	alar i	fields	under	$Z_2,$	in ord	ler to	ensure	that	Eq.(8)	is
the most	general set	of Yul	kawa cou	plings th	nat we	can v	vrite.								

Following the parity assignments of Table 3, Eq.(8) becomes the most general set of Yukawa couplings

³This specifically stems from the fact that the heavy doublet's Yukawa coupling to SM quarks is enhanced by $\sec(2\beta)$, so having $\tan \beta$ overly close to 1 will result in an unacceptably large magnitude of heavy scalar couplings to the third-generation quarks. In principle, one might accomodate this bound by having $\tan \beta \gtrsim 1.2$, but this is identical to simply choosing an appropriate $\tan \beta \lesssim 0.8$ and changing the Yukawa coupling parameters, in particular the angle α , in the Yukawa action given in Eq.(8).

that we can construct. We can now write the up- and down-like quark mass matrices explicitly by referencing Tables 1 and 2. The mass terms for the up- and down-like quarks take the form

$$\overline{\mathbf{U}}_{R}\mathcal{M}_{u}\mathbf{U}_{L},\tag{9}$$

$$\overline{\mathbf{D}}_{R}\mathcal{M}_{d}\mathbf{D}_{L},$$

where

$$\overline{\mathbf{U}}_{R} \equiv (\overline{u}_{R}, (\overline{U}_{2})_{R}, \overline{Q}_{R}^{u}, (\overline{U}_{1})_{R}, (\overline{U}_{3})_{R}),
\mathbf{U}_{L} \equiv (u_{L}, (U_{2})_{L}, Q_{L}^{u}, (U_{1})_{L}, (U_{3})_{L}),
\overline{\mathbf{D}}_{\mathbf{R}} \equiv (\overline{d}_{R}, (\overline{D}_{2})_{R}, \overline{Q}_{R}^{d}, (\overline{D}_{1})_{R}, (\overline{D}_{3})_{R}),
\mathbf{D}_{L} \equiv (d_{L}, (D_{2})_{L}, Q_{L}^{d}, (D_{1})_{L}, (D_{3})_{L}).$$
(10)

Here, $Q_{L,R}^{u(d)}$ denotes the up(down)-like component of the $SU(2)_L$ doublet $Q_{L,R}$. We also remind the reader that $u_{L,R}$ and $(U_2)_{L,R}$ are triplets of $SU(3)_F$, and therefore have three components in flavor space, in contrast to $Q_{L,R}^{u,d}$, $(U_1)_{L,R}$, and $(U_3)_{L,R}$, which each have only one. So, $\mathbf{U}_{L,R}$ and $\mathbf{D}_{L,R}$ are both 9-component vectors in flavor space, where we have grouped the $SU(3)_F$ triplets $u_{L,R}$ and $(U_2)_{L,R}$ and $(U_2)_{L,R}$ together in the first 6 components, leaving the last 3 components for the $SU(3)_F$ singlets. Notably, the $SU(3)_F$ triplets are all uncharged under $U(1)_D$, while the $SU(3)_F$ singlets all possess a common non-zero charge under this symmetry. The mass matrices $\mathcal{M}_{u,d}$ are then given by

$$\mathcal{M}_{u} = \begin{pmatrix} \mathbf{0}_{3\times3} & y_{S}v_{S}\mathbf{1}_{3\times3} & \mathbf{0}_{3\times1} & y_{P2}\vec{\gamma}_{P} & \mathbf{0}_{3\times1} \\ \frac{y_{u}v}{\sqrt{2}}\mathbf{1}_{3\times3} & \mathbf{M}_{u} + M\mathbf{1}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \vec{\gamma}_{u} \\ y_{P1}\vec{\gamma}_{P}^{\dagger} & \mathbf{0}_{1\times3} & y_{P1}v_{P} & \mathbf{0} & \mathbf{0} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & \mathbf{0} & y_{P2}v_{P} & y_{S}v_{S} \\ \mathbf{0}_{1\times3} & \vec{\gamma}_{u}^{\dagger} & \frac{y_{u}v}{\sqrt{2}} & \mathbf{0} & -Tr[\mathbf{M}_{u}] + M \end{pmatrix},$$

$$\mathcal{M}_{d} = \begin{pmatrix} \mathbf{0}_{3\times3} & y_{S}v_{S}\mathbf{1}_{3\times3} & \mathbf{0}_{3\times1} & y_{P2}\vec{\gamma}_{P} & \mathbf{0}_{3\times1} \\ \frac{y_{d}v}{\sqrt{2}}\mathbf{1}_{3\times3} & \mathbf{M}_{d} + M\mathbf{1}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \vec{\gamma}_{d} \\ y_{P1}\vec{\gamma}_{P}^{\dagger} & \mathbf{0}_{1\times3} & y_{P1}v_{P} & \mathbf{0} & \mathbf{0} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & \mathbf{0} & y_{P2}v_{P} & y_{S}v_{S} \\ \mathbf{0}_{1\times3} & \vec{\gamma}_{d}^{\dagger} & \frac{y_{d}v}{\sqrt{2}} & \mathbf{0} & -Tr[\mathbf{M}_{d}] + M \end{pmatrix}, \end{cases}$$

$$\mathbf{M}_{u} = \mathbf{u}_{u} \langle A \rangle + \mathbf{u}_{u} \langle B \rangle \quad \mathbf{M}_{u} = \mathbf{u}_{u} \langle A \rangle - \mathbf{u}_{u} \langle B \rangle$$

$$\begin{split} \mathbf{M}_{u} &\equiv y_{A} \langle A \rangle + y_{B} \langle B \rangle, \quad \mathbf{M}_{d} \equiv y_{A} \langle A \rangle - y_{B} \langle B \rangle, \\ \vec{\gamma}_{u} &\equiv y_{A} \vec{\gamma}_{A} + y_{B} \vec{\gamma}_{B}, \quad \vec{\gamma}_{d} \equiv y_{A} \vec{\gamma}_{A} - y_{B} \vec{\gamma}_{B}. \\ y_{u} &\equiv y_{H} \cos(\alpha - \beta), \quad y_{d} \equiv y_{H} \sin(\alpha + \beta), \end{split}$$

where we remind the reader that $v, \beta, v_S, v_P, \vec{\gamma}_{P,A,B}, \langle A \rangle$, and $\langle B \rangle$ are defined in Table 2, while all remaining parameters here appear in Eq.(8). In general, we can exploit $SU(4)_F$ gauge freedom to render \mathbf{M}_u or \mathbf{M}_d diagonal, and eliminate either $\vec{\gamma}_u$ (if we diagonalize \mathbf{M}_u) or $\vec{\gamma}_d$ (if we diagonalize \mathbf{M}_d). This corresponds to moving to a basis in which either the combination of Φ_A and Φ_B vev's which couple to the up-like or down-like quarks are diagonal. Although we shall find it most convenient to work in a basis in which \mathbf{M}_u is diagonal and $\vec{\gamma}_u = 0$, for now we will make no such choice, so that our analytical results for the up-like quarks will apply straightforwardly to the down-like sector as well. Continuing our analysis, the matrices \mathcal{M}_u and \mathcal{M}_d can be bi-diagonalized as usual to produce mass eigenstates. That is, there exist unitary matrices $\mathcal{U}_{L,R}^{u,d}$ such that

$$(\mathcal{U}_{L}^{u})^{\dagger}\mathcal{M}_{u}^{\dagger}\mathcal{M}_{u}\mathcal{U}_{L}^{u} = (\mathcal{U}_{R}^{u})^{\dagger}\mathcal{M}_{u}\mathcal{M}_{u}^{\dagger}\mathcal{U}_{R}^{u} = diag(m_{u1}^{2}, m_{u2}^{2}, m_{u3}^{2}, M_{u1}^{2}, M_{u2}^{2}, M_{u3}^{2}, (m_{P1}^{u})^{2}, (m_{P2}^{u})^{2}, (m_{P3}^{u})^{2}),$$
(12)

where the m_{ui} denotes the mass of the *i*th-generation up-like SM quark, M_{ui} is the mass of the (also

uncharged under $U(1)_D$ heavy vector-like partner to this quark, and $m_{P1,2,3}^u$ denote the masses of the three portal matter fields which are charged under $U(1)_D$. An analogous expression holds for \mathcal{M}_d . As an example, we shall now determine the matrices $\mathcal{U}_{L,R}^u$, with a completely analogous treatment holding for the determination of $\mathcal{U}_{L,R}^d$.

We start by determining \mathcal{U}_L^u . To begin, we find it useful to define

$$\mathbf{u} \equiv \mathbf{M}_u + M \mathbf{1}_{3 \times 3}, \quad \mathbf{u}_D = diag(u_1, u_2, u_3) = \mathbf{W}_u^{\dagger} \mathbf{u} \mathbf{W}_u, \quad X_u \equiv -Tr[\mathbf{M}_u] + M, \tag{13}$$

where \mathbf{u}_D is the diagonalized form of \mathbf{u} and \mathbf{W}_u is a 3 × 3 unitary matrix. Then, we may write (dropping terms of $O(\vec{\gamma}_{P,u}^2)$, which shall be significantly smaller than the other mass scales occurring in the matrix)

$$\mathcal{M}_{u}^{\dagger}\mathcal{M}_{u} = \begin{pmatrix} \frac{y_{u}^{2}v_{u}^{2}}{2}\mathbf{1}_{3\times3} & \frac{y_{u}v}{\sqrt{2}}\mathbf{u} & y_{P1}^{2}v_{P}\vec{\gamma}_{P} & \mathbf{0}_{3\times1} & \frac{y_{u}v}{\sqrt{2}}\vec{\gamma}_{u} \\ \frac{y_{u}v}{\sqrt{2}}\mathbf{u} & \mathbf{u}^{2} + y_{S}^{2}v_{S}^{2}\mathbf{1}_{3\times3} & \frac{y_{u}v}{\sqrt{2}}\vec{\gamma}_{u} & y_{P2}y_{S}v_{S}\vec{\gamma}_{P} & (\mathbf{u} + X_{u}\mathbf{1}_{3\times3})\cdot\vec{\gamma}_{u} \\ y_{P1}^{2}v_{P}\vec{\gamma}_{P}^{\dagger} & \frac{y_{u}v}{\sqrt{2}}\vec{\gamma}_{u}^{\dagger} & y_{P1}^{2}v_{P}^{2} + \frac{y_{u}^{2}v^{2}}{2} & \mathbf{0} & \frac{y_{u}v}{\sqrt{2}}X_{u} \\ \mathbf{0}_{1\times3} & y_{P2}y_{S}v_{S}\vec{\gamma}_{P}^{\dagger} & \mathbf{0} & y_{P2}^{2}v_{P}^{2} & y_{P2}v_{P}y_{S}v_{S} \\ \frac{y_{u}v}{\sqrt{2}}\vec{\gamma}_{u}^{\dagger} & \vec{\gamma}_{u}^{\dagger} \cdot (\mathbf{u} + X_{u}\mathbf{1}_{3\times3}) & \frac{y_{u}v}{\sqrt{2}}X_{u} & y_{P2}v_{P}y_{S}v_{S} & y_{S}^{2}v_{S}^{2} + X_{u}^{2} \end{pmatrix} \right).$$
(14)

To derive expressions for the mass eigenvectors here, it is easiest to split the problem into two basic parts. First, we shall determine the unitary matrix $(\mathcal{U}_L^u)_0$ which diagonalizes the matrix of Eq.(14) in the limit where all components of $\vec{\gamma}_{P,u}$ are zero, that is, in the absence of all mixing between the fields with no charge under $U(1)_D$ (namely, u_L , and $(U_2)_L$, the first six rows/columns of Eq.(14)) and those with nonvanishing $U(1)_D$ charge $(Q_L^u, (U_1)_L)$, and $(U_3)_L$, the last three rows/columns). With this diagonalization done, we can then determine the effects of $\vec{\gamma}_{P,u}$ as small perturbations, in a rotation matrix $(\mathcal{U}_L^u)_{\gamma}$. The matrix which diagonalizes Eq.(14) is then simply given as

$$\mathcal{U}_L^u \approx (\mathcal{U}_L^u)_0 (\mathcal{U}_L^u)_\gamma. \tag{15}$$

We now begin with our determination of $(\mathcal{U}_L^u)_0$. Since there's no mixing between the $U(1)_D$ -charged and $U(1)_D$ -uncharged fermions in this limit, we shall treat the two separately. We start with the mass matrix for the fermions with no $U(1)_D$ charge, which takes the form,

$$\begin{pmatrix} \frac{y_u^2 v^2}{2} \mathbf{1}_{3 \times 3} & \frac{y_u v}{\sqrt{2}} \mathbf{u} \\ \frac{y_u v}{\sqrt{2}} \mathbf{u} & \mathbf{u}^2 + y_S^2 v_S^2 \end{pmatrix}.$$
 (16)

We can break this matrix down into three 2×2 matrices via rotation by the unitary block matrix

$$\begin{pmatrix} \mathbf{1}_{3\times3} & \mathbf{0}_{3\times3} \\ \mathbf{0}_{3\times3} & \mathbf{W}_u \end{pmatrix},\tag{17}$$

where we recall that $\mathbf{W}_{u}^{\dagger}\mathbf{u}\mathbf{W}_{u} = diag(u_{1}, u_{2}, u_{3})$. Then, each 2 × 2 matrix can be easily diagonalized, yielding (assuming that $v \ll v_{S}, u_{i}$)

$$diag(m_{ui}^{2}, M_{ui}^{2}) = \begin{pmatrix} \cos(\rho_{i}^{u}) & \sin(\rho_{i}^{u}) \\ -\sin(\rho_{i}^{u}) & \cos(\rho_{i}^{u}) \end{pmatrix} \begin{pmatrix} \frac{y_{u}^{2}v^{2}}{2} & \frac{y_{u}v}{\sqrt{2}}u_{i} \\ \frac{y_{u}v}{\sqrt{2}}u_{i} & u_{i}^{2} + y_{S}^{2}v_{S}^{2} \end{pmatrix} \begin{pmatrix} \cos(\rho_{i}^{u}) & -\sin(\rho_{i}^{u}) \\ \sin(\rho_{i}^{u}) & \cos(\rho_{i}^{u}) \end{pmatrix}$$
$$\cos(\rho_{i}^{u}) \approx 1 - \frac{1}{2}\frac{m_{ui}^{2}}{M_{ui}^{2}} \left(\frac{M_{ui}^{2} - y_{S}^{2}v_{S}^{2}}{y_{S}^{2}v_{S}^{2}}\right), \quad \sin(\rho_{i}^{u}) \approx -sgn(y_{u}u_{i})\frac{m_{ui}}{M_{ui}} \left(\frac{M_{ui}^{2} - y_{S}^{2}v_{S}^{2}}{y_{S}^{2}v_{S}^{2}}\right)^{1/2}, \quad (18)$$
$$m_{ui}^{2} \approx \frac{y_{u}^{2}v^{2}}{2}\frac{y_{S}^{2}v_{S}^{2}}{(u_{i}^{2} + y_{S}^{2}v_{S}^{2})}, \quad M_{ui}^{2} \approx u_{i}^{2} + y_{S}^{2}v_{S}^{2}$$

Having diagonalized the mass matrix for the left-handed quarks with no charge under $U(1)_D$, we next turn our attention to the left-handed fields which are charged under $U(1)_D$. In the case of the up-like quark sector, these are Q_L^u , $(U_1)_L$, and $(U_3)_L$. From Eq.(14), we see that the mass matrix for these fields is

$$\begin{pmatrix} y_{P1}^2 v_P^2 + \frac{y_u^2 v^2}{2} & 0 & \frac{y_u v}{\sqrt{2}} X_u \\ 0 & y_{P2}^2 v_P^2 & (y_S v_S)(y_{P2} v_P) \\ \frac{y_u v}{\sqrt{2}} X_u & (y_S v_S)(y_{P2} v_P) & y_S^2 v_S^2 + X_u^2 \end{pmatrix}.$$
 (19)

In the absence of extreme fine-tuning, we shall see that natural Yukawa couplings yield X_u at roughly the scale $X_u \sim 10^5 y_S v_S$, so we can assume that $v_S, v_P \ll X_u$. Any mixing between the three fields of Eq.(19), then, is exceedingly small and may be treated perturbatively. Dropping terms of $O(v_P^2/X_u^2, v_S^2/X_u^2, v_Z^2/X_u^2)$, the mass matrix of Eq.(19) can be diagonalized by a rotation by the matrix

$$\mathbf{W}_{L}^{P} \equiv \begin{pmatrix} 1 & \alpha_{12}^{L} & \alpha_{13}^{L} \\ -\alpha_{12}^{L} & 1 & \alpha_{23}^{L} \\ -\alpha_{13}^{L} & -\alpha_{23}^{L} & 1 \end{pmatrix},$$
(20)

where

$$\alpha_{12}^{L} \equiv \frac{(y_{u}v)(y_{S}v_{S})(y_{P2}v_{P})}{\sqrt{2}X_{u}(y_{P1}^{2} - y_{P2}^{2})v_{P}^{2}}, \quad \alpha_{13}^{L} \equiv \frac{y_{u}v}{\sqrt{2}X_{u}}, \quad \alpha_{23}^{L} \equiv \frac{(y_{S}v_{S})(y_{P2}v_{P})}{X_{u}^{2}}.$$
(21)

In the end, we arrive at

$$diag((m_{P1}^{u})^{2}, (m_{P2}^{u})^{2}, (m_{P3}^{u})^{2}) = \mathbf{W}_{L}^{P\dagger} \begin{pmatrix} y_{P1}^{2}v_{P}^{2} + \frac{y_{u}^{2}v^{2}}{2} & 0 & \frac{y_{u}v}{\sqrt{2}}X_{u} \\ 0 & y_{P2}^{2}v_{P}^{2} & (y_{S}v_{S})(y_{P2}v_{P}) \\ \frac{y_{u}v}{\sqrt{2}}X_{u} & (y_{S}v_{S})(y_{P2}v_{P}) & y_{S}^{2}v_{S}^{2} + X_{u}^{2} \end{pmatrix} \mathbf{W}_{L}^{P}, \quad (22)$$
$$(m_{P1}^{u})^{2} \equiv y_{P1}^{2}v_{P}^{2}, \quad (m_{P2}^{u})^{2} \equiv y_{P2}^{2}v_{P}^{2}, \quad (m_{P3}^{u})^{2} \equiv X_{u}^{2}.$$

We can now apply the combined rotations of Eqs.(17), (18), and (20) in order to arrive at the full rotation matrix $(\mathcal{U}_L^u)_0$. The combined rotation matrix is

$$(\mathcal{U}_{L}^{u})_{0} \equiv \begin{pmatrix} \mathbf{W}_{u}\mathbf{c}_{\rho}^{u} & -\mathbf{W}_{u}\mathbf{s}_{\rho}^{u} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{W}_{u}\mathbf{s}_{\rho}^{u} & \mathbf{W}_{u}\mathbf{c}_{\rho}^{u} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & 1 & \alpha_{12}^{L} & \alpha_{13}^{L} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & -\alpha_{12}^{L} & 1 & \alpha_{23}^{L} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & -\alpha_{13}^{L} & -\alpha_{23}^{L} & 1 \end{pmatrix},$$
(23)
$$\mathbf{c}_{\rho}^{u} \equiv diag(\cos(\rho_{1}^{u}), \cos(\rho_{2}^{u}), \cos(\rho_{3}^{u})), \quad \mathbf{s}_{\rho}^{u} \equiv diag(\sin(\rho_{1}^{u}), \sin(\rho_{2}^{u}), \sin(\rho_{3}^{u})),$$

where we remind the reader that \mathbf{W}_u is defined in Eq.(13), ρ_1^u , ρ_2^u , and ρ_3^u are given in Eq.(18), and the variables α_{ij}^L are defined in Eq.(21). The mass matrix of Eq.(14) then takes the form

$$(\mathcal{U}_L^u)_0^{\dagger} \mathcal{M}_u^{\dagger} \mathcal{M}_u (\mathcal{U}_L^u)_0 = diag(m_u^2, m_c^2, m_t^2, M_u^2, M_c^2, M_t^2, (m_{P1}^u)^2, (m_{P2}^u)^2, (m_{P3}^u)^2) + O(\vec{\gamma}_u, \vec{\gamma}_P).$$
(24)

It is now straightforward to determine the perturbed eigenvectors of the matrix in Eq.(24) up to $O(\vec{\gamma}_u, \vec{\gamma}_P)$. Once the dust settles, we have

$$(\mathcal{U}_{L}^{u})_{\gamma} \approx \begin{pmatrix} \mathbf{1}_{3\times3} & \mathbf{0}_{3\times3} & \vec{\Delta}_{P1L}^{u} & \vec{\Delta}_{P2L}^{u} & \vec{\Delta}_{P3L}^{u} \\ \mathbf{0}_{3\times3} & \mathbf{1}_{3\times3} & \vec{\Delta}_{P1L}^{U} & \vec{\Delta}_{P2L}^{U} & \vec{\Delta}_{P3L}^{U} \\ -\vec{\Delta}_{P1L}^{u\dagger} & -\vec{\Delta}_{P1L}^{U\dagger} & 1 & 0 & 0 \\ -\vec{\Delta}_{P2L}^{u\dagger} & -\vec{\Delta}_{P2L}^{U\dagger} & 0 & 1 & 0 \\ -\vec{\Delta}_{P3L}^{u\dagger} & -\vec{\Delta}_{P3L}^{U\dagger} & 0 & 0 & 1 \end{pmatrix},$$
(25)

where

$$\begin{split} (\vec{\Delta}_{P1L}^{u})_{i} &= \frac{((y_{P1}^{2}v_{P}\mathbf{c}_{\rho}^{u} - \alpha_{12}^{L}y_{P2}y_{S}v_{S}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\frac{y_{u}v}{\sqrt{2}}\mathbf{s}_{\rho}^{u} - \alpha_{13}^{L}(\frac{y_{u}v}{\sqrt{2}}\mathbf{c}_{\rho}^{u} + \mathbf{u}_{D}\mathbf{s}_{\rho}^{u} + X_{u}\mathbf{s}_{\rho}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u})_{i}}{m_{P1}^{2} - m_{ui}^{2}}, \\ (\vec{\Delta}_{P2L}^{u})_{i} &= \frac{((\alpha_{12}^{L}y_{P1}^{2}v_{P}\mathbf{c}_{\rho}^{u} + y_{P2}y_{S}v_{S}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\alpha_{12}^{L}\frac{y_{u}v}{\sqrt{2}}\mathbf{s}_{\rho}^{u} - \alpha_{23}^{L}(\frac{y_{u}v}{\sqrt{2}}\mathbf{c}_{\rho}^{u} + \mathbf{u}_{D}\mathbf{s}_{\rho}^{u} + X_{u}\mathbf{s}_{\rho}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u})_{i}}{m_{P2}^{2} - m_{ui}^{2}}, \\ (\vec{\Delta}_{P3L}^{u})_{i} &= \frac{((\alpha_{13}^{L}y_{P1}^{2}v_{P}\mathbf{c}_{\rho}^{u} + \alpha_{23}^{L}y_{P2}y_{S}v_{S}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\alpha_{13}^{L}\frac{y_{u}v}{\sqrt{2}}\mathbf{s}_{\rho}^{u} + \frac{y_{u}v}{\sqrt{2}}\mathbf{c}_{\rho}^{u} + \mathbf{u}_{D}\mathbf{s}_{\rho}^{u} + X_{u}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u})_{i}}{m_{P3}^{2} - m_{ui}^{2}}, \\ (\vec{\Delta}_{P3L}^{U})_{i} &= \frac{((-y_{P1}^{2}v_{P}\mathbf{s}_{\rho}^{u} - \alpha_{12}^{L}y_{P2}y_{S}v_{S}\mathbf{c}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\frac{y_{u}v}{\sqrt{2}}\mathbf{c}_{\rho}^{u} - \alpha_{13}^{L}(\mathbf{u}_{D}\mathbf{c}_{\rho}^{u} + X_{u}\mathbf{c}_{\rho}^{u} - \frac{y_{uv}}{\sqrt{2}}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u})_{i}}{(m_{P1}^{u})^{2} - M_{ui}^{2}}, \\ (\vec{\Delta}_{P3L}^{U})_{i} &= \frac{((-\alpha_{12}^{L}y_{P1}^{2}v_{P}\mathbf{s}_{\rho}^{u} + y_{P2}y_{S}v_{S}\mathbf{c}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\alpha_{13}^{L}\frac{y_{uv}v}{\sqrt{2}}\mathbf{c}_{\rho}^{u} - \alpha_{23}^{L}(\mathbf{u}_{D}\mathbf{c}_{\rho}^{u} + X_{u}\mathbf{c}_{\rho}^{u} - \frac{y_{uv}v}{\sqrt{2}}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u})_{i}}{(m_{P2}^{u})^{2} - M_{ui}^{2}}, \\ (\vec{\Delta}_{P3L}^{u})_{i} &= \frac{((-\alpha_{13}^{L}y_{P1}^{2}v_{P}\mathbf{s}_{\rho}^{u} + \alpha_{23}^{L}y_{P2}y_{S}v_{S}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\alpha_{13}^{L}\frac{y_{uv}v}{\sqrt{2}}\mathbf{c}_{\rho}^{u} - \alpha_{23}^{L}(\mathbf{u}_{D}\mathbf{c}_{\rho}^{u} + X_{u}\mathbf{c}_{\rho}^{u} - \frac{y_{uv}v}{\sqrt{2}}\mathbf{s}_{\rho}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u})_{i}}{(m_{P2}^{u})^{2} - M_{ui}^{2}}, \end{cases}$$

where \mathbf{u}_D is the diagonalized form of \mathbf{u} discussed in Eq.(13).

The lengthy expressions of Eq.(26), while extremely cumbersome in their full form, can be dramatically simplified by examining the numerical hierarchies between various parameters, and dropping all but the numerically dominant contributions to each expression. To get a better sense of the numerics of the system, we begin by exploring the $SU(3)_F$ triplet sector, containing the SM quarks and their heavy partners. Examining Eq.(18), we note that the expressions for m_{ui} and M_{ui} only hold here up to $O(v^2/v_S^2)$. To maintain numerical accuracy, we then assume that $y_S v_S \gtrsim 1$ TeV– given that $v \sim 246$ GeV, this will ensure that the expressions given here are accurate to within a few percent. This also ensures that there is a significant hierarchy between m_{ui} and M_{ui} , as can be seen from the relation $m_{ui}^2/M_{ui}^2 =$ $2m_{ui}^4/(y_S^2 v_S^2 y_u^2 v^2)$ (which holds exactly in the limit that $\vec{\gamma}_{u,P} \to 0$). If $y_S v_S$ were much smaller than 1 TeV, the heavy counterpart of the top quark would have roughly the same mass as the SM top, and be subject to extremely severe constraints from direct production searches at the LHC and from modifications to top quark couplings; the former limit the mass of such a vector-like quark to $\gtrsim 1.3 - 1.5$ TeV, due to null results in searches for pair production [24]. We can also use the relation $M_{ui}^2 = (y_S v_S)^2 (y_u v)^2 / (2m_{ui}^2)$, and its trivially analogous expression for down-like quarks, to derive approximate estimates of the masses of the various heavy quark partners.

Assuming that we want to reproduce the SM quark masses at the scale of ~ 1 TeV, roughly the scale at which the lightest heavy partner quark, the top partner, will be integrated out⁴, we arrive at

$$\begin{split} m_u(1 \ {\rm TeV}) &= 1.07 \ {\rm MeV}, & M_u \sim (2 \times 10^5) y_S v_S, \\ m_c(1 \ {\rm TeV}) &= 532 \ {\rm MeV}, & M_c \sim 300 y_S v_S, \\ m_t(1 \ {\rm TeV}) &= 144 \ {\rm GeV}, & M_t \sim y_S v_S, \\ m_d(1 \ {\rm TeV}) &= 2.28 \ {\rm MeV}, & M_d \sim (8 \times 10^4) y_S v_S, \\ m_s(1 \ {\rm TeV}) &= 46.5 \ {\rm MeV}, & M_s \sim (4 \times 10^4) y_S v_S, \\ m_b(1 \ {\rm TeV}) &= 2.40 \ {\rm GeV}, & M_b \sim 70 y_S v_S. \end{split}$$

Given these estimates, and assuming $y_S v_S \gtrsim 1$ TeV, only the heavy partner to the top quark, with a mass M_t , can reasonably be expected to be experimentally directly observable in the foreseeable future. We do, however, note that Eq.(27) assumes that the up-like and down-like Yukawa couplings to the SM

⁴A technically more correct procedure (albeit one that still ignores the running of the CKM matrix) would be for each SM quark's mass to be determined near the scale of the mass of its specific partner, so the model would reproduce the mass m_u run up to ~ 10⁹ GeV, m_c near ~ 10⁶ GeV, m_t near 1 TeV, etc. However, due to the additional quarks in the model, this computation would be substantially more complicated than our treatment here, and the numerical changes to the results would be minimal, especially given the fact that the most pronounced discrepancies with our calculation would arise in the least certain quark masses, m_u and m_d .

Higgs, that is, $y_H \cos(\alpha - \beta)$ and $y_H \sin(\alpha + \beta)$ are both roughly of O(1). While this may be expected from naturalness, it is far from the only feasible scenario. If, for example, $y_H \sin(\alpha + \beta) \sim O(10^{-2})$, equivalent to a value arising in a number of multi-Higgs doublet scenarios which seek to explain the mass hierarchy between the top and *b* quark, we might expect that $M_b \sim M_t$, and hence the heavy partner of the *b* may also play a significant phenomenological role. In this case, we would expect significant additional constraints arising from both direct production of the *b* partner at the LHC and possibly significant modifications to the SM $Z\bar{b}b$ coupling. For the sake of simplicity, however, we leave a detailed exploration of the effects of a lighter *b* partner within this model to future work.

Meanwhile, in the mass matrix of Eq.(19), we note that $X_u = -Tr[\mathbf{M}_u] + M$ should, in the absence of extreme fine tuning, be roughly the same order of magnitude as the largest eigenvalue of the matrix \mathbf{u} . We have established, however, that this eigenvalue is approximately $\sim 10^5 \times y_S v_S$. The mass $m_{P3}^u \approx |X_u|$ will naturally be extremely large, on the order of $\sim 10^8$ GeV.

Our expressions in Eq.(26) can then be dramatically simplified by simply assuming that the extremely heavy particles (the fermion with mass $\approx |X_u|$ and the heavy partners of the up and charm quarks) are virtually entirely decoupled from the lower mass fields (the SM quarks, the fermions with masses $\sim v_P$, and the heavy partner to the top quark) – in practice, this involves taking the limits where $|u_{1,2}|, |X_u|, M_{u1,u2} \rightarrow \infty$ in the expressions of Eq.(26). This permits us to write

$$\begin{split} \vec{\Delta}_{P1L}^{u} &\approx \frac{\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P}}{v_{P}}, \\ (\vec{\Delta}_{P2L}^{u})_{3} &\approx -sgn(y_{u}y_{s}y_{P2}u_{3}) \frac{m_{t}}{M_{t}} \frac{(M_{t}^{2} - y_{S}^{2}v_{S}^{2})^{1/2}}{m_{P2}^{u}} \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{3}}{v_{P}} \\ (\vec{\Delta}_{P1L}^{U})_{3} &\approx sgn(y_{u}u_{3}) \frac{m_{t}}{M_{t}} \left(\frac{M_{t}^{2} - y_{S}^{2}v_{S}^{2}}{y_{S}^{2}v_{S}^{2}} \right)^{1/2} \frac{(m_{P1}^{u})^{2}}{((m_{P1})^{2} - M_{t}^{2})} \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{3}}{v_{P}} \\ (\vec{\Delta}_{P2L}^{U})_{3} &\approx sgn(y_{P2}) \frac{y_{S}v_{S}m_{P2}^{u}}{(m_{P2}^{u})^{2} - M_{t}^{2}} \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{3}}{v_{P}}, \end{split}$$

$$\end{split}$$

where the rest of the $\vec{\Delta}$ terms are either negligibly small or parameterize mixing between the extremely heavy states. The expressions in Eq.(28) already provide some useful information: In particular, in the limit where the super-heavy states decouple the mixing of the phenomenologically accessible states is independent of $\vec{\gamma}_u$, the parameter corresponding to the $U(1)_D$ -breaking components of the scalar vev's $\langle \Phi_A \rangle$ and $\langle \Phi_B \rangle$. Additionally, there exists a suppression factor of m_t/M_t in front of $(\vec{\Delta}_{P2L}^u)_3$ and $(\vec{\Delta}_{P1L}^U)_3$, corresponding to the mixing between the top quark and the m_{P2}^u state, and the mixing between the heavy top partner and the m_{P1}^u state, respectively – these couplings between $SU(3)_F$ triplet and singlet states only arise due to mixing between the top quark and its heavy partner, and therefore vanish in the limit where the top partner decouples from the SM.

Armed with Eq.(28), it is not difficult to extract a complete expression for \mathcal{U}_L^u , or at least the components of this rotation matrix that are relevant for the phenomenologically accessible particles. Taking the same decoupling limit in the expression for the diagonalization matrix $(\mathcal{U}_L^u)_0$ in Eq.(23) (which here corresponds to letting $\rho_{1,2}^u, \alpha_{ij}^L \to 0$), we can use Eqs.(15), (23), (25), and (28) in order to extract the approximate mass eigenvectors for each left-handed up-like (and, through trivial generalization, downlike) quark with a phenomenologically accessible mass. Truncating our 9-dimensional flavor-space quark vectors to omit the super-heavy particles, we can write these vectors as

$$(\overline{\mathbf{U}}_{R})_{\text{dec}} \equiv (\overline{u}_{R}, \overline{T}_{R}, \overline{Q}_{R}^{u}, (\overline{U}_{1})_{R}),$$

$$(\mathbf{U}_{L})_{\text{dec}} \equiv (u_{L}, T_{L}, Q_{L}^{u}, (U_{1})_{L}),$$

$$(\overline{\mathbf{D}}_{R})_{\text{dec}} \equiv (\overline{d}_{R}, \overline{Q}_{R}^{d}, (\overline{D}_{1})_{R}),$$

$$(\mathbf{D}_{L})_{\text{dec}} \equiv (d_{L}, Q_{L}^{d}, (D_{1})_{L}).$$

$$(29)$$

Notably, the above truncation in the up-like sector, in which the top quark's heavy partner is retained, is much easier in a basis in which $\mathbf{W}_u = \mathbf{1}_{3\times 3}$. Otherwise, as we can see from Eq.(10), the top partner

will be represented by a non-trivial combination of the elements of the $SU(3)_F$ triplets $(U_2)_L$ and $(U_2)_R$, which would require retaining rows and columns of $\mathbf{U}_{L,R}^u$ that correspond to heavy states until a rotation into a basis in which $\mathbf{W}_u = \mathbf{1}_{3\times 3}$ could be performed. Therefore, for the following expressions, we have assumed that $\mathbf{W}_u = \mathbf{1}_{3\times 3}$; as we mentioned at the beginning of this section, it is always possible to use $SU(4)_F$ freedom to work in such a basis from the beginning of our analysis. Using Eqs.(23) and (25), we can write the truncated up-like quark rotation matrix as

$$\mathcal{U}_{L}^{u} \approx \begin{pmatrix} 1 & 0 & 0 & 0 & \frac{(\vec{\gamma}_{P})_{1}}{v_{P}} & 0 \\ 0 & 1 & 0 & 0 & \frac{(\vec{\gamma}_{P})_{2}}{v_{P}} & 0 \\ 0 & 0 & 1 & r\frac{m_{t}}{M_{t}} & \frac{(\vec{\gamma}_{P})_{3}}{v_{P}} & rq\left(\frac{z_{P2}^{3}}{1-z_{P2}^{2}}\right)\frac{m_{t}}{M_{t}}\frac{(\vec{\gamma}_{P})_{3}}{v_{P}} \\ 0 & 0 & -r\frac{m_{t}}{M_{t}} & 1 & r\left(\frac{z_{P1}^{2}}{1-z_{P1}^{2}}\right)\frac{m_{t}}{M_{t}}\frac{(\vec{\gamma}_{P})_{3}}{v_{P}} & q\left(\frac{z_{P2}}{1-z_{P2}^{2}}\right)\frac{(\vec{\gamma}_{P})_{3}}{v_{P}} \\ -\frac{(\vec{\gamma}_{P})_{1}^{*}}{v_{P}} & -\frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & -r\left(\frac{1}{1-z_{P1}^{2}}\right)\frac{m_{t}}{M_{t}}\frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & 1 & 0 \\ 0 & 0 & rqz_{P2}\frac{m_{t}}{M_{t}}\frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & -\frac{qz_{P2}}{1-z_{P2}^{2}}\frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & 0 & 1 \end{pmatrix}$$

$$(30)$$

$$r \equiv sgn(y_u u_3) \left(\frac{M_t^2 - y_S^2 v_S^2}{y_S^2 v_S^2}\right)^{1/2}, \ q \equiv \frac{y_S v_S}{M_t}, \ z_{P1} \equiv sgn(y_{P2}) \frac{M_t}{m_{P1}^u}, \ z_{P2} \equiv sgn(y_{P2}) \frac{M_t}{m_{P2}^u},$$

where we have dropped terms of $O(m_t^2/M_t^2)$ and higher. Note that each of the variables r, q, z_{P1} , and z_{P2} are all O(1) parameters, if we assume that the portal scale v_P is close to the heavy top partner mass scale v_S . The result for the down-like quarks is dramatically simpler, since we assume that the heavy b quark partner, unlike the top partner, is too massive to be phenomenologically relevant. In this case, we simply have

$$\mathcal{U}_{L}^{d} \approx \begin{pmatrix} \mathbf{W}_{d} & \frac{\vec{\gamma}_{P}}{v_{P}} & \mathbf{0}_{3\times 1} \\ -\frac{\vec{\gamma}_{P}^{\dagger} \mathbf{W}_{d}}{v_{P}} & 1 & 0 \\ \mathbf{0}_{1\times 3} & 0 & 1 \end{pmatrix}.$$
 (31)

Having determined the relevant components of the left-handed rotation matrix \mathcal{U}_L^u , we can follow an analogous procedure for the right-handed rotation matrix \mathcal{U}_R^u identified in Eq.(12). For the sake of brevity, we shall simply summarize the results here. The right-handed fermion mass matrix is given (up to $O(\vec{\gamma}_{P,u})$) by

$$\mathcal{M}_{u}\mathcal{M}_{u}^{\dagger} = \begin{pmatrix} y_{S}^{2}v_{S}^{2}\mathbf{1}_{3\times3} & y_{S}v_{S}\mathbf{u} & \mathbf{0}_{3\times1} & y_{P2}^{2}v_{P}\vec{\gamma}_{P} & y_{S}v_{S}\vec{\gamma}_{u} \\ y_{S}v_{S}\mathbf{u} & \frac{y_{u}^{2}v^{2}}{2} + \mathbf{u}^{2} & \frac{y_{u}v}{\sqrt{2}}y_{P1}\vec{\gamma}_{P} & y_{S}v_{S}\vec{\gamma}_{u} & (\mathbf{u} + X_{u}\mathbf{1}_{3\times3})\vec{\gamma}_{u} \\ \mathbf{0}_{1\times3} & \frac{y_{u}v}{\sqrt{2}}y_{P1}\vec{\gamma}_{P}^{\dagger} & y_{P1}^{2}v_{P}^{2} & \mathbf{0} & \frac{y_{u}v}{\sqrt{2}}y_{P1}v_{P} \\ y_{P2}^{2}v_{P}\vec{\gamma}_{P}^{\dagger} & y_{S}v_{S}\vec{\gamma}_{u}^{\dagger} & \mathbf{0} & y_{P2}^{2}v_{P}^{2} + y_{S}^{2}v_{S}^{2} & X_{u}y_{S}v_{S} \\ y_{S}v_{S}\vec{\gamma}_{u}^{\dagger} & \vec{\gamma}_{u}^{\dagger}(\mathbf{u} + X_{u}\mathbf{1}_{3\times3}) & \frac{y_{u}v}{\sqrt{2}}y_{P1}v_{P} & X_{u}y_{S}v_{S} & \frac{y_{u}^{2}v^{2}}{2} + X_{u}^{2} \end{pmatrix}.$$
(32)

In the same framework as our treatment of \mathcal{U}_L^u , we shall split the result into a matrix $(\mathcal{U}_R^u)_0$ which diagonalizes $\mathcal{M}_u \mathcal{M}_u^{\dagger}$ in the limit where $\vec{\gamma}_{P,u} \to 0$, and a matrix $(\mathcal{U}_R^u)_{\gamma}$ which provides the leading-order

corrections due to the $\vec{\gamma}_{P,u}$ terms. For $(\mathcal{U}_R^u)_0$, we arrive at

$$(\mathcal{U}_{R}^{u})_{0} = \begin{pmatrix} -\mathbf{W}_{u}\mathbf{c}_{\eta}^{u}sgn(y_{S}y_{u}\mathbf{u}_{D}) & -\mathbf{W}_{u}\mathbf{s}_{\eta}^{u}sgn(\mathbf{u}_{D}) & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ -\mathbf{W}_{u}\mathbf{s}_{\eta}^{u}sgn(y_{S}y_{u}\mathbf{u}_{D}) & \mathbf{W}_{u}\mathbf{c}_{\eta}^{u}sgn(\mathbf{u}_{D}) & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & sgn(y_{P1}) & \alpha_{12}^{R}sgn(y_{P2}) & \alpha_{13}^{R}sgn(X_{u}) \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & -\alpha_{12}^{R}sgn(y_{P1}) & sgn(y_{P2}) & \alpha_{23}^{R}sgn(X_{u}) \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & -\alpha_{13}^{R}sgn(y_{P1}) & -\alpha_{23}^{R}sgn(y_{P2}) & sgn(X_{u}) \end{pmatrix},$$

$$(33)$$

$$\begin{split} \mathbf{c}_{\eta}^{u} &\equiv diag(\cos(\eta_{1}^{u}), \cos(\eta_{2}^{u}), \cos(\eta_{3}^{u})), \quad \mathbf{s}_{\eta}^{u} \equiv diag(\sin(\eta_{1}^{u}), \sin(\eta_{2}^{u}), \sin(\eta_{3}^{u})), \\ \cos(\eta_{i}^{u}) &= \left(\frac{M_{ui}^{2} - y_{S}^{2}v_{S}^{2}}{M_{ui}^{2}}\right)^{1/2}, \quad \sin(\eta_{i}^{u}) = -\frac{y_{S}v_{S}}{M_{ui}}sgn(u_{i}), \\ \alpha_{12}^{R} &\equiv \frac{y_{u}v}{\sqrt{2}X_{u}}\frac{(y_{S}v_{S})(y_{P1}v_{P})}{(y_{P1}^{2} - y_{P2}^{2})v_{P}^{2}}, \quad \alpha_{13}^{R} \equiv \frac{y_{u}v}{\sqrt{2}X_{u}}\frac{y_{P1}v_{P}}{X_{u}}\left(1 - \frac{y_{S}^{2}v_{S}^{2}}{(y_{P1}^{2} - y_{P2}^{2})v_{P}^{2}}\right), \quad \alpha_{2}3^{R} \equiv \frac{y_{S}v_{S}}{X_{u}}. \end{split}$$

The sole non-trivial difference in the result of Eq.(33) and that of Eq.(23) is the introduction of sign flips (or rather, phase rotations) of the right-handed quarks in order to ensure that the mass terms in the final action correspond to real positive fermion masses– these rotations could have just as well been done in the left-handed sector, without altering the physical results of the model. The leading perturbations to $(\mathcal{U}_R^u)_0$ arising from the $\vec{\gamma}_{P,u}$ terms of the mass matrix of Eq.(32) are then given by

$$(\mathcal{U}_{R}^{u})_{\gamma} = \begin{pmatrix} \mathbf{1}_{3\times3} & \mathbf{0}_{3\times3} & \vec{\Delta}_{P1R}^{u} & \vec{\Delta}_{P2R}^{u} & \vec{\Delta}_{P3R}^{u} \\ \mathbf{0}_{3\times3} & \mathbf{1}_{3\times3} & \vec{\Delta}_{P1R}^{U} & \vec{\Delta}_{P2R}^{U} & \vec{\Delta}_{P3R}^{U} \\ -\vec{\Delta}_{P1R}^{u\dagger} & -\vec{\Delta}_{P1R}^{U\dagger} & \mathbf{1} & \mathbf{0} & \mathbf{0} \\ -\vec{\Delta}_{P2R}^{u\dagger} & -\vec{\Delta}_{P2R}^{U\dagger} & \mathbf{0} & \mathbf{1} & \mathbf{0} \\ -\vec{\Delta}_{P3R}^{u\dagger} & -\vec{\Delta}_{P3R}^{U\dagger} & \mathbf{0} & \mathbf{0} & \mathbf{1} \end{pmatrix},$$
(34)

where

$$\begin{split} (\vec{\Delta}_{P1R}^{u})_{i} &= \frac{-\left(\frac{(\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{s}_{\eta}^{u} - \alpha_{12}^{R}y_{P2}^{2}v_{P}\mathbf{c}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} - (\alpha_{12}^{R}y_{S}v_{S}\mathbf{s}_{\eta}^{u} + \alpha_{13}^{R}(y_{S}v_{S}\mathbf{c}_{\eta}^{u} + \mathbf{u}_{D}\mathbf{s}_{\eta}^{u} + X_{u}\mathbf{s}_{\eta}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u}}{((m_{P1}^{u})^{2} - m_{ui}^{2})sgn(y_{P1}y_{S}y_{u}u_{i})} \\ (\vec{\Delta}_{P2R}^{u})_{i} &= \frac{-\left((\alpha_{12}^{R}\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{s}_{\eta}^{u} + y_{P2}^{2}v_{P}\mathbf{c}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (y_{S}v_{S}\mathbf{s}_{\eta}^{u} - \alpha_{23}^{R}(y_{S}v_{S}\mathbf{c}_{\eta}^{u} + \mathbf{u}_{D}\mathbf{s}_{\eta}^{u} + X_{u}\mathbf{s}_{\eta}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u}}{((m_{P2}^{u})^{2} - (m_{ui}^{2}))sgn(y_{P2}y_{S}y_{u}u_{i})} \\ (\vec{\Delta}_{P3R}^{u})_{i} &= \frac{-\left((\alpha_{13}^{R}\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{s}_{\eta}^{u} + \alpha_{23}^{R}y_{P2}^{2}v_{P}\mathbf{c}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u} + (\alpha_{23}^{R}y_{S}v_{S}\mathbf{s}_{\eta}^{u} + y_{S}v_{S}\mathbf{s}_{\eta}^{u} + \mathbf{u}_{D}\mathbf{s}_{\eta}^{u} + X_{u}\mathbf{s}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u}}{((m_{P2}^{u})^{2} - m_{ui}^{2})sgn(X_{u}y_{S}y_{u}u_{i})} \\ (\vec{\Delta}_{P3R}^{U})_{i} &= \frac{\left((\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{c}_{\eta}^{u} + \alpha_{12}^{R}y_{P2}^{2}v_{P}\mathbf{s}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} - (\alpha_{12}^{R}y_{S}v_{S}\mathbf{s}_{\eta}^{u} + \alpha_{13}^{R}(-y_{S}v_{S}\mathbf{s}_{\eta}^{u} + \mathbf{u}_{D}\mathbf{c}_{\eta}^{u} + X_{u}\mathbf{c}_{\eta}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u}}\right)_{i}}{((m_{P1}^{u})^{2} - M_{ui}^{2})sgn(y_{P1}u_{i})} \\ (\vec{\Delta}_{P2R}^{U})_{i} &= \frac{\left((\alpha_{12}^{R}\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{c}_{\eta}^{u} - y_{P2}^{2}v_{P}\mathbf{s}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (y_{S}v_{S}\mathbf{c}_{\eta}^{u} - \alpha_{23}^{R}(-y_{S}v_{S}\mathbf{s}_{\eta}^{u} + \mathbf{u}_{D}\mathbf{c}_{\eta}^{u} + X_{u}\mathbf{c}_{\eta}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u}}\right)_{i}}{((m_{P2}^{u})^{2} - M_{ui}^{2})sgn(y_{P2}u_{i})} \\ (\vec{\Delta}_{P3R}^{U})_{i} &= \frac{\left((\alpha_{13}^{R}\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{c}_{\eta}^{u} - \alpha_{23}^{R}y_{P2}^{2}v_{P}\mathbf{s}_{\eta}^{u})\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{P} + (\alpha_{23}^{R}y_{S}v_{S}\mathbf{s}_{\eta}^{u} - \alpha_{23}^{R}(-y_{S}v_{S}\mathbf{s}_{\eta}^{u} + \mathbf{u}_{D}\mathbf{c}_{\eta}^{u} + X_{u}\mathbf{c}_{\eta}^{u}))\mathbf{W}_{u}^{\dagger}\vec{\gamma}_{u}}\right)_{i}}{((m_{P3}^{u})^{2} - M_{ui}^{2})sgn(y_{P2}u_{i})} \\ (\vec{\Delta}_{P3R}^{U})_{i} &= \frac{\left((\alpha_{13}^{R}\frac{y_{u}v}{\sqrt{2}}y_{P1}\mathbf{s}_{\eta}^{u} - \alpha_{12}^{R}y_{P2}^{2}v_{P}\mathbf{s}_{\eta}^{u})\mathbf{W$$

In the same manner as the left-handed eigenvectors, the expressions in Eq.(35) simplify dramatically in the limit where the m_{P3}^u quark and the heavy up and charm partners decouple from the theory. In this limit, we arrive at

$$\begin{split} (\vec{\Delta}_{P1R}^{u})_{3} &\approx \frac{m_{t}}{m_{P1}^{u}} \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{3}}{v_{P}}, \\ (\vec{\Delta}_{P2R}^{u})_{i} &\approx -sgn(y_{u}y_{S}u_{i}y_{P2}) \left(\frac{M_{ui}^{2} - y_{S}^{2}v_{S}^{2}}{M_{ui}^{2}}\right)^{1/2} \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{i}}{v_{P}}, \\ (\vec{\Delta}_{P1R}^{U})_{3} &\approx sgn(y_{u}u_{3}) \frac{m_{P1}^{u}m_{t}}{(m_{P1}^{u})^{2} - M_{t}^{2}} \left(\frac{M_{t}^{2} - y_{S}^{2}v_{S}^{2}}{y_{S}^{2}v_{S}^{2}}\right)^{1/2} \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{3}}{v_{P}}, \\ (\vec{\Delta}_{P2R}^{U})_{3} &\approx sgn(y_{P2}) \frac{(m_{P2}^{u})^{2}}{(m_{P2}^{u})^{2} - M_{t}^{2}} \left(\frac{y_{S}v_{S}}{M_{t}}\right) \frac{(\mathbf{W}_{u}^{\dagger} \vec{\gamma}_{P})_{3}}{v_{P}}, \end{split}$$
(36)

with all other $\vec{\Delta}$ terms for the right-handed quarks being either numerically negligible or parameterizing mixing between the extremely heavy fermions. We can then insert the results of Eq.(36) into Eq.(34) in order to derive the rotation matrix $(\mathcal{U}_R^u)_{\gamma}$, which when combined with $(\mathcal{U}_R^u)_0$ given in Eq.(33) will give us the mass eigenstates for the right-handed up-like quarks in the model. As in the case of the left-handed diagonalization matrices, we shall restrict our attentions to the limit in which only the SM quarks, the heavy top partner, and the two lighter portal matter fields mix among one another, with the other much more massive additional fermions decoupled from the low-energy theory. Using the truncated flavor-space vectors of Eq.(29), we can write

$$\mathcal{U}_{R}^{u} \approx \mathbf{N}_{1} \begin{pmatrix} 1 & 0 & 0 & 0 & 0 & \frac{(\vec{\gamma}_{P})_{1}}{v_{P}} \\ 0 & 1 & 0 & 0 & 0 & \frac{(\vec{\gamma}_{P})_{2}}{v_{P}} \\ 0 & 0 & qr & q & rq\left(\frac{z_{P1}^{3}}{1-z_{P1}^{2}}\right) \frac{m_{t}}{M_{t}} \frac{(\vec{\gamma}_{P})_{3}}{v_{P}} & \left(\frac{1-q^{2}r^{2}z_{P2}^{2}}{1-z_{P2}^{2}}\right) \frac{(\vec{\gamma}_{P})_{3}}{v_{P}} \\ 0 & 0 & -q & qr & \frac{z_{P1}}{q} \left(\frac{1-q^{2}z_{P1}^{2}}{1-z_{P1}^{2}}\right) \frac{m_{t}}{M_{t}} \frac{(\vec{\gamma}_{P})_{3}}{v_{P}} & r\frac{q^{2}z_{P2}^{2}}{1-z_{P2}^{2}} \frac{(\vec{\gamma}_{P})_{3}}{v_{P}} \\ 0 & 0 & z_{P1} \frac{m_{t}}{M_{t}} \frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & -r\left(\frac{z_{P1}}{1-z_{P1}^{2}}\right) \frac{m_{t}}{M_{t}} \frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & 1 & 0 \\ -\frac{(\vec{\gamma}_{P})_{1}^{*}}{v_{P}} & -\frac{(\vec{\gamma}_{P})_{2}^{*}}{v_{P}} & -qr\frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & -\frac{q}{1-z_{P2}^{2}} \frac{(\vec{\gamma}_{P})_{3}^{*}}{v_{P}} & 0 & 1 \end{pmatrix} \mathbf{N}_{2}$$

$$(37)$$

 $\mathbf{N_1} \equiv diag(1, 1, 1, sgn(y_u y_S), 1, 1), \ \mathbf{N_2} \equiv diag(-sgn(u_1 y_S y_u), -sgn(u_2 y_S y_u), -1, 1, sgn(y_{P1}), sgn(y_{P2})), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_{P2})), sgn(y_{P2}), sgn(y_{P2})), sgn(y_$

up to $O(m_t^2/M_t^2)$ corrections, where r, q, z_{P1} , and z_{P2} are all defined as in Eq.(30). Considering the case of the down-like quarks, where just as we did for \mathcal{U}_L^d we have assumed that the heavy b partner has decoupled from the theory, we arrive at

$$\mathcal{U}_{R}^{d} \approx \begin{pmatrix} \mathbf{W}_{d} & \mathbf{0}_{3\times 1} & \frac{\vec{\gamma}_{P}}{v_{P}} \\ \mathbf{0}_{1\times 3} & 1 & 0 \\ -\frac{\vec{\gamma}_{P}^{\dagger}\mathbf{W}_{d}}{v_{P}} & 0 & 1 \end{pmatrix} \begin{pmatrix} -sgn(y_{S}y_{d}\mathbf{d}_{D}) & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{0}_{1\times 3} & sgn(y_{P1}) & 0 \\ \mathbf{0}_{1\times 3} & 0 & sgn(y_{P2}) \end{pmatrix}.$$
(38)

At this point, we have determined the rotation matrices \mathcal{U}_L^u and \mathcal{U}_R^u necessary to diagonalize the up-like quark mass matrix, and by trivial generalization, these results will also give us the rotation matrices $\mathcal{U}_{L,R}^d$ needed to diagonalize the down-like quark mass matrices. With this knowledge, we can now comment briefly on how to ensure that this construction reproduces the observed CKM matrix. Considering Eq.(23) and its down-like equivalent, we can see that in the limit where $\vec{\gamma}_{P,u,d} \to 0$, the CKM matrix (as defined by the coupling matrix to the SM W boson, which couples to the left-handed SM quarks but not their heavy partners) is approximately given by

$$V_{CKM} = \mathbf{c}^{u}_{\rho} \mathbf{W}^{\dagger}_{u} \mathbf{W}_{d} \mathbf{c}^{d}_{\rho}, \tag{39}$$

where we remind the reader that \mathbf{W}_u and \mathbf{W}_d are the unitary matrices which diagonalize the Hermitian matrices **u** and **d** (as defined in Eq.(13) for **u**, with an analogous expression for **d**) respectively. Since $\mathbf{c}_{\rho}^u \approx$

 $\mathbf{c}_{\rho}^{d} \approx 1$ up to $O(m_{q}^{2}/M_{q}^{2})$ corrections (where q = u, d, s, c, b, t), the CKM matrix can be well-approximated as $\mathbf{W}_{u}^{\dagger}\mathbf{W}_{d}$, or in other words, by specifying **u** and **d** such that the clash of their diagonalization matrices matches the observed CKM.

Our recipe for reconstructing the quark sector in this construction is therefore compete, and it is useful at this point to take stock of the parameters which we may freely specify while still producing SM quark masses and mixings consistent with observation. First, we recall that there are two $SU(4)_F$ adjoint scalars in the model, Φ_A and Φ_B , with vev's that we can write $\langle \Phi_A \rangle$ and $\sigma_2/2 \otimes \langle \Phi_B \rangle$. Then, we can use SU(4) freedom to work in a basis in which the combination $y_A \langle \Phi_A \rangle + y_B \langle \Phi_B \rangle$ is diagonal, which we can see from the action in Eq.(8) results in $\gamma_u = 0$ and $\mathbf{u} = \mathbf{u}_D = diag(u_1, u_2, u_3)$. Then, we have $V_{CKM} \approx \mathbf{W}_d$, that is, the rotation matrix to diagonalize **d** is uniquely determined by the need to recreate the CKM matrix (\mathbf{W}_u is, of course, simply the identity matrix in this basis). Turning to quark masses, we note that if we specify $y_u \equiv y_H \cos(\alpha - \beta) \sim O(1)$, $y_d \equiv y_H \sin(\alpha + \beta) \sim O(1)$, and $y_S v_S \sim 1$ TeV, Eqs.(18) and (33) uniquely determine the eigenvalues of \mathbf{u} and \mathbf{d} up to a sign, and therefore also give us the masses of the heavy partner quarks and the mixing angles $\rho_{1,2,3}^{u,d}$ and $\eta_{1,2,3}^{u,d}$. If we additionally specify $y_{P1} \sim 1$, $y_{P2} \sim 1$, $v_P \gtrsim 1$ TeV, and the mass term M (as seen in Eqs.(8), (11), and (13)), we can then specify the masses and mixings of the portal matter fermions (that is, those that consist primarily of fields with non-zero charge under $U(1)_D$). Notably, in our present construction, the masses of the some of the portal matter fermions are degenerate between the up- and down-quark sectors (specifically, $m_{P1}^u = m_{P1}^d$ and $m_{P2}^u = m_{P2}^d$) up to radiative corrections. In practice, we might expect such a relation to not hold precisely, due to differing renormalization group evolution of the Yukawa couplings $y_{P1,2}$ to the up- and down-like sectors, but for the purposes of our simple numerical study we shall take this relation at face value. The sole remaining parameters we may specify are the $U(1)_D$ -breaking terms, in form of the complex vectors $\vec{\gamma}_{P,d} \lesssim 1$ GeV (recalling that in our basis, $\vec{\gamma}_u = 0$), which in turn give us the effects of mixing between states of different $U(1)_D$ charge, as parameterized in Eqs. (26) and (35).

In summary, then, we can generate a point in the model parameter space that produces the observed quark masses and mixings by specifying the O(1) real parameters y_H , y_{P1} , y_{P2} , as well as $y_S v_S$, the angles α and β , the real vev's v_P , $v_S \gtrsim 1$ TeV, and the complex $U(1)_D$ -breaking vev's $\vec{\gamma}_P$ and $\vec{\gamma}_d$. Altogether, these selections generate a unique point in parameter space up to the signs of the eigenvalues of the matrices **u** and **d**.

Having addressed the quark sector, we move on to discussing the lepton sector of the theory. Notably, because none of the scalars introduced thus far break the group $SU(4)_c$, the Yukawa couplings in Eq.(8) do not account for any discrepancy between down-like quark and charged lepton masses at tree level. However, because we've already assumed that the scale for $SU(4)_c$ breaking, which would distinguish between quark and lepton mixing, is incredibly high (and indeed must be, due to constraints on processes mediated by new $SU(4)_c$ gauge bosons), we can assume that discrepancies between the charged lepton and down-like quark masses are due to significant renormalization group effects, in particular differing running of the couplings y_A , y_B , and y_H , as well as the mass term M, for color triplets and singlets. In practice, the task of generating a specific numerical realization of the leptonic Yukawa couplings necessary to reproduce the observed charged lepton spectrum consistent with the vev structure necessary to reproduce the SM quark masses and CKM matrix, is analytically onerous and not terribly enlightening. However, we can expect that the effect of the specific realization of the charged lepton sector is largely phenomenologically irrelevant: Because the spectrum of charged leptons resembles that of the down-like quarks (at least in order of magnitude), it suffices here to point out that such a realization is almost certainly achievable with choices of leptonic y_A , y_B , y_H , and M of roughly the same order of magnitude as those for the down-like quark sector, as would be consistent with RG evolution. Phenomenologically, then, we would expect that the $SU(3)_F$ heavy partner leptons here are, like their counterparts in the down-like quark sector, too heavy to be realistically producible in existing and currently planned collider experiments. The only potentially accessible new fermions arising from the charged leptons, then, would be the $SU(3)_F$ singlet portal matter fields, specifically those which should have masses of O(TeV). However, due to their lack of color charge, experimental constraints on their production at the LHC are somewhat weaker than those of the quark portal matter fields with comparable masses, so we shall not address their phenomenology in detail here.

The neutrino sector within this model meanwhile presents substantial additional challenges. In par-

ticular, the tiny neutrino masses $\sim 0.1 \text{ eV}$ and near-maximal neutrino flavor mixings are in general inconsistent with any sort of high-energy degeneracy with the up-like quark sector broken by renormalization group running, as the Yukawa couplings in Eq.(8) suggest. There do, however, exist extensions of the model that can account for this discrepancy. For example, including scalars in the representations (15, 1, 1, 15, 0), (15, 1, 3, 15, 0), (15, 1, 1, 1, 0) and/or (15, 1, 3, 1, 0) could allow for dramatically different Yukawa couplings and mixings between the quark and lepton sectors, at the expense of a vet larger scalar sector and extreme fine-tuning to reproduce the tiny neutrino masses. Alternatively, the introduction of a scalar in the representation (10, 1, 3, 10, -1/2) with the appropriate vev would permit the inclusion of a large Majorana mass for the right-handed $SU(3)_F$ triplet neutrinos, which would in turn suppress the SM neutrino masses via a seesaw mechanism. In both of these scenarios, the extension of the scalar sector would also result in substantial additional contributions to the masses of the new $SU(3)_F$ gauge bosons- given the fact that the SM neutrino masses are so small, it's even probable that given these setups the vev's of some of these scalars would dominate the mass terms of the gauge bosons. Given the model-building ambiguity here and the substantial complexity that such considerations would introduce, however, we determine that a full exploration of potential ways to realize the neutrino mass spectrum and mixing within this framework is beyond the scope of our present work. In our discussion of the gauge boson masses in Section 2.4, however, we shall note that even in the absence of any additional mass terms which might arise from added scalar content to the model, the new gauge bosons associated with the $SU(3)_F$ symmetry will be extremely heavy, with only one of these bosons even approaching a low enough mass scale to have a potentially observable effect even in highly constrained measurements of flavor-changing neutral currents. As a result, we can expect that any modifications to the model setup that must be made to accommodate the neutrino masses and mixings should have a minimal effect on the experimentally observable new physics in the model considered here.

2.4 Gauge Boson Spectrum

Having selected our scalar vacuum expectation values such that the SM quark masses and mixings can be faithfully recreated, we now turn to how these selections will generate masses for the new gauge bosons in the theory. As noted in Section 2.2, we shall assume for the sake of simplicity that the Pati-Salam symmetry group, $SU(4)_c \times SU(2)_L \times SU(2)_R$, is broken down to the SM gauge group $SU(3)_c \times$ $SU(2)_L \times U(1)_Y$ at a significantly higher energy scale than any of the symmetry breakings of the group $SU(4)_F \times U(1)_F$. While this is certainly feasible, we do also note that even if that assumption is dropped, the $SU(2)_R$ symmetry must still be broken at a scale much higher than is presently observable, due to the vev of the field $\Phi_B \sim (\mathbf{1}, \mathbf{1}, \mathbf{3}, \mathbf{15}, \mathbf{0})$, which we can estimate from our results in Section 2.3, will break this symmetry at a scale of roughly $\sim 10^8$ GeV. Assuming that the new gauge bosons associated with the Pati-Salam extension to the SM are too heavy to be relevant, we turn to the mass matrices of the gauge bosons corresponding to the $SU(4)_F \times U(1)_F$ generators. In this case, the squared mass matrix of the 16 gauge bosons here takes the form,

$$\mathcal{M}_{G}^{2} = \begin{pmatrix} g_{4}^{2} \mathbf{M}_{SU(4)} & \frac{3}{2} g_{4} g_{1} \vec{M}_{SU(4) \times U(1)} \\ \frac{3}{2} g_{4} g_{1} \vec{M}_{SU(4) \times U(1)}^{T} & \frac{9}{8} g_{1}^{2} |\langle \Phi_{P} \rangle|^{2} \end{pmatrix},$$

$$(\mathbf{M}_{SU(4)})_{ij} \equiv -2 (Tr[[t^{i}, \langle \Phi_{A} \rangle][t^{j}, \langle \Phi_{A} \rangle]] + Tr[[t^{i}, \langle \Phi_{B} \rangle][t^{j}, \langle \Phi_{B} \rangle]]) + \langle \Phi_{P} \rangle^{\dagger} \{t^{i}, t^{j}\} \langle \Phi_{P} \rangle,$$

$$(\vec{M}_{SU(4) \times U(1)})_{i} \equiv \langle \Phi_{P} \rangle^{\dagger} t^{i} \langle \Phi_{P} \rangle,$$

$$(40)$$

where g_4 and g_1 are the $SU(4)_F$ and $U(1)_F$ coupling constants respectively, [A, B] denotes the commutator of A and B, $\{A, B\}$ denotes the anticommutator, and t^i denotes the i^{th} generator matrix for the fundamental representation of the group SU(4), in the basis described in Appendix A. Note that here, $\mathbf{M}_{SU(4)}$ is a 15 × 15 real symmetric matrix, while $\vec{M}_{SU(4)\times U(1)}$ is a 15-component real vector, and both of these terms have dimensions of mass squared.

Referring to Eq.(40), our first task to determine the mass eigenstates of $SU(4) \times U(1)$ gauge bosons is to determine what the vev's $\langle \Phi_A \rangle$ and $\langle \Phi_B \rangle$ are, given our requirements that the SM quark masses and mixing matrices are properly reproduced. To that end, we consider the combinations of Φ_A and Φ_B that are coupled to the up- and down-like quarks, referring to Eq.(11) and several useful variable definitions in Eq.(13). Comparing these to the vev's of Φ_A and Φ_B as given in Table 2, we arrive at

$$\begin{pmatrix} \mathbf{u}_{D} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & X_{u} \end{pmatrix} = y_{A} \begin{pmatrix} \langle A \rangle & \vec{\gamma}_{A} \\ \vec{\gamma}_{A}^{\dagger} & -Tr[\langle A \rangle] \end{pmatrix} + y_{B} \begin{pmatrix} \langle B \rangle & \vec{\gamma}_{B} \\ \vec{\gamma}_{B}^{\dagger} & -Tr[\langle B \rangle] \end{pmatrix} + M\mathbf{1}_{4\times4},$$

$$\begin{pmatrix} V_{CKM}^{\dagger} \mathbf{d}_{D} V_{CKM} & \vec{\gamma}_{d} \\ \vec{\gamma}_{d}^{\dagger} & X_{d} \end{pmatrix} = y_{A} \begin{pmatrix} \langle A \rangle & \vec{\gamma}_{A} \\ \vec{\gamma}_{A}^{\dagger} & -Tr[\langle A \rangle] \end{pmatrix} - y_{B} \begin{pmatrix} \langle B \rangle & \vec{\gamma}_{B} \\ \vec{\gamma}_{B}^{\dagger} & -Tr[\langle B \rangle] \end{pmatrix} + M\mathbf{1}_{4\times4},$$

$$(41)$$

where we remind the reader that we can use SU(4) gauge freedom to work in a basis where **u** is diagonal and $\vec{\gamma}_u = 0$, as we have done in Section 2.3, and that V_{CKM} is the CKM matrix. We recall that in that Section, the procedure we developed to produce a point in parameter space that recreates the SM quark masses and mixings already leaves us with \mathbf{u}_D , \mathbf{d}_D , and M specified. Then, simply selecting the Yukawa couplings y_A and y_B here will allow us to uniquely determine the matrices $\langle A \rangle$ and $\langle B \rangle$, as well as the vectors $\vec{\gamma}_A$ and $\vec{\gamma}_B$. Specifically, we have

$$\langle A \rangle = \frac{1}{2y_A} (\mathbf{u}_D + V_{CKM}^{\dagger} \mathbf{d}_D V_{CKM} - 2M), \quad \langle B \rangle = \frac{1}{2y_B} (\mathbf{u}_D - V_{CKM}^{\dagger} \mathbf{d}_D V_{CKM}), \tag{42}$$
$$\vec{\gamma}_A = \frac{1}{2y_A} \vec{\gamma}_d, \quad \vec{\gamma}_B = -\frac{1}{2y_B} \vec{\gamma}_d.$$

Hence, given a construction which generates the SM quark masses and mixings, we now can readily determine the necessary vev's $\langle \Phi_A \rangle$ and $\langle \Phi_B \rangle$ by only specifying two additional parameters, the Yukawa couplings y_A and y_B . With this information, we can now discuss the spectrum of the gauge bosons which results. In general, the eigenvalues of the mass matrix in Eq.(40) are highly complicated expressions and difficult to present in a compact closed form. However, we can avail ourselves of the hierarchies present between eigenvalues of \mathbf{u}_D and \mathbf{d} , as well as the nearly-diagonal form of the CKM matrix, in order to dramatically simplify matters. Specifically, we will employ the Wolfenstein parameterization of the CKM matrix,

$$\begin{pmatrix} 1 - \frac{\lambda^2}{2} - \frac{\lambda^4}{8} & \lambda & A\lambda^3(\rho - i\eta) \\ -\lambda & 1 - \frac{\lambda^2}{2} - (1 + 4A^2)\frac{\lambda^4}{8} & A\lambda^2 \\ -A(\rho + i\eta - 1)\lambda^3 & -A(\lambda^2 + \frac{\lambda^4}{2}(2(\rho + i\eta) - 1)) & 1 - \frac{A^2\lambda^4}{2} \end{pmatrix},$$
(43)

$$\lambda = 0.22453, A = 0.836, \rho = 0.122, \eta = 0.355,$$

and note that, numerically, the eigenvalues of \mathbf{u}_D and \mathbf{d} $(u_{1-3}$ and d_{1-3} , respectively) satisfy $u_2 \sim \lambda^4 u_1$, $u_3 \sim \lambda^8 u_1, d_2 \sim \lambda^2 d_1$, and $d_3 \sim \lambda^4 d_1$ for values of these parameters that reproduce the SM quark masses. By rewriting Eq.(40) using the identities in Eq.(42), and making the substitutions $u_2 \to \lambda^4 v_2, u_3 \to \lambda^8 v_3$, $d_2 \to \lambda^2 \delta_2$, and $d_3 \to \lambda^4 \delta_3$, where $u_1 \sim d_1 \sim v_2 \sim \delta_2 \sim v_3 \sim \delta_3$, we can determine an approximate hierarchy between gauge boson mass eigenvalues by expanding in the parameter λ . To start, we note that 13 of the gauge bosons acquire masses that are far in excess of any currently observable energy scale. In particular, we find that 10 gauge bosons possess masses that are of $O(u_1, d_1) \sim 10^8$ GeV, while three more have masses of $O(\lambda^2 u_1, \lambda^2 d_1) \sim 10^7$ GeV. For particles at this scale, the only even remotely observable effects stem from the imaginary part of effective 4-quark operators that facilitate $K^0 - \overline{K}^0$ mixing [25, 26]. However, from the λ expansions outlined above, it can be seen that effects which give a non-trivial phase to the coefficients of these 4-quark operators only appear at the $O(\lambda^3)$ level or higher. Since the masses of the potentially offending gauge bosons are already near the limit of the sensitivity of this tree-level probe, this essentially rules out any observable effect from the heavy bosons. Given that any observable signatures of these bosons are likely well outside the sensitivity of any experiment, we can shift our attention to the three remaining gauge bosons that acquire mass at a far lower scale.

The first of these, which we shall refer to as Z_F (the "flavor Z") consists almost entirely (up to numerically negligible corrections due to $\vec{\gamma}_{d,P}$ terms) of a combination of the generators of the $SU(3)_F$ group embedded in $SU(4)_F$ (corresponding to the first eight t^i generators, and the first eight rows/columns of the mass matrix in Eq.(40)). There does not exist a compact numerically accurate approximation for the mass of Z_F , however, expanding in the Wolfenstein parameter λ as above yields the result that $m_{Z_F} \sim O(\lambda^4 d_1, \lambda^4 u_1) \sim 10^5$ GeV. While this initially seems quite large, we note that a boson of this mass may still mediate observable tree-level flavor-changing neutral currents [15, 26, 27]. As such, it behooves us to determine how Z_F couples to the quarks, and crucially what flavor-changing interactions it can mediate. Expanding Eq.(40) in the Wolfenstein parameter λ offers an appealing approximate expression for the relevant combinations of generators corresponding to this state. Specifically, up to $O(\lambda^2)$, the $SU(4)_F$ generator combination corresponding to Z_F becomes

$$O(\lambda^7) t^1 + O(\lambda^7) t^2 + O(\lambda^4) t^3 + O(\lambda^5) t^4 + O(\lambda^5) t^5 + \sqrt{3}A\lambda^2(\rho - 1) t^6 + \sqrt{3}A\lambda^2\eta t^7 + t^8.$$
(44)

Consulting the explicit form of the generator matrices t^i in Appendix A, we see that Z_F only mediates significant flavor-changing neutral currents between the second and third generations – in both the up- and down-like sectors, flavor-changing neutral currents featuring the first generation of quarks only appear at the $O(\lambda^5) \sim 10^{-4}$ level, at most. Our chief phenomenological concern regarding Z_F , then, is the constraint arising from tree-level flavor-changing neutral currents that it mediates between the second and third generation of quarks. We shall discuss phenomenological constraints arising from this type of interaction in Section 4.2.2.

The next two light gauge bosons we consider will arise from the generators that are left unbroken by $\langle \Phi_A \rangle$ and $\langle \Phi_B \rangle$, at least in the approximation where $\vec{\gamma}_d \to 0$. In particular, we see that in the limit where $\vec{\gamma}_{d,P} \to 0$, $\langle \Phi_A \rangle$ and $\langle \Phi_B \rangle$ break $SU(4)_F \times U(1)_F$ down to $U(1)'_F \times U(1)_F$, where the generator for $U(1)'_F$ is given by the matrix t^{15} . In the limit where $\vec{\gamma}_P \to 0$, the scalar $\langle \Phi_P \rangle = (\vec{\gamma}_P, v_P)$ breaks $U(1)'_F \times U(1)_F$ down to $U(1)_D$ at the scale $v_P \sim 1 - 10$ TeV. The $\vec{\gamma}_d$ and $\vec{\gamma}_P$ terms then finally break the $U(1)_D$ symmetry at $\sim 0.1 - 1$ GeV. The two mass eigenstates we consider here, then, will be well-approximated as combinations of the $U(1)_F$ and $U(1)'_F$ bosons, with one achieving a mass of $O(v_P) \sim 1 - 10$ TeV, the scale at which $U(1)'_F \times U(1)_F$ breaks down to $U(1)_D$, and the other achieving mass only at the scale where $U(1)_D$ is broken, at approximately 0.1 - 1 GeV. Referring to Eq.(40), we see that in the limit where $\vec{\gamma}_{d,P} \to 0$, the squared mass matrix for the eventual mass eigenstates, Z_P and A_D (*i.e.*, the dark photon), becomes

$$\begin{pmatrix} B' & B \end{pmatrix} \begin{pmatrix} \frac{3g_4^2 v_P^2}{4} & -\sqrt{\frac{3}{2}} \frac{3g_4 g_1 v_P^2}{4} \\ -\sqrt{\frac{3}{2}} \frac{3g_4 g_1 v_P^2}{4} & \frac{9g_1^2 v_P^2}{8} \end{pmatrix} \begin{pmatrix} B' \\ B \end{pmatrix},$$
(45)

where B and B' refer to the $U(1)_F$ and $U(1)'_F$ gauge bosons, respectively. It is convenient to define an angle θ_P such that $g_1 = \sqrt{2/3}g_4 \tan \theta_P$. Then, we may write the mass squared matrix as

$$\frac{3g_4^2 v_P^2}{4} \begin{pmatrix} B' & B \end{pmatrix} \begin{pmatrix} 1 & -\tan \theta_P \\ -\tan \theta_P & \tan^2 \theta_P \end{pmatrix} \begin{pmatrix} B' \\ B \end{pmatrix}.$$
(46)

The mass eigenstates for these gauge bosons can then be easily determined, up to $O(\vec{\gamma}_{P,d}^2/v_P^2) \sim 10^{-4}$ corrections. Given the definition of θ_P we've provided, we see that the eigenstates are simply

$$A_D \approx B's_P + Bc_P, \ Z_P \approx B'c_P - Bs_P, \tag{47}$$

where A_D refers to the dark photon, with a mass of $\sim 0.1 - 1$ GeV, while Z_P refers to the heavier gauge boson, a dark "portal Z", with $O(v_P)$ mass. Here we have also used the abbreviations $c_P \equiv \cos \theta_P$ and $s_P \equiv \sin \theta_P$, which for simplicity we shall also employ going forward.

The masses of A_D and Z_P can also be straightforwardly determined. In the case of Z_P , we can directly extract the mass from the mass matrix of Eq.(46), arriving at

$$m_{Z_P}^2 \approx \frac{3g_4^2 v_P^2}{4c_P^2},$$
 (48)

which is accurate up to numerically negligible $O(\vec{\gamma}_{P,d}^2/v_P^2)$ corrections. The mass of the dark photon,

meanwhile, can be given up to $O(\lambda^2) \sim O(10^{-2})$ corrections by

$$\begin{split} m_{A_D}^2 &\approx \frac{2g_4^2 s_P^2}{3} \left(\frac{7}{8} (\vec{\gamma}_P^* \cdot \vec{\gamma}_P) + \mu_1^{A_D} (\vec{\gamma}_d^* \cdot \vec{\gamma}_d) + \mu_2^{A_D} |(\vec{\gamma}_d)_1|^2 \right), \end{split} \tag{49} \\ \mu_1^{A_D} &\equiv \frac{4(4M - u_1)^2}{y_A^2 (d_1 - u_1)^2 + y_B^2 (d_1 + u_1 - 8M)^2}, \\ \mu_2^{A_D} &\equiv \frac{16M(d_1 - u_1)(y_A^2 (d_1 - u_1)(-3M + u_1) + y_B^2 ((-4M + u_1)^2 - Mu_1 + d_1 (-3M + u_1))))}{(y_A^2 (d_1 - u_1)^2 + y_B^2 (d_1 + u_1 - 8M)(d_1 + u_1 - 4M))^2 + 16y_A^2 y_B^2 M^2 (d_1 - u_1)^2}, \end{split}$$

where to arrive at this expression, we have exploited the fact that $d_1, u_1 \gg d_{2,3}, u_{2,3}, v_P$. At this point, we have obtained the masses and eigenvectors for the three new gauge bosons that may be phenomenologically relevant – namely, the $SU(3)_F$ boson Z_F with a mass of $O(10^5 \text{ GeV})$, the "portal Z" field Z_P with a mass roughly of O(1 TeV), and the dark photon field A_D , with a mass of O(0.1 - 1 GeV).

3 Couplings, Kinetic Mixing, and Low-Energy Parameters

Before computing the observable experimental signatures of this seup, it is useful to explicitly determine the couplings between various phenomenologically relevant particles in the model. In particular, in light of our discussion of the fermion spectrum in Section 2.3, we shall focus on the SM and new physics gauge boson and scalar couplings of the SM fermions, the portal matter fermions, and the heavy top partner.

3.1 $SU(4)_F \times U(1)_F$ Gauge Bosons

We begin our discussion by deriving the couplings of the new gauge bosons that arise in our setup due to our $SU(4)_F \times U(1)_F$ extension of the SM gauge group. In Section 2.4, we have determined that only three of the new gauge bosons are sufficiently light so as to have an observable phenomenological effect: A (comparatively) light $SU(3)_F$ boson that we have dubbed Z_F , with a mass of roughly $O(10^5 \text{ GeV})$, a "portal Z" field with a mass of O(1 - 10 TeV), which we denote as Z_P , and a dark photon, A_D , with a mass of approximately O(0.1 - 1 GeV).

To start, we consider the couplings of the dark photon A_D to the fermions of the theory before KM takes place. In the limit where $\vec{\gamma}_{P,d} \to 0$, the dark photon is given as a simple combination of two U(1) bosons, described in Eq.(47). In fact, the results of Eq.(47) are exceedingly numerically accurate, even once contributions from $\vec{\gamma}_{P,d}$ terms are included: Mixing between A_D , the gauge boson Z_P , and the gauge bosons corresponding to the $SU(3)_F$ embedded in $SU(4)_F$ only occurs at second order in the quantities $\vec{\gamma}_{P,d}$, and is hence numerically negligible, while mixing between A_D and the gauge bosons corresponding to the other $SU(4)_F$ generators is suppressed by the latter's enormous masses – in Section 2.4, we found these to be of $O(10^{7-8})$ GeV. Thus, we can use the combinations given in Eq.(47) to derive the couplings of A_D and Z_P without concerning ourselves with any additional complicating effects. Noting that the B' boson corresponds to the 15th generator of $SU(4)_F$ (t^{15} , as listed in Appendix A), and B is simply the $U(1)'_F$ boson, we can straightforwardly find a coupling matrix by writing the fermions as 9-dimensional vectors in flavor space, as in Eq.(10). We find that, for up-like and down-like quarks, charged leptons, and neutrinos (at least ignoring any extra structure the model may need to accommodate neutrino flavor phenomenology), the dark photon couplings are given by

$$g_4 \sqrt{\frac{2}{3}} s_P (\mathbf{C}_{A_D}^{u,d,e,\nu})_{L,R} = -g_4 \sqrt{\frac{2}{3}} s_P \mathcal{U}_{L,R}^{u,d,e,\nu\dagger} \begin{pmatrix} \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} \\ \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} \\ \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{1}_{3\times3} \end{pmatrix} \mathcal{U}_{L,R}^{u,d,e,\nu}, \tag{50}$$

where we have used the fact that the $U(1)_F$ coupling constant, g_1 , is given in terms of the $SU(4)_F$ coupling constant by the relation $g_1 = \sqrt{2/3}g_4 \tan \theta_P$. The overall negative sign in this expression is simply an artifact of the sign conventions we have selected for the definitions of $U(1)_F$ and $U(1)_D$. At this point, we have omitted the additional couplings arising from kinetic mixing with the SM hypercharge

field as noted earlier – these will introduce shifts of $e\epsilon Q \mathbf{1}_{9\times9}$ to the above expression, where e is the proton charge, ϵ is the kinetic mixing parameter and Q is the electric charge of the given field. While these shifts are obviously significant in dark matter phenomenology, since they encapsulate the means by which, for example, a dark matter field might annihilate to form SM particles, they are of less importance to us when considering other constraints here: Because the $e\epsilon Q$ shift is always proportional to the identity matrix for a given fermion electric charge, it does not facilitate flavor-changing neutral currents or portal matter decay to SM particles. Restricting our attention to only fermions which are light enough to be phenomenologically relevant, namely the SM quarks, the heavy top partner, and the lightest two portal matter fields in each sector, we can rewrite the coupling matrices using the truncated flavor vectors of Eq.(29). In the up-like quark sector, we arrive at the truncated coupling matrices

$$(\mathbf{C}_{A_D}^u)_L \approx \begin{pmatrix} 0 & -\frac{(\vec{\gamma}_P)_1}{v_P} & 0\\ (\mathbf{C}_{A_D}^u)_L^{SM} & 0 & -\frac{(\vec{\gamma}_P)_1}{v_P} & 0\\ 0 & 0 & 0 & -\frac{(\vec{\gamma}_P)_3}{v_P} & qrz_{P2} \frac{m_t}{v_P} \frac{(\vec{\gamma}_P)_3}{v_P} \\ 0 & 0 & 0 & 0 & -\frac{r}{(1-z_{P1}^2)} \frac{m_t}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} & -\frac{qz_{P2}}{(1-z_{P2}^2)} \frac{(\vec{\gamma}_P)_3}{w_P} \\ 0 & 0 & qrz_{P2} \frac{m_t}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} & -\frac{qz_{P2}}{(1-z_{P2}^2)} \frac{(\vec{\gamma}_P)_3}{w_P} & 1 & 0\\ 0 & 0 & qrz_{P2} \frac{m_t}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} & -\frac{qz_{P2}}{(1-z_{P2}^2)} \frac{(\vec{\gamma}_P)_3}{v_P} & 0 & 1 \end{pmatrix} \end{pmatrix},$$

$$(\mathbf{C}_{A_D}^u)_R \approx \mathbf{N}_2 \begin{pmatrix} (\mathbf{C}_{A_D}^u)_R^{SM} & 0 & 0 & \frac{(\vec{\gamma}_P)_1}{w_P} \\ 0 & 0 & 0 & 0 & \frac{(\vec{\gamma}_P)_1}{v_P} \\ 0 & 0 & 0 & 0 & -z_{P1} \frac{m_t}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} & -\frac{qz_{P2}}{(1-z_{P1}^2)} \frac{(\vec{\gamma}_P)_3}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} & -\frac{q}{(1-z_{P2}^2)} \frac{(\vec{\gamma}_P)_3}{v_P} \\ 0 & 0 & 0 & 0 & -z_{P1} \frac{m_t}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} & -\frac{q}{(1-z_{P1}^2)} \frac{m_t}{M_t} \frac{(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_1}{(1-z_{P1}^2)} \frac{(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_3}{v_P} \frac{(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_1}{(\frac{\vec{\gamma}_P}{v_P})_1} & \frac{(\vec{\gamma}_P)_2}{v_P} & -\frac{q}{(1-z_{P2}^2)} \frac{m_t}{W_t} \frac{(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_1}{v_P} \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_1(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}}{v_P} & \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_2}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_2}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_2}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_1}{v_P} & qr \frac{(\vec{\gamma}_P)_2(\vec{\gamma}_P)_3}{v_P} \\ \frac{(\vec{\gamma}_P)_2(\vec$$

where q, r, z_{P1} , and z_{P2} are defined as in Eq.(30), and N₂ is defined as in Eq.(37). The down-like quark sector, meanwhile, has its truncated coupling matrices given by

$$(\mathbf{C}_{A_{D}}^{d})_{L} \approx \begin{pmatrix} \frac{(\mathbf{W}_{d}^{\dagger}\vec{\gamma}\vec{p})\otimes(\vec{\gamma}_{P}^{\dagger}\mathbf{W}_{d})}{v_{P}^{2}} & -\frac{\mathbf{W}_{d}^{\dagger}\vec{\gamma}_{P}}{v_{P}} & \mathbf{0}_{3\times1} \\ -\frac{\vec{\gamma}_{P}^{\dagger}\mathbf{W}_{d}}{v_{P}} & 1 & 0 \\ \mathbf{0}_{1\times3} & 0 & 1 \end{pmatrix},$$

$$(\mathbf{C}_{A_{D}}^{d})_{R} \approx \begin{pmatrix} \frac{(\mathbf{W}_{d}^{\dagger}\vec{\gamma}\vec{p})\otimes(\vec{\gamma}_{P}^{\dagger}\mathbf{W}_{d})}{v_{P}^{2}} & \mathbf{0}_{3\times1} & -\frac{\mathbf{W}_{d}^{\dagger}\vec{\gamma}_{P}}{v_{P}} \\ \mathbf{0}_{1\times3} & 1 & 0 \\ -\frac{\vec{\gamma}_{P}^{\dagger}\mathbf{W}_{d}}{v_{P}} & 0 & 1 \end{pmatrix},$$

$$(52)$$

where we remind the reader that \mathbf{W}_d is the unitary matrix first described in Eq.(13), and that in the $SU(4)_F$ basis we have chosen, \mathbf{W}_d is equal to the CKM matrix. The corresponding coupling matrix for the charged leptons can be given in complete analogy to Eq.(52), with the sole exception that the matrix \mathbf{W}_d must be replaced by the appropriate rotation matrix given the lepton couplings to the scalars Φ_A and Φ_B .

In I and II, it was seen that in the event that the portal matter mixes with electrons, there exists a parity-violating interaction of the right-handed electrons with the dark photon field. Consulting Eq.(52), we observe that the analogous terms emerge at the $O(\vec{\gamma}_P^2/v_P^2)$ level, however, they occur *identically* for the right- and left-handed fields (at least up to corrections due to the super-heavy fermions that we have

omitted from our truncated coupling matrices). As a result, no such parity-violating interaction occurs in this model, in contrast to those of **I** and **II**. Generically, this can be expected to be the case due to the more left-right symmetric form of the model: The left-handed electrons receive a chiral correction from their coupling from the $SU(2)_L$ doublet portal matter, while the right-handed ones receive the same correction from $SU(2)_L$ singlet portal matter. The couplings in the neutrino sector may be qualitatively similar to those that we have already explored, however, we note that the recreation of light neutrino masses and the observed mixing matrix will likely require significant modifications to the neutrino mass matrix, which are beyond the scope of this paper. Notably, we have retained $O(\vec{\gamma}_P^2/v_P^2)$ terms in the components of Eqs.(51) and (52) which correspond to mixing among the SM quarks.⁵ In spite of their minute magnitude, with $\vec{\gamma}_P/v_P \sim 10^{-(3-4)}$, we have kept them here because they facilitate highlyconstrained flavor-changing neutral currents mediated by the extremely light A_D boson. In Section 4.1, we shall in fact derive significant non-trivial constraints on the model parameter space from these interactions. We do note in passing that similar $O(\vec{\gamma}_P^2/v_P^2)$ terms exist elsewhere in Eqs.(51) and (52), however, we have omitted them here since they have a negligible effect on any observable physics.

Apart from the flavor-changing interactions we observe in the SM, the dark photon also facilitates $O(\vec{\gamma}_P/v_P)$ interactions between the portal matter fermion fields and those uncharged under $U(1)_D$. namely the SM quarks and the heavy top partner. In practice, due to A_D 's tiny mass compared to other gauge bosons in the theory, these interactions make meaningful contributions to a number of interesting processes; most notably, they dominate the decay of portal matter to an SM quark, or the decay of a top partner to portal matter (or vice versa, depending on which process is kinematically allowed). The presence of these couplings for the extremely light dark photon field can substantially simplify our later discussions of the couplings of heavier gauge bosons: From the Equivalence Theorem [23], and the fact that $|\vec{\gamma}_P| \sim m_{A_D} \sim 0.1 - 1$ GeV, we would anticipate that the overall strength of the A_D -facilitated interaction between portal and non-portal matter would undergo a substantial enhancement over what the $\vec{\gamma}_P/v_P$ suppression in its coupling would suggest, since this suppression would be cancelled by m_{A_D} (at least for the longitudinal mode of A_D). Since obviously such an enhancement doesn't exist for $\vec{\gamma}_P/v_P$ suppressed couplings for heavier gauge bosons, which in turn mediate the same $U(1)_D$ -breaking couplings as we see emerging from A_D , we therefore can omit a detailed evaluation of $O(\vec{\gamma}_P/v_P)$ -suppressed effects in the couplings of Z_F , Z_P , the SM electroweak gauge bosons and the SM-like Higgs; in all cases we consider, these effects are overwhelmed by those arising from A_D .

We next turn to the couplings for the "portal Z" boson, Z_P . Again referencing Eq.(47), this time to get the approximate mass eigenvector for Z_P , we arrive at the coupling matrices

$$-\frac{g_4}{2\sqrt{6}c_P} (\mathbf{C}_{Z_P}^{u,d,e,\nu})_L = -\frac{g_4}{2\sqrt{6}c_P} \mathcal{U}_L^{u,d,e,\nu\dagger} \begin{pmatrix} \mathbf{1}_{6\times 6} & \mathbf{0}_{6\times 1} & \mathbf{0}_{6\times 1} & \mathbf{0}_{6\times 1} \\ \mathbf{0}_{1\times 6} & 1 - 4c_P^2 & 0 & 0 \\ \mathbf{0}_{1\times 6} & 0 & 4s_P^2 & 0 \\ \mathbf{0}_{1\times 6} & 0 & 0 & 1 - 4c_P^2 \end{pmatrix} \mathcal{U}_L^{u,d,e,\nu}, \quad (53)$$
$$-\frac{g_4}{2\sqrt{6}c_P} (\mathbf{C}_{Z_P}^{u,d,e,\nu})_R = -\frac{g_4}{2\sqrt{6}c_P} \mathcal{U}_R^{u,d,e,\nu\dagger} \begin{pmatrix} \mathbf{1}_{6\times 6} & \mathbf{0}_{6\times 1} & \mathbf{0}_{6\times 1} & \mathbf{0}_{6\times 1} \\ \mathbf{0}_{1\times 6} & 4s_P^2 & 0 & 0 \\ \mathbf{0}_{1\times 6} & 0 & 1 - 4c_P^2 & 0 \\ \mathbf{0}_{1\times 6} & 0 & 0 & 1 - 4c_P^2 \end{pmatrix} \mathcal{U}_R^{u,d,e,\nu}.$$

We can then determine what the couplings are for the quarks that are light enough to remain phenomenologically relevant, as we've already done for A_D , by inserting our results for $\mathcal{U}_{L,R}^{u,d}$ from Eqs.(30), (31), (37), and (38) into Eq.(53). In fact, up to $O(\vec{\gamma}_P/v_P)$ corrections, which we note are negligible compared

⁵In spite of the fact that our expressions for $\mathcal{U}_{L,R}^{u}$ and for A_D 's mass eigenvector are only valid up to $O(\vec{\gamma}_P/v_P)$, these expressions (at least for flavor-changing interactions, which are of phenomenological interest here) are numerically valid, because the $O(\vec{\gamma}_P^2/v_P^2)$ corrections to the matrices $\mathcal{U}_{L,R}^{u}$ don't contribute to the $O(\vec{\gamma}_P^2/v_P^2)$ flavor-changing couplings in the SM, and we shall see that the gauge boson Z_P , which may mix with A_D at the $O(\vec{\gamma}_P^2/v_P^2)$ level, has SM flavor-universal couplings up to $O(\vec{\gamma}_P/v_P)$ corrections.

to similar interactions arising from A_D , we find that we can write the truncated coupling matrices as

$$(\mathbf{C}_{Z_{P}}^{u})_{L} \approx diag(1, 1, 1, 1, 1 - 4c_{P}^{2}, 4s_{P}^{2}), (\mathbf{C}_{Z_{P}}^{u})_{R} \approx diag(1, 1, 1, 1, 4s_{P}^{2}, 1 - 4c_{P}^{2}), (\mathbf{C}_{Z_{P}}^{d,e})_{L} \approx diag(1, 1, 1, 1 - 4c_{P}^{2}, 4s_{P}^{2}), (\mathbf{C}_{Z_{P}}^{d,e})_{R} \approx diag(1, 1, 1, 4s_{P}^{2}, 1 - 4c_{P}^{2}),$$

$$(\mathbf{C}_{Z_{P}}^{d,e})_{R} \approx diag(1, 1, 1, 4s_{P}^{2}, 1 - 4c_{P}^{2}),$$

$$(\mathbf{C}_{Z_{P}}^{d,e})_{R} \approx diag(1, 1, 1, 4s_{P}^{2}, 1 - 4c_{P}^{2}),$$

where we have noted that in the absence of corrections due to mixing with heavy partner states and $\vec{\gamma}_P$ terms, the coupling matrix for charged leptons here is the same as that for down-like quarks. We again shall not comment on the neutrino sector coupling matrix, which may be dependent on substantial additional physics.

We finally address couplings which arise from the gauge boson Z_F , the "flavor Z" that represents the only gauge boson from the $SU(3)_F$ flavor symmetry which possesses a low enough mass to have some phenomenological impact. Consulting Eq.(44), we see that the coupling matrix for this gauge boson may be written,

$$\frac{g_4}{2\sqrt{3}} (\mathbf{C}_{Z_F}^{u,d,e,\nu})_{L,R} = \frac{g_4}{2\sqrt{3}} \mathcal{U}_{L,R}^{u,d,e,\nu\dagger} \begin{pmatrix} \mathbf{\Lambda}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} \\ \mathbf{0}_{3\times3} & \mathbf{\Lambda}_{3\times3} & \mathbf{0}_{3\times3} \\ \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} \end{pmatrix} \mathcal{U}_{L,R}^{u,d,e,\nu},$$

$$\mathbf{\Lambda} \equiv \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 3A\lambda^2(\rho - i\eta - 1) \\ 0 & 3A\lambda^2(\rho + i\eta - 1) & -2 \end{pmatrix},$$
(55)

where we remind the reader that A, λ , ρ , and η are the Wolfenstein parameters. Notably, Z_F couples to SM fields and their heavy partners equivalently, and as such doesn't facilitate any couplings between them. However, we see that this gauge boson can produce flavor-changing neutral currents in the SM quark sector. Focusing on this possibility, we consider what the coupling matrices for the SM quarks look like in our setup, arriving at

$$(\mathbf{C}_{Z_F}^u)_{L,R}^{SM} = \mathbf{\Lambda}, \quad (\mathbf{C}_{Z_F}^d)_{L,R}^{SM} = \mathbf{W}_d^{\dagger} \mathbf{\Lambda} \mathbf{W}_d \approx \begin{pmatrix} 1 & 0 & 0\\ 0 & 1 & 3A\lambda^2(\rho - i\eta)\\ 0 & 3A\lambda^2(\rho + i\eta) & -2 \end{pmatrix},$$
(56)

where to derive the expression for $(\mathbf{C}_{Z_F}^d)_{L,R}^{SM}$, we have used the fact that \mathbf{W}_d is simply given by the CKM matrix, and used the Wolfenstein paramterization of the CKM given in Eq.(43), keeping terms up to $O(\lambda^2)$. Note that in spite of its appearance (and explicit dependence on Wolfenstein parameters), the coupling of Eq.(56) is not simply an artifact of our choice of $SU(4)_F$ gauge: Effecting an $SU(4)_F$ transformation here to a frame which, for example, \mathbf{W}_d is equal to the identity matrix and \mathbf{W}_u is non-trivial should correspondingly alter the gauge boson mass matrix so that the resulting coupling is preserved. We do, however, note that the right-handed coupling expressions in Eq.(56) ignore the sign flips (phase rotations) of various right-handed quark fields observed in Eqs.(37) and (38). Since they cancel in any phenomenological results we shall discuss, we have omitted them above for the sake of brevity.

3.2 SM Gauge Bosons and the Higgs

Having discussed the couplings of the fermions to new gauge bosons in the theory, we now address the coupling matrices for usual SM fields, namely the Z and W gauge bosons and the light SM-like Higgs doublet embedded in the bidoublet H. We begin our discussion with the Z boson. Writing the fermions as 9-component vectors in flavor space as outlined in Eq.(10), we can write the coupling matrix of the Z

as

$$\frac{g}{c_w} (\mathbf{C}_Z^{u,d,e,\nu})_L = \frac{g}{c_w} \mathcal{U}_L^{u,d,e,\nu\dagger} \begin{pmatrix} (T_{3L} - Qs_w^2) \mathbf{1}_{3\times 3} & \mathbf{0}_{3\times 3} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{0}_{3\times 3} & -Qs_w^2 \mathbf{1}_{3\times 3} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & (T_{3L} - Qs_w^2) & \mathbf{0} & \mathbf{0} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & \mathbf{0} & -Qs_w^2 & \mathbf{0} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & \mathbf{0} & \mathbf{0} & -Qs_w^2 & \mathbf{0} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & \mathbf{0} & \mathbf{0} & -Qs_w^2 \end{pmatrix} \mathcal{U}_L^{u,d,e,\nu},$$
(57)

$$\frac{g}{c_w} (\mathbf{C}_Z^{u,d,e,\nu})_R = \frac{g}{c_w} \mathcal{U}_R^{u,d\dagger} \begin{pmatrix} -Qs_w^2 \mathbf{1}_{3\times 3} & \mathbf{0}_{3\times 3} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{0}_{3\times 3} & -Qs_w^2 \mathbf{1}_{3\times 3} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} & \mathbf{0}_{3\times 1} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & (T_{3L} - Qs_w^2) & \mathbf{0} & \mathbf{0} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & \mathbf{0} & -Qs_w^2 & \mathbf{0} \\ \mathbf{0}_{1\times 3} & \mathbf{0}_{1\times 3} & \mathbf{0} & \mathbf{0} & -Qs_w^2 \end{pmatrix} \mathcal{U}_R^{u,d,e,\nu},$$

up to $O(\epsilon)$ corrections due to kinetic mixing, which won't be phenomenologically significant here, since they'll only represent a uniform small correction to the couplings of the three portal matter states, arising from mixing with the dark photon A_D . Here, T_{3L} refers to the left-handed isospin of the SM fermion species and Q refers to its electric charge. We can then find the coupling matrices for the phenomenologically relevant mass eigenstates by simply truncating the above matrices to exclude the extremely heavy states and rotating by the approximate diagonalization matrices given in Eqs.(30), (31), (37), and (38). Up to numerically negligible $O(\vec{\gamma}_P/v_P)$ terms, we arrive at

$$(\mathbf{C}_{Z}^{u})_{L} \approx \begin{pmatrix} \frac{1}{2} - \frac{2}{3}s_{w}^{2} & 0 & 0 & 0 & 0 & 0 \\ 0 & \frac{1}{2} - \frac{2}{3}s_{w}^{2} & 0 & 0 & 0 & 0 \\ 0 & 0 & \frac{1}{2} - \frac{2}{3}s_{w}^{2} & \frac{7}{2}\frac{m_{t}}{M_{t}} & 0 & 0 \\ 0 & 0 & \frac{7}{2}\frac{m_{t}}{M_{t}} & -\frac{2}{3}s_{w}^{2} & 0 & 0 \\ 0 & 0 & 0 & 0 & \frac{1}{2} - \frac{2}{3}s_{w}^{2} & 0 \\ 0 & 0 & 0 & 0 & 0 & -\frac{2}{3}s_{w}^{2} \end{pmatrix},$$
(58)
$$(\mathbf{C}_{Z}^{u})_{R} \approx \begin{pmatrix} -\frac{2}{3}s_{w}^{2} & 0 & 0 & 0 & 0 \\ 0 & -\frac{2}{3} & 0 & 0 & 0 & 0 \\ 0 & 0 & -\frac{2}{3}s_{w}^{2} & 0 & 0 & 0 \\ 0 & 0 & 0 & -\frac{2}{3}s_{w}^{2} & 0 & 0 \\ 0 & 0 & 0 & 0 & \frac{1}{2} - \frac{2}{3}s_{w}^{2} & 0 \\ 0 & 0 & 0 & 0 & -\frac{2}{3}s_{w}^{2} \end{pmatrix},$$

where we remind the reader that the variable r is defined in Eq.(30). We see that at this level of approximation, the sole new coupling for the Z boson (other than its diagonal couplings to the new fermions, which directly follow their $SU(2)_L \times U(1)_Y$ quantum numbers) is between the top quark and its vector-like partner, which can be quite large – the suppression ratio m_t/M_t for this coupling can be as high as ~ 0.1, for top partner masses near 1 TeV. We note that there do exist $O(m_t^2/M_t^2)$ corrections to the $Z\bar{t}t$ coupling, however, since these corrections would be at most on the order of a few percent, they are well within current constraints for modifications of the top-Z coupling [28]. Of course, such small variations in the top quark couplings to the Z may be probed by precision measurements made at future e^+e^- colliders. In the down-like and charged lepton sectors, we have an analogous result to Eq.(58), however, assuming that the b and τ partners are too massive to influence the low-energy phenomenology of the theory, we find no significant departures from the SM behavior of the Z and diagonal coupling matrices.⁶

Next, we address the W boson couplings, restricting our attentions to the quark sector in order to avoid ambiguities arising in the neutrino sector in this model. We may write our coupling matrices here

⁶As an aside, we note that even in the event of a much lighter *b* partner, brought about by a percent-level tuning of the Higgs Yukawa coupling to the down quarks, which might then have a mass comparable to the top partner mass $M_t \sim 1$ TeV, we would not observe a measurable effect in the tightly-constrained $Z\bar{b}b$ coupling, since this would still be suppressed by $O(m_b^2/M_b^2) \sim 10^{-6}$, assuming $M_b \sim M_t$, which is several orders of magnitude below present constraints [29]

$$\frac{g}{\sqrt{2}}(\mathbf{C}_{W})_{L} = \frac{g}{\sqrt{2}} \mathcal{U}_{L}^{u^{\dagger}} \begin{pmatrix} \mathbf{1}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 1 & 0 & 0 \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 0 & 0 & 0 \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 1 & 0 & 0 \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 1 & 0 & 0 \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} \end{pmatrix} \mathcal{U}_{R}^{d}.$$
(59)

Up to numerically insignificant $O(\vec{\gamma}_P/v_P)$ corrections, we can derive the coupling matrix for the lefthanded phenomenologically relevant fermions using Eqs.(30) and (31) once again, yielding a coupling matrix (and hence the CKM matrix) of

$$(\mathbf{C}_W)_L \approx \begin{pmatrix} (\mathbf{W}_d)_{11} & (\mathbf{W}_d)_{12} & (\mathbf{W}_d)_{13} & 0 & 0\\ (\mathbf{W}_d)_{21} & (\mathbf{W}_d)_{22} & (\mathbf{W}_d)_{23} & 0 & 0\\ (1 - \frac{r^2}{2} \frac{m_t^2}{M_t^2})(\mathbf{W}_d)_{31} & (1 - \frac{r^2}{2} \frac{m_t^2}{M_t^2})(\mathbf{W}_d)_{32} & (1 - \frac{r^2}{2} \frac{m_t^2}{M_t^2})(\mathbf{W}_d)_{33} & 0 & 0\\ r \frac{m_t}{M_t} (\mathbf{W}_d)_{31} & r \frac{m_t}{M_t} (\mathbf{W}_d)_{32} & r \frac{m_t}{M_t} (\mathbf{W}_d)_{33} & 0 & 0\\ 0 & 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 0 & 0 \end{pmatrix},$$
(60)

where r remains as defined in Eq.(30). Note, at least to this order, that the first row of this coupling matrix remains unitary when restricted to the SM quarks. Here, we emphasize that in the truncated coupling matrix, only five down-like quarks remain physically relevant (the three SM quarks and two portal matter quarks), while six up-like quarks do (the three SM quarks, the top partner, and two portal matter quarks), as can be seen from the original definition of the truncated flavor-space vectors in Eq.(29). We also note that, unlike elsewhere in this work, we have retained the $O(m_t^2/M_t^2)$ terms in the coupling matrix here; we shall see that terms of this order represent the leading contribution of the top partner's loop-induced correction to neutral meson mixing, and as such, we must retain these terms for numerical consistency. The right-handed coupling matrix $(\mathbf{C}_W)_R$ is substantially less phenomenologically interesting– since the only right-handed fermions which couple to the W boson are portal matter fields, and they do so diagonally (at least in the limit, which holds to $O(10^{-6})$ as discussed in Section 2.3), that mixing between the portal matter states is numerically negligible), the W does not exhibit numerically significant couplings to the right-handed SM quarks, nor does it facilitate decays of any of the new fermions in the model.

We conclude our exploration of the couplings in our model by considering the SM Higgs field, or rather the combination of elements of the bidoublet scalar H that corresponds to such a field. As noted in Section 2.2, we can estimate the scalar eigenstate corresponding to the SM Higgs field as in Eq.(7), which in turn allows us, with reference to the Yukawa action of Eq.(8), to write the Higgs coupling matrices to the quarks as

$$y_{u,d}\mathbf{C}_{H}^{u,d} = y_{u,d}\mathcal{U}_{R}^{u,d\dagger} \begin{pmatrix} \mathbf{0}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{1}_{3\times3} & \mathbf{0}_{3\times3} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} & \mathbf{0}_{3\times1} \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 0 & 0 & 0 \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 0 & 0 & 0 \\ \mathbf{0}_{1\times3} & \mathbf{0}_{1\times3} & 1 & 0 & 0 \end{pmatrix} \mathcal{U}_{L}^{u,d},$$
(61)

where the constants $y_{u,d}$ are given in Eq.(11). An analogous matrix for the charged leptons should be identical to that of the down-like quarks, up to radiative corrections to the parameter y_d . Rather than relying on our truncated rotation matrices here, which by removing the heavy partners of the up and charm quarks, omits the seesaw mechanism by which these SM quarks acquire mass, it is more instructive

as

here to simply work in the limit where $\vec{\gamma}_{P,d} \to 0$ using the rotation matrices of Eqs.(23) and (33). Doing this, and then truncating the matrix to remove the extremely heavy fermions as earlier yields the coupling matrix

where the complete coupling is given by this matrix plus that generated by its Hermitian conjugate. Notably, among the SM quarks the Higgs coupling matrix is simply given by the normal SM Higgs coupling matrix – this conclusion holds up to $O(m_t^2/M_t^2) \sim O(10^{-2})$ corrections to the $H\bar{t}t$ coupling. There are, however, additional couplings between the top partner and the top quark itself – the largest of these terms is, in fact O(1). In practice, these couplings will contribute significantly to the decay of the top partner to the top quark, as is often the case in models with additional vector-like quarks mixed with the third generation [30]. In the down-like quark and charged lepton sectors, the results are analogous, however, given the fact that the heavy b and τ partners are likely too massive to be observed, the approximate Higgs couplings in these sectors precisely matches the SM result, up to insignificant numerical corrections.

3.3 Kinetic Mixing

Having addressed the fermion and gauge boson spectra here, it is useful at this point to comment on the magnitude and effects of the kinetic mixing between $U(1)_F$ and $U(1)_Y$ that will arise from the one-loop contributions of the model's fermion fields. For simplicity, we shall assume that kinetic mixing vanishes until the scale at which the Pati-Salam group is broken down to the SM and the dark/flavor group remains $SU(4)_F \times U(1)_F$. At this scale, the only Abelian groups which may enjoy kinetic mixing are the $U(1)_F$ and $U(1)_Y$ groups, so we shall compute this mixing here.⁷ As in **II**, we note that both SM and portal matter fields will contribute to the kinetic mixing via vacuum polarization-like diagrams at one loop. In the original basis, the SM hypercharge boson \hat{B} will mix with the $U(1)_F$ boson \hat{B}_F via a term of the form,

$$\mathcal{L}_{KM} = \frac{\epsilon}{2c_w c_P} \hat{B}_{\mu\nu} \hat{B}_F^{\mu\nu},\tag{63}$$

where c_w is the familiar Weinberg angle and $c_P \equiv \cos \theta_P$ is the cosine of the angle θ_P described directly above Eq.(46), and $\hat{B}_{\mu\nu}$ and $\hat{B}_F^{\mu\nu}$ are the field strength tensors of the $U(1)_Y$ and $U(1)_F$ fields, respectively. Given this normalization convention, the kinetic mixing term ϵ here becomes

$$\epsilon = \sqrt{\frac{2}{3}} \frac{(gs_w)(g_4 s_P)}{24\pi^2} \sum_i \left(\frac{Y_i}{2}\right) Q_i^F \log \frac{m_i^2}{\mu^2},\tag{64}$$

where s_w and s_P are simply the sines of the same angles referenced in c_w and c_P , g is the $SU(2)_L$ coupling constant, and g_4 is the $SU(4)_F$ coupling constant. The sum over i is performed over all the fermions in the theory, $Y_i/2$ is the SM hypercharge of fermion i, Q_i^F is its charge under $U(1)_F$, m_i is its mass, and

⁷As was noted in Section 2.1, in the UV theory as written (with all SM gauge symmetries are contained in the non-Abelian Pati-Salam group factors), kinetic mixing mediated by a vacuum polarization-like diagram is forbidden. However, higher-order operators stemming from insertions of scalar vev's will still generate kinetic mixing here, and a truly concerned reader can assume that the $U(1)_F$ is unified with either the $SU(4)_c$ or $SU(2)_R$, the two Pati-Salam groups which contain the $U(1)_Y$ symmetry, at some higher scale, and some form of symmetry breaking at this scale breaks the resulting theory down to the SM gauge group by $U(1)_F$.

 μ is an arbitrary mass scale which will cancel out of the final calculation. Ignoring the mixing between various states of different representations under $U(1)_Y$ or $U(1)_F$, the effects of which are numerically negligible anyway, gives

$$\epsilon \approx (4.2 \times 10^{-4}) \left(\frac{g_4 s_P}{g s_w}\right) \left[\log \frac{m_{P1}^e}{m_{P1}^u} + \log \frac{m_{P1}^\nu}{m_{P1}^d} + 2\log \frac{m_{P2}^d}{m_{P2}^u} + 2\log \frac{m_{P2}^e}{m_{P2}^u} + \frac{4}{5}\log \frac{m_{P3}^d}{m_{P3}^u} + \frac{4}{5}\log \frac{m_{P3}^e}{m_{P3}^u} \right] + \sum_i \left(\frac{1}{5}\log \frac{m_{di}}{m_{ui}} + \frac{3}{5}\log \frac{m_{ei}}{m_{ui}} + \frac{1}{5}\log \frac{m_{\nu i}}{m_{ui}} + \frac{4}{5}\log \frac{M_{di}}{M_{ui}} + \frac{4}{5}\log \frac{M_{ei}}{M_{ui}} \right),$$
(65)

where here we have used the same mass labelling convention as Eq.(12), with the sub/superscripts u referring to the up-like quark sector, d to the down-like quark sector, e to the charged lepton sector, and ν to the neutrino sector (of course, since the model's neutrino sector remains incomplete, the contributions from it to this mixing are somewhat suspect; we include them here for the sake of completeness). We note that some of logarithms of ratios of the exotic particles' masses may have either sign. In the absence of significant hierarchies, we would anticipate that ϵ would be of $O(10^{-4})$ if $g_4 s_P$ were approximately equal to gs_W , however, we note that there exist several mass ratios in Eq.(65) that are necessarily quite hierarchical. For example, if we assume that each SM neutrino possesses a mass of approximately 0.1 eV and that the masses of the the portal matter fields $m_{P1,P2,P3}^{u,d,e,\nu}$ are close to degenerate (that is, $m_{Pi}^u \sim m_{Pi}^d \sim m_{Pi}^e \sim m_{Pi}^{\nu}$ for i = 1, 2, 3, at least for the purposes of computing the logs of their ratios), then we can estimate the magnitude of Eq.(65) by noting that $\log(m_{di,ei}/m_{ui}) \sim -\log(M_{di,ei}/M_{ui})$, which holds as long as there are no significant hierarchies between the couplings to the scalar H among the charged leptons, up-like quarks, and down-like quarks. We then arrive at a rough numerical estimate of

$$\epsilon \sim (3 \times 10^{-3}) \left(\frac{g_4 s_P}{g s_w}\right),\tag{66}$$

which is in fact an order of magnitude larger than the numerical coefficient in front of Eq.(65) might suggest. While this level of kinetic mixing is still not unfeasible if $g_4s_P \simeq gs_w = e$, it does suggest that this coupling is unlikely to be much greater than this, and that to produce smaller $O(10^{-4})$ values for the kinetic mixing parameter, the coupling combination g_4s_P should likely be somewhat smaller, perhaps closer to ~ 0.1e.

With ϵ computed, the treatment of kinetic mixing is ultimately entirely analogous to that of **I** and **II** (albeit with no additional mixing due to scalars charged under both the dark and SM gauge groups, which occur in **II** but not here), with dark photon couplings to SM fields of ϵeQ , where e is the proton charge and Q is the electromagnetic charge of a given fermion. As these results are well-known, we do not reproduce them here.

3.4 Low-Energy Parameters From the High-Energy Model

With the field content, mass spectra, and coupling terms for the model now determined, we have only one remaining task before being able to explore this setup's phenomenological implications: Properly identifying the parameters with which we might conduct a probe of the model space. Over the course of our development of the model in Section 2, we have noted that in spite of the large number of new particles present in the model at high energy, only a handful of these can be expected to have any significant effect at scales that can be experimentally probed now or in the near future. It stands to reason, then, that the large number of parameters in our model at high energy can in fact be reduced to a more manageable quantity at low energy. For the sake of clarity, we shall distinguish now between these two pictures: The "high-energy" model shall refer to the complete model with the field content outlined in Section 2. The "low-energy" model shall refer to the model only the fields of mass $\leq O(10 \text{ TeV})$, namely those which might have an observable effect on current and upcoming experiments, retained.

In Section 2.3, we found that the quark sector of the model is uniquely specified by the SM Higgs Yukawa-coupling parameters $y_H \sim O(1)$ and $\alpha \in [-\pi, \pi]$, the two portal matter Yukawa couplings

 $y_{P1} \sim O(1)$ and $y_{P2} \sim O(1)$, the vector-like mass term M, the scalar vev parameters $\vec{\gamma}_P$, $\vec{\gamma}_d$, v_P , and $y_S v_S^8$, and finally the signs of the eigenvalues of the matrices **u** (defined in Eq.(13)) and **d** (its analogous quantity in the down-like sector). In the limit where fields which we estimate to be too heavy to be experimentally observable decouple from the theory, our work simplifies substantially. Referencing our expressions for the masses and eigenvectors of the fermion fields in Section 2.3, we see that up to signs of various quantities (which we shall see do not affect any physical results up to numerically small corrections), we note that we can completely specify the physics of the accessible quark sector, namely the masses in Eqs.(18) and (22), their analogous values in the down-like sector, and the mass eigenvectors given in Eqs.(30), (31), (37), and (38), simply with the vev parameters v_P , $y_S v_S$, and $\vec{\gamma}_P$, the mass of the top quark partner M_t , and the masses of the accessible portal matter fields, $m_{P1}^u = m_{P1}^d$ and $m_{P2}^u = m_{P2}^d$.

Moving on to the gauge boson sector, we see that can enjoy a similar drastic reduction in independent parameters. Consulting Section 3.1, we see that the coupling matrices of Z_F , Z_P , and A_D , the three new gauge bosons that are light enough to potentially have experimentally observable effects, depend only on the CKM matrix, the mass eigenvectors of the fermions, the $SU(4)_F$ coupling constant g_4 , and the angle θ_P which functions as a Weinberg-like angle for the group $SU(4)_F \times U(1)_F$. Turning to Section 2.4, we note that while the mass of Z_P , given in Eq.(48), is entirely specified by the parameters g_4 , v_P , and θ_P , the masses of the gauge bosons Z_F and A_D arise as complicated functions of parameters which aren't otherwise relevant in the low-energy theory, namely the parameters $u_{1,2}$ (defined in Eq.(13)) and $d_{1,2,3}$ (their counterparts in the down-like quark sector), the Yukawa couplings y_A and y_B , and, in the case of A_D , the $U(1)_D$ -breaking vev components $\vec{\gamma}_d$. Because the mass of A_D depends on one set of parameters that the mass of Z_F doesn't, meanwhile, there is no obvious rigid relationship between these masses. It is therefore easier to simply specify the masses of Z_F and A_D as independent low-energy parameters in their own right, for the purposes of probing the phenomenology of the model.

Finally, we note that the above treatment has neglected to include any discussion of the leptons in the model. While we have noted earlier that the neutrino sector will lie largely unaddressed in this work, we cannot afford the same luxury with the charged leptons, so the fact that we are only introducing phenomenological parameters which cover the emergence of new physics in the quark sector bears some discussion. In the charged lepton sector, we note that we would require specifying some additional parameters, since we've posited that the charged lepton spectrum will be generated by the same mechanism as that of the down-like quarks, up to radiative corrections in the couplings to the scalars Φ_A , Φ_B , the parameters y_H and α , and the mass term M. Among the accessible elements of the theory, these radiative corrections would result in a modification of the rotation matrix \mathbf{W}_l (defined analogously to \mathbf{W}_{μ} is defined in the up-like sector in Eq.(13)), rather than it being precisely equal to \mathbf{W}_d , its counterpart in the down-like sector which must be approximately given by the CKM matrix. Referencing the couplings of various interactions in Section 3.1, we see that the matrix \mathbf{W}_l plays a role in the lepton-flavor-changing currents facilitated by the dark photon A_D , and the couplings of the accessible leptonic portal matter states to SM leptons (which in turn will govern the lifetimes and branching fractions of these new particles). However, we note that constraints on lepton-flavor-violating two-body decays of the τ lepton are much less restrictive than those in the quark sector [31], while the most restricted leptonic flavor-changing processes mediated by $A_D, \mu^- \to e^- A_D$, is essentially always kinematically disallowed in the range of dark photon masses that we consider $(m_{A_D} > 100 \text{ MeV})$.⁹ Meanwhile, discovery limits on the color-singlet leptonic portal matter will likely be far less constraining than those of the colored quark portal matter particles. As a result, we note that the new physics arising from additional parameters we must include to describe the charged lepton sector is likely beneath any notice, and we therefore have no need to expand our parameter space beyond what is necessary to specify the kinematically accessible fermions in the quark sector.

⁸For the purposes of this analysis, we have combined the coupling y_S with the vev parameter v_S of the singlet scalar Φ_S . Because the only instances of these parameters occurring separately happen when considering interactions of the physical scalar arising from Φ_S , a detailed analysis of which we have omitted here, this simplification is sufficient for our purposes.

⁹It should be noted that for lighter dark photon masses, the two-body decay of a muon to a dark photon and an electron would have a distinctive experimental signal, with a sharp peak in the electron energy spectrum. Limits on flavor violating decays in the quark sector remain more stringent than these constraints for now, but null results in searches for $\mu^+ \rightarrow e^+ X$ decays, where X is some undetected boson, from the upcoming Mu3e experiment can be expected to constrain this ratio to $\sim 10^{-8}$, which may allow it to begin to compete with current limits on the model from flavor-violating K meson decays [32].

Our full parameterization of the lower energy observable sector of the model then simply consists of two vev parameter v_P and $y_S v_S$, a three-component complex vector of vev's $\vec{\gamma}_P$, two gauge coupling parameters g_4 and θ_P , and five particle masses: the mass M_t of the heavy top partner, the masses $m_{P1}^u = m_{P1}^d$ and $m_{P2}^u = m_{P2}^d$ of the accessible portal matter fields, the mass m_{Z_F} of the "flavor Z" boson Z_F , and the mass m_{A_D} of the dark photon. For the reader's convenience, we have listed the parameters that are relevant in the high- and low-energy theories, as well as their approximate ranges, in several tables. Table 4 contains the parameters which are given in the underlying high-energy theory, and retained unchanged as independent parameters in our probe of the sector of the theory which is experimentally accessible, as well as the ranges that we have assumed here.

Table 5 contains the parameters which must be specified to generate the complete high-energy theory, but can be substituted for other parameters in the lower-energy theory. Finally, in Table 6, we present the parameters which may be used in lieu of those of Table 5 when probing the model parameter space where only phenomenologically accessible new particles are included. Notably, while there are simple expressions for $m_{P1}^{u,d}$ and $m_{P2}^{u,d}$ in terms of the parameters of Table 5 in Eq.(22), allowing for an easy estimate of the range these parameters might take on in our model, the natural ranges for the other three parameters in Table 6, namely the masses M_t , m_{A_D} , and m_{Z_F} , are not immediately obvious: There are no such compact expressions for these masses, at least in terms readily corresponding to those in Table 5. Since m_{A_D} is the only parameter which directly depends on the magnitude of $\vec{\gamma}_P$, rather than the ratio $\vec{\gamma}_P/v_P$, we find that by specifying v_P , $\vec{\gamma}_P$, and $\vec{\gamma}_d$ properly we can reproduce virtually any m_{A_D} between 0.1 and 1 GeV, so it is not unreasonable to treat this as a free parameter in this range, but the masses M_t and m_{Z_F} are more restricted. We can, however, produce estimates for the ranges of these parameters via a simple numerical probe of the high-energy parameter space.

Before beginning this exercise, however, we may simplify our task with a handful of observations. For m_{Z_F} , we see from Eqs.(40), (41), and (42) that the mass matrix which produces Z_F (namely, the first 8 rows and columns of the matrix given in Eq.(40), up to tiny $O(\vec{\gamma}_{P,d})$ corrections) should only have terms which depend on the matrices \mathbf{u} (as defined in Eq.(13)) and \mathbf{d} , its down-like quark counterpart - even dependence on the parameter M cancels out in this portion of the gauge boson mass squared matrix. Furthermore, inspection of Eq.(18), in particular the SM fermion mass expression which may be rewritten (in the up-like sector, with a corresponding expression applying in the down-like sector) as $u_i^2 \approx y_S^2 v_S^2 (y_u^2 v^2 / (2m_{ui}^2) - 1)$, that the eigenvalues $u_{1,2,3}$ and $d_{1,2,3}$ of **u** and **d** are all directly proportional to $y_S v_S$. That is, for all other model parameters held constant, the matrices **u** and **d** scale directly with $y_S v_S$. This in turn allows us to say that the mass matrix which governs m_{Z_F} , which in turn consists solely of combinations of **u** and **d**, must be directly proportional to $y_S v_S$, and therefore the mass m_{Z_F} is directly proportional to $y_S v_S$ as well. Meanwhile, a similar argument can be made regarding the mass of the top partner M_t , noting in Eq.(18) that we may rewrite the equation for the mass of this fermion as $M_t^2 = y_S^2 v_S^2 y_u^2 v^2 / (2m_t^2)$, and hence this mass is also directly proportional to $y_S v_S$. Therefore, when probing the masses m_{Z_F} and m_{A_D} , we only need to consider the range of these parameters at one specified value for $y_S v_S$ and extrapolate from there, and not probe the entire range we list in Table 4. Similarly, we need not probe the entire range of the $SU(4)_F \times U(1)_F$ gauge coupling parameters g_4 and θ_P , rather noting that m_{Z_F} , consisting up to tiny corrections entirely of $SU(4)_F$ bosons, is proportional to g_4 .

To perform our numerical examination, we next generate a sample of 10^5 points in parameter space (we note that larger samples grant us the same results for our parameter ranges, indicating that this sample size is likely large enough to fully probe this setup), specifying $y_S v_S = 1$ TeV, and selecting y_H , α , β , M, y_A , and y_B randomly (*i.e.*, with flat priors) in the ranges described in Table 5. We also impose several additional conditions on y_H , α , and β , the three parameters which govern the SM-like Higgs bidoublet sector. Specifically, we require that y_u and y_d , as defined in Eq.(11), are such that $y_u^2 > (2m_t^2)/v^2$ and $y_d^2 > 1/100$. The condition on y_u ensures that the SM quark spectrum can be reproduced using the relations of Eq.(18), while the condition on y_d ensures that it is not hierarchically small, which would in turn yield a heavy b partner with a mass roughly comparable to that of the heavy top partner, a region of parameter space that we have already established lies beyond the scope of this paper. To achieve a full sample of 10^5 points which meet the the aforementioned constraints on y_u and y_d , we generate y_H , α , and β , then re-generate these points (again with flat priors) if the original set fails them, until a point is found that meets the constraints; this in turn ensures that we uniformly sample the parameter space of y_H , α , and β that meet the y_u and y_d constraints without bias. Finally, we note that outside of numerically negligible corrections, the particular values of the parameters $\vec{\gamma}_P$, $\vec{\gamma}_d$, v_P , y_{P1} , and y_{P2} have no effect on the masses we wish to probe here, so there is no need to specify them. Given these conditions, we estimate the expected "natural" ranges of M_t and m_{Z_F} , as between the 5% and 95% quantile of their values in this scan – because we engage in a sampling of all of the parameters in our high energy theory with flat priors, we expect the results of our numerical probe to at least qualitatively reflect the necessary fine-tuning to effect certain values of these parameters. The results are then given in the ranges quoted in Table 6. We should note that although our sampling of the high-energy parameters is uniform over their ranges, the same cannot be said for the ranges of the parameters M_t and m_{Z_F} . To get a sense of the shapes of the M_t and m_{Z_F} distributions, we depict probability histograms of them in Figure 1. Consulting this Figure, we see that while M_t has a straightforward (if nonuniform) distribution in which smaller values of $M_t/(y_S v_S)$ are slightly favored over larger ones, the distribution of m_{Z_F} is somewhat more complex, with a large peak appearing near $m_{Z_F} \sim 30 g_4 y_S v_S$ followed by a steady drop-off. As a result of this long tail in the distribution, the range between the 5% and 95% quantiles of our sample given in Table 6 is unusually large. As we shall see in Section 4.2.2, however, the phenomenologically interesting region of parameter space lies in the region where $m_{Z_F} \lesssim 60 g_4 y_S v_S$, so we shall restrict our attentions there, merely noting that ample parameter space exists for significantly larger m_{Z_F} without significant fine-tuning.

$y_S v_S$	v_P	$ec{\gamma}_P$	g_4	θ_P
$1-10 { m TeV}$	$1-10 { m TeV}$	$0.1-1 { m GeV}$	(0.1-0.6)	$0 - \frac{\pi}{2}$

Table 4: The parameters which must be set to specify a unique point in parameter space in both the complete high-energy model and the low-energy model, as described in the text. Here, g denotes the electroweak coupling constant. The philosophy behind the chosen ranges of $y_S v_S$, v_P , and $\vec{\gamma}_P$ are discussed in Section 2.3, while we assume that $g_4 \leq g$ to keep the magnitude of the kinetic mixing $\leq O(10^{-3})$, as discussed in Section 3.3

y_H	α	β	y_A	y_B	y_{P1}	y_{P2}	$M/(y_S v_S)$	$ec{\gamma_d}$
$\pm [1/3,3]$	$[-\pi,\pi]$	$[0, \arctan(0.8)]$	$\pm [1/3, 3]$	$\pm [1/3,3]$	$\pm [1/3,3]$	$\pm [1/3, 3]$	$[-10^6, 10^6]$	0.1-1 GeV

Table 5: The parameters which must be set to specify a unique point in parameter space in the complete high-energy model, but which can be eliminated in favor of a simpler set of parameters in the low-energy model. The ranges quoted here assume that the Yukawa couplings are all of O(1) and β is restricted based on the requirement for perturbativity in the left-right-symmetric model Higgs sector. The magnitudes of M and $\vec{\gamma}_d$ are discussed in Section 2.3.

M_t	$m_{P1}^u = m_{P1}^d$	$m_{P2}^u = m_{P2}^d$	m_{A_D}	m_{Z_F}
$(1.1-3.2)y_S v_S$	$(1/3 - 3)v_P$	$(1/3-3)v_P$	$0.1-1 { m GeV}$	$g_4 y_S v_S (17 - 210)$

Table 6: The parameters which which may be used to specify a unique point in the parameter space of the low-energy model in lieu of the high-energy model parameters in Table 5. The large range of m_{Z_F} , and the shape of the m_{Z_F} distribution in the numerical probe, is discussed in the text and Figure 1.



Figure 1: (Top) A probability histogram of the values of m_{Z_F} in units of $g_4 y_S v_S$ obtained in our numerical sample, described in the text. The green region denotes the values of m_{Z_F} between the 5% and 95% quantiles. (Bottom) Same as above, but the probability histogram now depicts values of $M_t/(y_S v_S)$.

4 Phenomenology: Flavor-Changing Neutral Currents

Having set up the model and determined the relevant parameters we can begin to analyze the sector of the model within experimental reach, the sole remaining task left to us is to actually do the necessary exploration. To begin, we note that our significant expansion of the model's flavor sector has resulted in substantial additional sources of flavor-changing couplings, in particular flavor-changing neutral currents (FCNC's), which have the potential to contribute to highly constrained processes. To examine these flavor-changing effects, then, we explore the impact of the model's FCNC's in two sectors: rare meson decays and neutral meson oscillation.

4.1 Flavor-Changing A_D Interactions

We begin our discussion of potential phenomenological effects with a brief foray into the most constrained couplings which emerge here, namely the light flavor-changing neutral currents mediated by the dark photon. In particular, the tiny flavor-changing couplings among the SM quarks in Eq.(52) allow for $b \to s + A_D$, $b \to d + A_D$, and $s \to d + A_D$ transitions. As in **II**, we shall assume that the on-shell longlived light dark photon will escape our detector and/or decay to DM, and therefore these interactions can facilitate meson decays which mimic to some extent the highly-constrained rare decay channels $B \to K \nu \bar{\nu}$, $B \to \pi \nu \bar{\nu}$, and $K \to \pi \nu \bar{\nu}$.

It should be noted that the naive bounds from the three-body $B \to K \nu \overline{\nu}, B \to \pi \nu \overline{\nu}$, and $K \to \pi \nu \overline{\nu}$ are not precisely applicable to the two-body decay of a B and K mesons to a lighter meson and a longlived dark photon: The latter will result in a sharply peaked momentum distribution for the visible light meson in the final state. This difference has a significant effect on our analysis of $K \to \pi A_D$ constraints in particular: Searches in the "golden channel" $K^+ \to \pi^+ \nu \overline{\nu}$, which are normally highly constraining, are substantially weakened in cases where $A_D \sim m_{\pi}$ due to high backgrounds from the $K^+ \rightarrow \pi^+ \pi_0$ channel [33–35]. Instead, it is more instructive to consider the constraints from $K_L \to \pi_0 X$ searches such as [35], where X is simply some light long-lived invisible particle, which are not subject to the same limitations. Specifically, to derive for our model's constraints from $K_L \to \pi_0 X$ branching fractions, we compute the branching fraction for this decay arising from A_D emission and compare it to the 90% CL upper limits from [35] obtained for various mass values. In the case of the other flavor-changing decays we consider, a more careful analysis of $B^+ \to \pi^+, K^+ \nu \overline{\nu}$ searches, which lack the same troubling kinematic cuts as appear in $K^+ \to \pi^+ \nu \overline{\nu}$ searches, doesn't result in significant differences from the constraints derived by simply quoting the 90% CL upper bounds of $\mathcal{B}(B^+ \to \pi^+ \nu \overline{\nu}) \leq 0.8 \times 10^{-5}$ [36] and $\mathcal{B}(B^+ \to K^+ \nu \overline{\nu}) \leq 1.6 \times 10^{-5}$ [37] as direct limits on the corresponding flavor-changing decays mediated by A_D (although it should be noted that we should treat these constraints as order-of-magnitude, not precise, limits barring a more detailed analysis of the final-state light meson spectrum). Since we shall see that these constraints are much less rigorous than those which are derived from the $K_L \rightarrow \pi_0 A_D$ system, we also note that the changes in restrictions on the parameter space from O(1) changes in these constraints are not significantly large.

A straightforward calculation employing the coupling constants in Eq.(52) gives us the decay width of the process $K_L \rightarrow \pi_0 A_D$

$$\Gamma_{K_L \to \pi_0 A_D} = g_4^2 s_P^2 \frac{|\xi_1|^2 |\xi_2|^2}{24\pi} \frac{m_K^3}{m_{A_D}^2} \left(1 + \frac{(m_{A_D}^2 - m_\pi^2)^2}{m_K^4} - \frac{2(m_{A_D}^2 + m_\pi^2)}{m_K^2} \right)^{3/2} |f_+^{K^0 \pi^0}|^2, \quad (67)$$

$$\xi_i \equiv \frac{(\mathbf{W}_d \vec{\gamma}_P)_i}{v_P}$$

where m_K and m_{π} are the masses of the neutral K and π mesons, respectively, $f_+^{K^0\pi^0}$ is a hadronic form factor we can extract from [38] (up to percent level corrections, this factor is equal to 1 for the process $K_L \to \pi_0 A_D$), and we remind the reader that, also up to percent level corrections, \mathbf{W}_d is equal to the CKM matrix. Expressions for $B^+ \to K^+ A_D$ and $B^+ \to \pi^+ A_D$ decays can be easily determined by replacing the appropriate indices of ξ and meson masses in Eq.(67), and extracting the appropriate



Figure 2: Maximum values of the products $\xi_1\xi_2$ (top left), $\xi_1\xi_2$ (top right), and $\xi_2\xi_3$ (bottom) based on the 90% CL measured limits on $\mathcal{B}(K_L \to \pi_0 X)$ (where X is some invisible long-lived particle) [35], $\mathcal{B}(B^+ \to \pi^+ \nu \overline{\nu})$ [36], and $\mathcal{B}(B^+ \to K^+ \nu \overline{\nu})$ [37], respectively. Benchmark values of $g_4 s_P$ are taken as 0.1 (blue), 0.2 (red), and 0.3 (magenta). The ξ parameters are described in the text.

hadronic form factors from [39]. Consulting the definition of ξ_i in Eq.(67), we see that these decay processes should naturally undergo substantial suppression, since $\vec{\gamma}_P \sim 0.1 - 1$ GeV and $v_P \gtrsim 1$ TeV. We might expect, therefore, that a natural value of ξ_i would be in the realm or 10^{-4} , leading to a $\sim 10^{-16}$ suppression of these decay processes, as was noted in **II**. In spite of this suppression, we shall find, as in **II**, that constraints on these decays are severe enough to provide meaningful limits on the values of the ξ_i 's beyond even their naturally small magnitudes.

We can straightforwardly derive expressions for the maximum allowed magnitudes for products of ξ_i values, upon which the constrained meson decays depend, for various representative choices of the parameters m_{A_D} and g_{4s_P} by comparing our results to upper experimental limits on the branching ratios of these mesons to the corresponding two-neutrino final states (or in the case of $K_L \to \pi_0 A_D$, the mass-dependent limits on the branching ratios $\mathcal{B}(K_L \to \pi_0 X)$). The maximum values of the relevant components of $(\mathbf{W}_d^{\dagger} \vec{\gamma}_P)/v_P$ as functions of m_A at various values of g_4s_P are depicted in Figure 2.

From Figure 2, we can derive some approximate constraints on the parameters ξ_i , which suggest that some of their magnitudes must be well below the natural value of $\sim O(10^{-4})$ that we might expect. Up to O(1) variations as depicted in this Figure, we arrive at the rough constraints

$$\xi_1\xi_2 \lesssim 10^{-11}, \quad \xi_1\xi_3 \lesssim 10^{-8}, \quad \xi_2\xi_3 \lesssim 10^{-8}, \tag{68}$$

where the first (and most stringent) constraint, coming from $K_L \to \pi_0 A_D$, only applies when $m_{A_D} \lesssim 260$ MeV, past which the study of [35] doesn't offer constraints due to overly large $K_L \to \pi^+\pi^-\pi_0$ backgrounds. The much less severe constraints from the other decay processes have no such similar restriction on the dark photon mass for which they apply; these processes remain kinematically accessible over the entire range of m_{A_D} we consider. Because the constraints in Eq.(68) only apply to the products of ξ_i terms, there is a fair amount of flexibility in how these constraints might be satisfied. For example, we might meet the harsh constraint on the product of $\xi_1\xi_2$ by allowing one of either ξ_1 or ξ_2 to be ~ 10⁻⁷, several orders of magnitude below its expected magnitude of ~ 10⁻⁴, while the other stays at the natural higher magnitude, or we can require that $\xi_1 \sim \xi_2 \sim 3 \times 10^{-6}$. Given that the other constraints here are or comparatively little significance, we can outline several phenomenologically allowed benchmark values of the parameters ξ_i , listed in Table 7. An astute reader may be concerned that, in suppressing various ξ_i (and hence $\vec{\gamma}_P$) values to such low values, we may anticipate that contributions to fermion mixing from $\vec{\gamma}_d$, which we have neglected in our treatment of the low-energy phenomenology, may be significant. However, in practice we find that even assuming all components of ξ_i are as small as $O(10^{-10})$, the $\vec{\gamma}_P$

Benchmark	ξ_1	ξ_2	ξ_3
A1	$\lesssim 10^{-4}$	0	0
A2	0	$\lesssim 10^{-4}$	0
A3	0	0	$\lesssim 10^{-4}$
B1	$\lesssim 10^{-4}$	0	$\lesssim 10^{-4}$
B2	0	$\lesssim 10^{-4}$	$\lesssim 10^{-4}$
C1	$\lesssim 10^{-4}$	$\lesssim 10^{-4}$	$\lesssim 10^{-4}$
C2	$\lesssim 10^{-6}$	$\lesssim 10^{-6}$	$\lesssim 10^{-6}$

Table 7: The different benchmarks for ξ_i magnitudes which can satisfy the rough constraints of Eq.(68). Scenarios A1, A2, and A3 correspond to cases where one ξ_i is hierarchically larger than the others, with the number denoting which generation, first, second, or third, has the dominant ξ_i . Scenario B1(B2) assumes that either $\xi_1(\xi_2)$ is roughly equal in magnitude to ξ_3 , while $\xi_2(\xi_1)$ is hierarchically smaller. Scenario C1, in which all ξ_i 's are approximately equal, is only viable if $m_{A_D} \gtrsim 250$ MeV, and hence the decay $K_L \to \pi_0 A_D$ is either hidden by $K_L \to \pi^+ \pi^- \pi_0$ backgrounds or is kinematically inaccessible; scenario C2 is the equivalent of C1 in the event that $m_{A_D} \lesssim 250$ MeV.

Before moving on, we note that, as mentioned in Section 2.2, we expect the model should contain a physical scalar of mass ~ m_{A_D} , *i.e.*, the 'dark Higgs', which should generically have similar flavorchanging couplings as the dark photon. If we likewise assume that the light scalar is long-lived or can primarily only decay to dark photons as we would expect, we might anticipate that it will make additional contributions to the branching fractions we have discussed in this section, of comparable magnitude to those facilitated by the dark photon. However, given the complexity of the scalar sector within this model, computing the exact magnitude of these contributions is highly non-trivial, and will likely depend on a number of additional parameters in the scalar potential. For the sake of simplicity, then, we shall not compute the light scalar contributions to these flavor-changing decays. We can assume that at worst, these likely manifest O(1) corrections to the constraints appearing in Figure 2, and given the highly non-trivial nature of the scalar sector, it is just as plausible that large regions of parameter space in the scalar potential exist such that these decays are either kinematically forbidden or highly suppressed by a small phase space factor.

Finally, it should be noted that similar transitions exist in the up-like quark and charged lepton sectors $(e.g., t \rightarrow cA_D, c \rightarrow uA_D, \mu \rightarrow eA_D)$. However, in the case of up-like quarks the experimental constraints on decays mediated by these processes are much weaker, while the case against including restrictions in the leptonic sector was discussed in Section 3.4. In either case, these processes provide no significant

limit on our model parameters at present.

4.2 Neutral Meson Oscillation

While the dark photon's flavor-changing couplings are capable of facilitating distinctive and highlyconstrained meson decays, they are hardly the only potentially dangerous flavor-changing couplings in the theory. In particular, we find that $\overline{B}_d - B_d$ and $\overline{B}_s - B_s$ oscillations can suffer significant contributions to their dispersive amplitudes, given by $(M_{12})_q = \langle \overline{B}_q^0 | \mathcal{H}_{eff}^{\Delta B=2} | B_q^0 \rangle / (2m_{B_q})$ (where $\mathcal{H}_{eff}^{\Delta B=2}$ is simply the effective Hamiltonian for the relevant flavor-changing interactions) from two additional sources: The flavor-changing Z_F couplings, identified in Eq.(56), and the one-loop contributions of the heavy top partner to the box diagram which generates meson mixing in the SM. We note that effects in $\overline{K}^0 - K^0$ mixing, by contrast, are generally quite muted: The contribution to this process due to the top partner loops is suppressed by CKM factors, while the corresponding FCNC's mediated by Z_F vanish at $O(\lambda^4)$. To obtain our results, we shall compute the new physics contributions to $(M_{12})_q$ and parameterize them as [27, 40, 41]

$$(M_{12})_q = (M_{12})_q^{SM} (1 + h_q e^{2i\sigma_q}), \tag{69}$$

where $(M_{12})_q^{SM}$ is the SM contribution to B_q meson mixing, while h_q and σ_q are real parameters for the contribution of new physics to these processes. $(M_{12})_q^{SM}$ is in turn given by [42]

$$(M_{12})_{q}^{SM} \approx \frac{G_{F}^{2} m_{W}^{2}}{12\pi^{2}} f_{B_{q}}^{2} m_{B_{q}} B_{B_{q}} (\lambda_{t}^{(q)})^{2} S_{0}(x_{t}),$$

$$S_{0}(x_{t}) \equiv \left(\frac{x_{t}(4 - 11x_{t} + x_{t}^{2})}{4(1 - x_{t})^{2}} - \frac{3x_{t}^{3} \log(x_{t})}{2(1 - x_{t})^{3}}\right),$$

$$\lambda_{i}^{(q)} \equiv V_{ib}^{*} V_{iq}, \quad x_{i} \equiv \frac{(m_{i}^{\text{pole}})^{2}}{m_{W}^{2}},$$
(70)

We can then compare the new physics contributions to the limits in a fit of CKM observables to new physics in the meson mixing sector in [27] in order to extract approximate constraints on our model parameters. In particular, this fit gives

$$h_d \le 0.26, \ h_s \le 0.12$$
 (71)

at the 95% CL level as of Summer 2019. We note that simply referring to these ranges represents only a very approximate assessment of the constraints afforded by flavor-changing neutral currents in this model; for example, the above ranges assume uncorrelated h_d and h_s values, which we shall see is not the case here, and ignore non-trivial constraints on the phase σ_d , which we shall see may substantially tighten constraints on h_d . In spite of these limitations, however, we find that the limits in Eq.(71) will afford us an approximate picture of the effect our model has on meson oscillation parameters.

4.2.1 Neutral Meson Oscillation: Top Partner Loop Contribution

In spite of being a loop-level interaction, the contribution of the heavy top quark partner to neutral meson mixing processes, in particular those of the B_s and B_d mesons, is significant enough to warrant discussion, and as its computation can be performed in direct analogy with the SM contribution to these oscillation processes, we shall begin our discussion here. Before continuing, we note that the new physics contributions to meson mixing from the top partner take two forms: Additional loop diagrams featuring the heavy top quark, and, as can be seen in Eq.(60), tree-level modifications of the SM CKM matrix elements V_{td} , V_{ts} , and V_{tb} . As discussed in [27], the fit performed in that work extrapolates these CKM matrix elements from unitarity, rather than directly, and as such the tree-level modifications to these parameters must be included as "new physics" that contributes to $h_{d,s}$ and $\sigma_{d,s}$. We can now determine the contribution of the top partner here in direct analogy to the SM calculation done in, for example, [42].

Inserting Eq.(60) for the CKM couplings of the top partner, we find that up to $O(10^{-2})$ corrections due to subdominant loops featuring up and charm quarks, the top partner's contribution to $(M_{12})_q$ is

$$\frac{(M_{12})_q^{\mathrm{T}}}{(M_{12})_q^{\mathrm{SM}}} = h_q^T e^{2i\sigma_q^T} \approx \frac{r^2}{S_0(x_t)} \frac{m_t^2}{M_t^2} (-2\tilde{S}(x_t, x_t) + 2\tilde{S}(0, x_t) + 2\tilde{S}(x_t, x_T) + r^2 \frac{m_t^2}{M_t^2} \tilde{S}(x_T, x_T)), \quad (72)$$

$$\tilde{S}(x_i, x_j) = \frac{4 - 7x_i x_j}{4(1 - x_i - x_j + x_i x_j)} + \frac{(4 - (8 - x_i)x_j)x_i^2 \log(x_i)}{4(x_i - 1)^2(x_i - x_j)} + \frac{(4 - (8 - x_j)x_i)x_j^2 \log(x_j)}{4(x_j - 1)^2(x_j - x_i)}, \quad (72)$$

$$x_T \equiv \frac{M_t^2}{m_W^2}, \quad x_t \equiv \frac{(m_t^{\mathrm{pole}})^2}{m_W^2},$$

where we have assumed that the QCD corrections for the heavy top partner loops are approximately equal to those of the top quark itself (as done in, for example, [43]), $S_0(x_t)$ is the Inami-Lim function given here in Eq.(70), and we remind the reader that r is the variable first defined in Eq.(30). It is useful to point out that here, as elsewhere in this work, when we refer to an SM quark mass such as m_t , we are explicitly referring to its value at a scale of ~ 1 TeV (as we noted in Section 2.3), the scale at which we assumed the SM fermion masses were generated via a seesaw mechanism. Hence, m_t in Eq.(72) refers to the top quark mass at this scale, or in this case $m_t(\sim 1 \text{ TeV}) \approx 144 \text{ GeV}$, and not its pole mass, which is explicitly denoted by $m_t^{\text{pole}} \approx 173 \text{ GeV}$. As both the pole mass m_t^{pole} and the RG-evolved mass m_t appear in Eq.(72), some care must be taken in order to use the formula correctly. Since the scale 1 TeV is dramatically closer to the top partner pole mass than it is to that of the top quark, we neglect any effects of RG running in M_t . In the limit where $x_T \gg x_t$, which applies here, we can combine Eqs.(70) and (72) as

$$h_{d,s}^{T} \approx \frac{r^{2}x_{t}}{2S_{0}(x_{t})} \frac{m_{t}^{2}}{M_{t}^{2}} \left[\frac{6x_{t}}{(-1+x_{t})^{2}} - \frac{3(-1+3x_{t})}{(-1+x_{t})^{3}} \log(x_{t}) - 1 + \frac{r^{2}}{2} \left(\frac{m_{t}}{m_{t}^{\text{pole}}} \right)^{2} - 2\log\left(\frac{m_{t}^{\text{pole}}}{M_{t}}\right) \right], \quad (73)$$
$$\sigma_{d,s}^{T} = 0.$$

Inserting $m_t^{\text{pole}} \approx 173 \text{ GeV}, m_t \approx 144 \text{ GeV}$, and $x_t \approx 4.63$, we arrive at

$$h_{d,s}^{T} \approx 0.064 \left(\frac{1 \text{ TeV}}{y_{S} v_{S}}\right)^{2} \left(1 - \frac{y_{S}^{2} v_{S}^{2}}{M_{t}^{2}}\right) \left[1 + 0.102 \left(\frac{M_{t}^{2}}{y_{S}^{2} v_{S}^{2}} - 1\right) - 0.592 \log\left(\frac{1 \text{ TeV}}{M_{t}}\right)\right],\tag{74}$$

where for clarity we have explicitly inserted the equation $r^2 = M_t^2/(y_S^2 v_S^2) - 1$ in the above expression. Notably, we see that the heavy top partner contributes the same correction to both B_d and B_s oscillation, and also that this correction can be quite significant, even of O(10%). Obtaining a realistic range for $M_t/(y_S v_S)$ from Table 6, we can then consider the correction $h_{d,s}^T$ as a function of the parameters $y_S v_S$ and $M_t/(y_S V_S)$. In the case of h_d , the top partner loops represent the dominant new physics contribution to this parameter, so our computation of this variable can be completed here. We depict the results for h_d for various values of $y_S v_S$ and $M_t/(y_S v_S)$ in Figure 3 (we remind the reader that $M_t > y_S v_S$, which can be straightforwardly seen from the equation for $M_t^2 = M_{u3}^2$ in the last line of Eq.(18)).

From Figure 3, we can see several interesting characteristics of this new physics contribution. First, comparing the magnitude of the effects in this Figure to the constraints in Eq.(71), we see that for the region of parameter space we are considering, the effects of the top partner loops on meson mixing parameters should lie within current experimental constraints. We do, however, note that for regions with the largest effect, which can reach $h_d \sim 0.14$, the results exceed the expected sensitivity of LHCb and Belle-II data in the coming years, which may be sensitive to $h_d \sim 0.07$ within the next decade [27]. Furthermore, we note that a quick reference to Eq.(72) allows us to see that $\sigma_d = 0$ from this contribution (indicating a *positive* proportional contribution to $(M_{12})_d$), while the best-fit value of σ_d in the fit of [27] is $\sigma_d = -1.40^{+0.97}_{-0.23}$ (or roughly corresponding to a *negative* proportional contribution to the mixing matrix- see [40] for a full chart of allowed values in the $h_d - \sigma_d$ plane, albeit with slightly outdated data but exhibiting similar favored regions of $h_d - \sigma_d$ as the present constraints). We might therefore expect that constraining our model more rigorously, with a CKM fit of our own, might result in non-trivial



Figure 3: (Top) Value of h_d as a function of $y_S v_S$ for, from top to bottom, $M_t/(y_S v_S) = 3, 2.5, 2, 1.5,$ and 1.1, respectively. (Bottom) Value of h_d as a function of $M_t/(y_S v_S)$ for, from top to bottom, $y_S v_S = 1, 1.5, 2, 2.5, and 3$ TeV, respectively.

constraints on the scenarios with low $y_S v_S$ values from measurements of B_d oscillation. However, the sharp dropoff of these contributions as $y_S v_S$ increases likely removes this sensitivity as $y_S v_S \gtrsim 1.75$ TeV, unless these constraints are an order of magnitude more sensitive than the generic boundaries found in [27]. In practice, therefore, we find that the loop-level contributions of the top partner to neutral meson mixing are presently not subject to any substantial constraints; however additional data from LHCb and Belle-II in the coming years will likely constrain the parameter space near the lower end of the range $y_S v_S \sim 1$ TeV from effects in this sector.

4.2.2 Neutral Meson Oscillation: Z_F Contribution

While the top partner loops represent by far the dominant effect in the B_d -meson mixing system, the B_s mixing system suffers additional contributions from another source, *i.e.*, the "flavor Z" boson, Z_F . In particular, we can see from the coupling in Eq.(56) that the dominant contribution of the Z_F boson will be to B_s mixing alone. We can compute this contribution straightforwardly by following the treatment of, *e.g.*, [44]. In this case, we can write the effective flavor-changing Hamiltonian as

$$\mathcal{H}_{Z_F}^{\Delta B=2} = \mathcal{C}_1(m_{Z_F})\mathcal{O}_1(m_{Z_F}) + \tilde{\mathcal{C}}_1(m_{Z_F})\tilde{\mathcal{O}}_1(m_{Z_F}) + \mathcal{C}_4(m_{Z_F})\mathcal{O}_4(m_{Z_F}) + \mathcal{C}_5(m_{Z_F})\mathcal{O}_5(m_{Z_F}),$$
(75)

with

$$\mathcal{O}_1 = (\bar{b}_L^{\alpha} \gamma_\mu q_L^{\alpha}) (\bar{b}_L^{\beta} \gamma_\mu q_L^{\beta}), \quad \tilde{\mathcal{O}}_1 = (\bar{b}_R^{\alpha} \gamma_\mu q_R^{\alpha}) (\bar{b}_R^{\beta} \gamma_\mu q_R^{\beta}), \quad \mathcal{O}_4 = (\bar{b}_R^{\alpha} s_L^{\alpha}) (\bar{b}_L^{\beta} s_R^{\beta}), \quad \mathcal{O}_5 = (\bar{b}_R^{\alpha} s_L^{\beta}) (\bar{b}_L^{\beta} s_R^{\alpha}), \quad (76)$$

using operators defined identically to those of [44]. Our computation of $(M_{12})_s$ stemming from these effects, then, simply requires identifying the Wilson coefficients C_i and extracting the expectation values of the operators \mathcal{O}_i from [44]. Using Eq.(56), we can straightforwardly see that at some high scale μ_{high} , we have

$$\mathcal{C}_{1}(\mu_{\text{high}}) = \tilde{\mathcal{C}}_{1}(\mu_{\text{high}}) = \frac{3g_{4}^{2}}{4m_{Z_{F}}^{2}}A^{2}\lambda^{4}(\rho + i\eta)^{2}, \quad \mathcal{C}_{4}(\mu_{\text{high}}) = 0, \quad \mathcal{C}_{5}(\mu_{\text{high}}) = -\frac{3g_{4}^{2}}{m_{Z_{F}}^{2}}A^{2}\lambda^{4}(\rho + i\eta)^{2}. \quad (77)$$

We can then run these coefficients down to hadronic scales using anomalous dimension matrices which can be extracted from [45]. For the sake of simplicity, we shall assume that $\mu_{\text{high}} = 10$ TeV, which should be within a factor of ~ O(a few) from m_{Z_F} , consulting the range for this parameter given in Table 6 and assuming $g_4 \sim 0.3$, $y_S v_S \sim 1$ TeV. We also assume, consulting the likely ranges for these masses in Table 6, that at the scale ~ 10 TeV, only five non-SM quark flavors are dynamical: The top partner, two up-like portal matter quarks, and two down-like portal matter quarks, and that for the purposes of our RGE calculation all five of these extra flavors can be integrated out at the same scale, which we take to be 1 TeV. Running the coupling constants of Eq.(77) down to the *b* quark mass, extracting the expectation value of the operators $\mathcal{O}_{1,4,5}$ (and the bag parameter as well as the decay constant from the SM contribution to B_s oscillation given in Eq.(70)) from [44], we arrive at

$$\frac{(M_{12})_s^{Z_F}}{(M_{12})_s^{SM}} = h_s^{Z_F} e^{2i\sigma_s^{Z_F}} \approx (0.024) e^{2i(-0.099)\pi} \left(\frac{10 \text{ TeV}}{m_{Z_F}}\right)^2 \left(\frac{g_4}{0.3}\right)^2.$$
(78)

We note that while this correction has a magnitude which varies depending on the $SU(4)_F$ coupling g_4 and the mass m_{Z_F} , its phase remains fixed. This is the result of the coupling constants in Eq.(77) all having the same complex phase, that of $\rho + i\eta$, which is of course fixed by the Wolfenstein parameters. The result of Eq.(78) suggests, as in the case of the loop-level corrections due to the top partner, that we may anticipate roughly O(a few%) corrections from Z_F exchange: That is, these two processes have roughly comparable effects. We should, then, estimate total contribution to h_s as

$$h_{s} = |h_{s}^{Z_{F}} e^{2i\sigma_{s}^{Z_{F}}} + h_{s}^{T} e^{2i\sigma_{s}^{T}}|,$$
(79)

with $h_s^{Z_F}$ and $\sigma_s^{Z_F}$ given by Eq.(78), h_s^T given by Eq.(74), and $\sigma_s^T = 0$, as noted in Section 4.2.1. Taking numerical results of Eq.(79) and comparing the results to the constraints on h_s in Eq.(71), we can also



Figure 4: The full value of h_s , including tree-level Z_F and top partner loop contributions, for $m_{Z_F}/(g_4y_Sv_S) = 20$ (blue), 40 (red), and 60 (magenta). These plots assume that $M_t/(y_Sv_S) = 3$ (top left), 2.5 (top right), 2 (bottom left), and 1.5 (bottom right). For comparison, the 95% CL limit on h_s is depicted in each plot as a dashed orange line.

extract some rough constraints on the parameters $y_S v_S$ for various choices of the proportional mass parameters $M_t/(y_S v_S)$ and $m_{Z_F}/(g_4 y_S v_S)$. In Figure 4, we depict the combined effect of the Z_F and top partner contributions to the variable h_s .

Comparing the results of Figure 4 to those of Figure 2, which should be identical to h_s^T , or the new physics parameter in the absence of any Z_F contributions, we see that for lower masses of Z_F , in particular $m_{Z_F} = 20g_4y_Sv_S$, the lower end of the range we consider, the effect of the Z_F boson on h_s is significant, on the order of 0.06 for $y_Sv_S = 1$ TeV. For lower values of $M_t/(y_Sv_S)$, the Z_F contribution at these low masses is even the same magnitude as the contribution due to vector-like quarks. However, some caveats are required here. First, a Z_F mass as low as $20g_4y_Sv_S \sim 7y_Sv_S$ is the result of somewhat finely-tuned parameters in the high-energy theory: Our numerical probe of the high-energy parameter space in Section 3.4 places such a mass only just above the 5% quantile. Larger values of m_{Z_F} , such as $60g_4y_Sv_S \sim 20y_Sv_S$, which is close to the median m_{Z_F} of our sample, generally offer only an O(10%)correction to the mixing result purely from considering top partner loops. For larger values of m_{Z_F} , which can reach as high as $200g_4y_Sv_S \sim 70y_Sv_S$ in our sample, the effects of the Z_F boson quickly become negligible. We also note that, as discussed in Section 2.4, the Z_F boson's mass may be sensitive to modifications in the model to accomodate the SM neutrino sector. In particular, the mass of any gauge bosons corresponding to the $SU(3)_F$ flavor group may increase dramatically. As a result, we might see any observable new physics contribution from Z_F vanish.

These concerns aside, however, we do see in Figure 4 that any constraint from B_s meson mixing is quite mild – while there do exist points in parameter space which can exceed the 95% CL upper bound for $h_s \simeq 0.12$, we see that they tend to only occur in the "optimal" scenario for these corrections: The only scenarios we depict in which the present h_s cannot be evaded by simply requiring $y_S v_S \gtrsim 1.1$ TeV, a mere 10% increase from the lowest value of $y_S v_S$ we consider, occur when we have assumed $m_{Z_F} = 20g_4 y_S v_S$, near its 5% quantile value in our numerical sample, and $M_t \geq 2.5 y_S v_S$. As in the case of B_d mixing, however, we can note that future improvements to h_s observations from Belle-II and LHCb measurements (and improvements in lattice computations of hadronic matrix elements), which might be expected to reduce the 95% CL bound on h_s to ~ 0.06 [27], may begin to place more meaningful constraints on our model parameter space for the Z_F and top partner masses.

5 Phenomenology: Portal Bosons and Exotic Matter

Having discussed the low energy flavor-violating implications of the model, we can turn to some other simple processes in the model, in particular focusing on where our results differ from those of I and II. We shall organize this discussion by considering each of the likely kinematically accessible heavier exotic particles, the Z_P , top partner, and the portal matter separately.

5.1 Phenomenology: Portal Matter Fermions

We shall begin our discussion of the phenomenology of our fields with the portal matter: The fermion fields which possess non-trivial $U(1)_D$ charge. From Section 2.3, we note that our theory anticipates two such fields for each sector of the theory (up-like quarks, down-like quarks, charged leptons, and neutrinos) that we can expect to have masses potentially accessible to experiment in the foreseeable future. For convenience, we shall label the two portal matter states based on their masses: P_1^u , for example, will be the up-like portal matter quark with mass m_{P1}^u (as given in Eq.(22)), while P_2^u will be the up-like portal matter quark with mass m_{P2}^u (again given in Eq.(22)). Analogous definitions exist for the portal matter in the down-like quark and leptonic sectors. For the sake of simplicity, and because the constraints on color triplet new particles are dramatically stronger than those for color singlets, we shall limit our discussion here to the portal matter in the quark sector, $P_{1,2}^{u,d}$.

Much of the behavior of the portal matter phenomenology in the present setup is analogous to that of I and II, and as such does not bear particular repeating. However, the additional structure present in this model does introduce some non-trivial complications to the existing frameworks discussed there. In particular, the couplings of the portal matter states to various generations of SM matter are entirely controlled by the orientation of the vector $\vec{\gamma}_P$ in three-dimensional flavor space: Couplings of i^{th} -generation up-like quarks are proportional to $(\vec{\gamma}_P)_i/v_P$, while couplings of the i^{th} -generation down-like quarks are proportional to $\xi_i = (\mathbf{W}_d^{\dagger} \tilde{\gamma}_P)_i / v_P$, where we remind the reader that \mathbf{W}_d is well-approximated by the CKM matrix. Since the CKM matrix is nearly diagonal, this in turn limits the freedom with which we may couple different quark portal matter states to various generations: If the benchmark A1 is taken from the ξ_i benchmarks in Table 7, for example, the branching fractions of the up-like portal matter to the second- and third-generation up-like quarks will be suppressed by CKM factors of $\leq O(10^{-2})$. Finally, we note that the universal dependence on $\vec{\gamma}_P$ to generate couplings between portal matter and SM states limits the freedom with which discrepancies may appear between the mixing of P_1 states with the SM and that of P_2 states: With the exception of the third generation of the up-like sector, where corrections occur due to mixing between the SM top quark and its non-portal partner, the magnitude of P_1 and P_2 couplings to SM states are the same, with P_1 states coupled to left-handed $(SU(2)_L \text{ doublet})$ SM states and P_2 states coupled to right-handed $(SU(2)_L \text{ singlet})$ ones. So, for example, the branching fractions of P_1^d and P_2^d portal matter particles to each of b, s, and d quarks are always equal. Meanwhile, the fact that the P_1 states overwhelmingly couple to $SU(2)_L$ doublet SM fields and the P_2 overwhelmingly couple to $SU(2)_L$ singlet fields is unsurprising: Given that the P_1 states are $SU(2)_L$ doublets, mixing between P_1 states and right-handed $(SU(2)_L$ singlet) fields violates $SU(2)_L$, and so only occurs due to mass terms in Eqs.(14) and (32) proportional to the $SU(2)_L$ -breaking vev, $v/\sqrt{2} \sim 170$ GeV. Mixing between P_1 states and left-handed fermions, however, has no such restriction, and can be proportional to the significantly larger vev terms v_S and v_P . An analogous argument applies for P_2 states preferring to couple to $SU(2)_L$ singlet fields over doublets.

Regarding the production of portal matter states, the QCD pair production cross section of these fermions will be the same as the results already given I and II for 13 TeV and 14 TeV center-of-mass energy at the LHC, respectively. Additionally, as in II, there exist potentially significant non-QCD contributions to the $q\bar{q} \rightarrow \bar{P}^u_i P^u_i$ process from t-channel dark photon exchange. We note that the dark photon exchange contribution has one piece of additional structure compared to that of II: Because all three generations may mix with the portal matter states, depending on the orientation of the vector $\vec{\gamma}_P$. the contribution of the t-channel A_D exchange to portal matter pair production may stem from mixing with multiple generations at once. However, we also note that the effect of this exchange varies radically depending on the generation in question, with mixing with lighter quarks having a dramatically stronger effect than mixing with heavier quarks, due to the greater content of the lighter quarks in proton parton distribution functions. We can then see, consulting the results of II for the down-like sector of our model, that restricting considerations of mixing in this contribution to just that of the lightest generation for which the mixing is non-trivial (in the language of Section 4.1, the ξ_i for the lowest index i that isn't hierarchically smaller than any other ξ_i) will likely be accurate to within O(10%) corrections, at worst. Given the even larger hierarchies among quark masses in the up-like sector, we anticipate this approximation to be even more accurate for up-like portal matter.

Finally, we can discuss the decay branching fractions of portal matter fields to SM states. As in **I** and **II**, the dominant decay process is the emission of a dark photon to decay into an SM state with the same SM quantum numbers as the portal matter field. Consulting the couplings in Eqs.(50), (51), and (52), we find that the decay widths for these processes are given by

$$\Gamma_{P_{1,2}^{d} \to d_{i}A_{D}} = \frac{g_{4}^{2}s_{P}^{2}(m_{P1,P2}^{d})^{3}}{48\pi m_{A_{D}}^{2}} \frac{|(\mathbf{W}_{d}^{\dagger}\vec{\gamma}_{P})_{i}|^{2}}{v_{P}^{2}}, \ i = 1, 2, 3,$$

$$\Gamma_{P_{1}^{u} \to u_{i}A_{D}} = \frac{g_{4}^{2}s_{P}^{2}(m_{P1}^{u})^{3}}{48\pi m_{A_{D}}^{2}} \frac{|(\vec{\gamma}_{P})_{i}|^{2}}{v_{P}^{2}}, \ i = 1, 2, 3,$$

$$\Gamma_{P_{2}^{u} \to u_{i}A_{D}} = \frac{g_{4}^{2}s_{P}^{2}(m_{P2}^{u})^{3}}{48\pi m_{A_{D}}^{2}} \frac{|(\vec{\gamma}_{P})_{i}|^{2}}{v_{P}^{2}}, \ i = 1, 2, 3,$$

$$\Gamma_{P_{2}^{u} \to u_{i}A_{D}} = \frac{g_{4}^{2}s_{P}^{2}(m_{P2}^{u})^{3}}{48\pi m_{A_{D}}^{2}} \frac{|(\vec{\gamma}_{P})_{i}|^{2}}{v_{P}^{2}}, \ i = 1, 2,$$

$$\Gamma_{P_{2}^{u} \to tA_{D}} = \frac{g_{4}^{2}s_{P}^{2}(m_{P2}^{u})^{3}}{48\pi m_{A_{D}}^{2}} \frac{|(\vec{\gamma}_{P})_{3}|^{2}}{v_{P}^{2}} \left(\frac{M_{t}^{2} - y_{S}^{2}v_{S}^{2}}{M_{t}^{2}}\right),$$
(80)

where $d_1 \equiv d$, $d_2 \equiv s$, $d_3 \equiv b$, analogous definitions hold for the up-like quarks, and we have neglected the effects of the top quark mass (which should only introduce $O(m_t^2/(m_{P1,2}^u)^2) \sim O(10^{-2})$ corrections). For clarity, we note that the enormous ratios $(m_{P1,P2}^{u,d})^2/(m_{A_D})^2$ in these expressions are cancelled by the comparably tiny ratios $|(\vec{\gamma}_P)_i|^2/v_P^2$, leading to results with reasonable magnitudes – this is in fact analogous to the discussion in Section 3.3 of **II** in which the dark photon coupling between SM and portal matter states is strongly suppressed, but the suppression is counteracted by the dark photon's small mass to give an O(1) overall strength for the interaction. In addition to these decay widths, we would expect equal (at least in the limit where $m_{A_D} \to 0$) contributions from a light scalar, corresponding to a physical scalar mode associated with $U(1)_D$ breaking, from Goldstone boson equivalence. As the scalar sector is not well-explored here, and such changes would not affect the branching fractions of the portal matter anyway, we shall ignore this scalar from here on out, with one exception discussed in Section 5.2. As discussed in **I**, assuming that the dark photon is long-lived or only decays to DM (which we will assume here), the signal for a portal matter pair production event here should consist of two (possibly fat) jets, which may be from t, b, or lighter quarks depending on $\vec{\gamma}_P$'s orientation in flavor space, plus a significant amount of missing energy stemming from the dark photon decays. Perhaps more interesting in this model are the possibilities afforded us by the existence of the heavy top partner: In addition to decays to SM particles, decays with comparable rates should exist for the portal matter fields into the top partner, provided the latter is lighter enough that these channels are kinematically accessible. We should note that, because the masses M_t , m_{P1}^u , and m_{P2}^u are simply free parameters in our theory with masses of at least O(TeV), we have little a priori guidance about whether the top partner is heavier than the portal fields or vice versa. It therefore behooves us to consider both the possibility that a portal matter field can decay into a top partner, and, in Section 5.2, whether the reverse process might occur in a different region of parameter space. Turning our attention back to the decay of portal matter to a top partner, up to $O(m_t^2/M_t^2)$ corrections and assuming that m_{A_D} and $\vec{\gamma}_P$ are much smaller than any other mass scale in the expression, we find that these decay widths are given by

$$\Gamma_{P_1^u \to TA_D} \approx \frac{g_4^2 s_P^2 (m_{P_1}^u)^3}{48\pi m_A^2} \left(1 - \frac{M_t^4}{(m_{P_1}^u)^4} \right) \frac{m_t^2}{y_S^2 v_S^2} \frac{|(\vec{\gamma}_P)_3|^2}{v_P^2}, \tag{81}$$

$$\Gamma_{P_2^u \to TA_D} \approx \frac{g_4^2 s_P^2 (m_{P_2}^u)^3}{48\pi m_A^2} \left(1 - \frac{M_t^4}{(m_{P_2}^u)^4} \right) \frac{y_S^2 v_S^2}{M_t^2} \frac{|(\vec{\gamma}_P)_3|^2}{v_P^2}.$$

The somewhat strange dependence of these decay widths on the fourth power of the ratio $M_t/m_{P1,2}^u$ bears a brief discussion here: It occurs because the phase space factor in the decay (which approaches $1 - M_t^2/(m_{P1,P2}^u)^2$ in the limit where $m_{A_D} \to 0$) multiplies the combination of coupling constants appearing in the squared amplitude, which are proportional to $1 + M_t^2/(m_{P1,P2}^u)^2$. Meanwhile, we also note that the decay of P_1^u to a top partner quark is suppressed by a factor of $m_t^2/(y_S^2v_S^2) \lesssim 10^{-2}$, and so is unlikely to have much phenomenological impact. However, the decay of P_2^u to the heavy top partner can be quite significant. The branching fraction for this decay should be given by

$$\mathcal{B}(P_2^u \to T A_D) = \frac{\left(1 - \frac{M_t^4}{(m_{P_2}^u)^4}\right) \frac{y_S^2 v_S^2}{M_t^2} |(\vec{\gamma}_P)_3|^2}{(\vec{\gamma}_P^* \cdot \vec{\gamma}_P) - \frac{M_t^4}{(m_{P_2}^u)^4} \frac{y_S^2 v_S^2}{M_t^2} |(\vec{\gamma}_P)_3|^2}.$$
(82)

Note that because $M_t < (m_{P2}^u)$, $y_S v_S < M_t$, and $|(\vec{\gamma}_P)_3|^2$ will always be less than $\vec{\gamma}_P^* \cdot \vec{\gamma}_P$ here, the above branching fraction will always be positive. To get a feel for the magnitude of this effect, we show this branching fraction as a function of the ratio M_t/m_{P1}^u in Figure 5 for some of the benchmarks of $\vec{\gamma}_P$ orientation outlined in Table 7. We note that the benchmarks in Table 7 list values of $\xi_i \equiv (\mathbf{W}_d^{\dagger}\vec{\gamma}_P)_i/v_P$, not $\vec{\gamma}_P$, however, the corrections due to letting $\xi_i \approx (\vec{\gamma}_P)_i/v_P$ to Eq.(82) will be suppressed by $O(\lambda^2) \sim 10^{-2}$ CKM factors and so are negligible for demonstrative purposes.

In Figure 5, we can see that when kinematically allowed, the branching fraction of P_2^u to the top partner can be quite significant. In cases where the portal matter mixes predominantly with the top quark, it can even account for nearly half of possible decays. As we shall see in Section 5.2, in this scenario the top partner will then decay via the emission of an SM Z, W, or Higgs into an SM top quark, reflecting the more conventional decay channels for vector-like quarks. This behavior in turn can yield interesting collider signals: Rather than simply producing a pair of jets plus missing energy, as will overwhelmingly occur for the other portal fields in this model, one of the P_2^u fields decaying via the top partner might result in, for example, missing energy, a pair of fat top quark jets, and a pair of leptons from a Z boson. This is an atypical signature.

5.2 Phenomenology: Top Quark Partner

Other than the portal fields, the top partner T is the sole other exotic fermion field that may be accessible at current or planned experiments in this model. Anticipating that the top partner here will behave similarly to conventional up-like vector-like quarks, we have quoted the constraint of [24], which places a 95% CL limit on the mass of a generic electroweak singlet vector-like quark with this electric charge as > 1.31 TeV. However, we note that this constraint assumes that the top partner will always decay via the standard processes $T \to ht$, $T \to Zt$, or $T \to W^+ b$. In our present construction, additional decay



Figure 5: (Top) The branching fraction of the P_2^u quark to the top partner, assuming that $M_t/(y_S v_S) = 1.5$ and that $\vec{\gamma}_P$ is oriented according to the benchmarks A_3 (blue), B_1 (red), and C_1 (magenta), described in Table 7. (Bottom) Same as above, but assuming that $M_t/(y_S v_S) = 2$.

channels may also be present, so it is useful to determine the extent to which this assumption is valid.

The interactions of this top partner with the SM closely follow the familiar behavior of vector-like quarks in other models; most notably, as we can see from the couplings in Section 3.2, the SM Z, W, and Higgs all mediate interactions between the top partner and the SM top, facilitating the $T \rightarrow Z t$, $T \rightarrow W^+ b$, and $T \rightarrow h t$ decay processes which are assumed in [24], and which are by contrast highly suppressed for the portal matter fields. Specifically, for these processes we have (up to $O(m_t^2/M_t^2) \sim 10^{-2}$ corrections)

$$\Gamma_{T \to Wb} = 2\Gamma_{T \to Zt} = 2\Gamma_{T \to ht} = \frac{G_F m_t^2}{\sqrt{2} \ 8\pi} \left(\frac{M_t^2 - y_S^2 v_S^2}{y_S^2 v_S^2}\right) M_t,\tag{83}$$

where $G_F = 1.1663787 \times 10^{-5} \text{ GeV}^{-2}$ is the Fermi constant. Notably, while there are *no* decays of this quark into other SM up-like quarks (at least up to highly-suppressed flavor-changing neutral currents from, for example, the dark photon), there are CKM-suppressed decays of the form $T \to W^+ d$ and $T \to W^+ s$; these are, however, suppressed by at least $\lambda^4 \sim O(10^{-3})$ compared to the $T \to W^+ b$ decay, and therefore have no observable experimental effect at present.

In the event that all portal matter fields in the model are more massive than T, the above exhausts the possible decay channels for the top partner, indicating that it strongly resembles a conventional vector-like quark. The scenario can become more complicated, however, when the up-like portal matter is light enough to allow for the top partner to decay into it. In this case, the widths for decays of T to P_1^u and P_2^u are

$$\Gamma_{T \to P_1^u A_D} = \frac{g_4^2 s_P^2 (m_{P1}^u)^2}{48\pi m_{A_D}^2} \left(1 - \frac{(m_{P1}^u)^4}{M_t^4} \right) \left(1 - \frac{y_S^2 v_S^2}{M_t^2} \right) \frac{m_t^2}{y_S^2 v_S^2} M_t \frac{|(\vec{\gamma}_P)_3|^2}{v_P^2}, \tag{84}$$

$$\Gamma_{T \to P_2^u A_D} = \frac{g_4^2 s_P^2 (m_{P2}^u)^2}{48\pi m_{A_D}^2} \left(1 - \frac{(m_{P2}^u)^4}{M_t^4} \right) \frac{y_S^2 v_S^2}{M_t^2} M_t \frac{|(\vec{\gamma}_P)_3|^2}{v_P^2}.$$

We see that, analogous to the case for the reversed decay pattern discussed in Section 5.1, the decay width from T to P_1^u is suppressed by a factor of $m_t^2/(y_S^2 v_S^2) \leq 10^{-2}$, and so is likely not numerically significant, while the decay width from T to P_2^u has no such suppression and may be large. Furthermore, we note (as discussed in Section 5.1), we would anticipate from Goldstone boson equivalence that a physical scalar, the 'dark' Higgs, which we shall call S_D , associated with the breaking of $U(1)_D$ will likely also exist, with a decay width such that $\Gamma_{T \to P_2^u} S_D = \Gamma_{T \to P_2^u} A_D$, at least in the limit where $m_{A_D} \to 0$. Unlike in the case of portal matter decays, for which each numerically significant decay channel has such an identical channel with S_D , and hence branching fractions are unaffected by ignoring the scalar, the top partner's dominant SM decays $T \to Z t$, $T \to h t$, and $T \to W^+ b$ do not have an equivalent S_D channel, while the decay to portal matter does. Therefore, when discussing the decay of the top partner to portal matter, we must include the decay width of the process $T \to P_2^u S_D$ explicitly, arriving at the combined branching fraction from Eqs.(83) and (84) of

$$\mathcal{B}(T \to P_2^u A_D, S_D) \approx \frac{2\Gamma_{T \to P_2^u A_D}}{4\Gamma_{T \to ht} + 2\Gamma_{T \to P_2^u A_D}}.$$
(85)

To get a sense of the possible magnitude of the branching fraction given in Eq.(85), we depict it for various choices of M_t and $M_t/(y_S v_S)$ in Figure 6, assuming that $|(\vec{\gamma}_P)_3|/v_P = 10^{-4}$, the "natural" magnitude we discussed in Section 4.1.

In Figure 6, we have selected values of $g_{4}s_{P}$ and $m_{A_{D}}$ so as to maximize the branching fraction to the portal matter fields, selecting $m_{A_{D}}$ at the bottom of the range we consider, while selecting $g_{4}s_{P}$ close to the upper end of the range that still permits $O(10^{-3})$ kinetic mixing (as discussed in Section 3.3). Given that the decay width of the top partner to m_{P1} is proportional to the square of $g_{4}s_{P}$, and the inverse square of $m_{A_{D}}$, we can anticipate that larger values of $m_{A_{D}}$ and smaller couplings $g_{4}s_{P}$ will result in dramatically smaller branching fractions – letting $m_{A_{D}} = 0.3$ GeV, for example, will reduce the branching fraction by a factor of 9. Even in the maximal case we depict in Figure 6, we generally only attain limited



Figure 6: (Top) The branching fraction of the top partner quark to P_2^u as a function of the ratio of their masses, assuming that $g_4s_P = 0.3$, $m_{A_D} = 0.1$ GeV, $|(\vec{\gamma}_P)_3|/v_P = 10^{-4}$, and $M_t/(y_Sv_S) = 1.5$, for $M_t = 2$ TeV (blue), 4 TeV (red), and 6 TeV (magenta). (Bottom) Same as above, but assuming that $M_t/(y_Sv_S) = 2$.

branching fractions of T to P_u^2 here, however – we achieve a ~ 30% branching fraction at the largest when $M_t = 6$ TeV, half that when $M_t = 4$ TeV, and in all other cases we depict, the branching fraction is less than 10%. This branching fraction can be enhanced if we assume a larger value of $|(\vec{\gamma}_P)_3|/v_P$ (for example, enhancing this parameter by a factor of 3 would increase the decay width of T to portal matter by a factor of 9), however, considering a value of $|(\vec{\gamma}_P)_3|/v_P$ much larger than 10^{-4} , as we have taken it to be in Figure 6, begins to be in tension with constraints on flavor-changing B meson decays discussed in Section 4.1. In short, in contrast to the scenario with the possible decay of portal matter to the top partner the reverse scenario is likely to have a minimal effect on the signal of top partner production, except in certain special corners of parameter space. In these corners, however, we might anticipate some interesting potential top partner events – for example, top partner pair production might give us one 'conventional' top partner which decays via the channels $T \to Z t$, $T \to h t$, or $T \to W b$, and the other that decays to a top (or, depending on the orientation of $\vec{\gamma}_P$ in flavor space, a lighter up-like quark) through sequentially emitting two dark photons, which would appear as missing energy. Determining the degree to which the latter decay path might be kinematically distinguishable from, for example, $T \to Z t$ with an invisibly decaying $Z \to \bar{\nu}\nu$ may be of interest, but lies beyond the scope of this work.

5.3 Phenomenology: Z_P Production

To start, we shall consider the production of the "portal Z" boson Z_P , representing the heavy counterpart to the lighter dark photon, with a mass of O(1 TeV). Notably, since the other gauge bosons with which Z_P might couple (specifically, those associated with $SU(4)_F$ generators other than t^{15} , so in principle any gauge boson except A_D) are all much heavier than Z_P itself, there is no region of parameter space in which Z_P might decay into any other gauge bosons at tree-level. Our sole concern at this point, then, would be the decay of the Z_P into fermion pairs. Consulting the coupling matrix for Z_P given in Eqs.(53) and (54), we can then straightforwardly determine the decay widths of the Z_P boson. In particular, for decay into any particular SM fermion, we have

$$\Gamma_{Z_P \to \overline{f}f} \approx C_f \frac{g_4^2/c_P^2}{288\pi} m_{Z_P} \left[1 + O\left(\frac{m_f^2}{m_{Z_P}^2}\right) \right],\tag{86}$$

where C_f is simply a color factor (3 for quarks, 1 for charged leptons and neutrinos). Depending on the relative masses of the top partner and the portal matter fields to m_{Z_P} , however, Z_P may also decay into these states. For the top partner, the decay width is

$$\Gamma_{Z_P \to \overline{T}T} = \frac{g_4^2/c_P^2}{96\pi} m_{Z_P} \left(1 + \frac{2M_t^2}{m_{Z_P}^2} \right) \sqrt{1 - \frac{4M_t^2}{m_{Z_P}^2}}.$$
(87)

The phenomenologically relevant portal matter fields, meanwhile, yield decay processes with widths

$$\Gamma_{Z_P \to \overline{P}_i^u P_i^u} = 3 \; \frac{g_4^2 / c_P^2}{64\pi} m_{Z_P} \left[\left(1 + \frac{8}{9} x_P \right) \left(1 - \frac{(m_{Pi}^u)^2}{m_{Z_P}^2} \right) + \frac{8}{3} x_P \frac{(m_{Pi}^u)^2}{m_{Z_P}^2} \right] \sqrt{1 - \frac{4(m_{Pi}^u)^2}{m_{Z_P}^2}}, \quad (88)$$
$$x_P \equiv s_P^2 (1 - 4c_P^2),$$

where P_i^u refers to the portal matter field of mass $m_{P_i}^u$, and analogous expressions exist for the portal matter in the down-like quark, charged lepton, and neutrino sectors (note that the factor of 3 in front of Eq.(88) is a color factor, and will hence be equal to one for the corresponding case of neutrino and charged lepton portal matter). Because the couplings of the Z_P to SM states are flavor-universal, this model makes easy contact with existing searches for high-mass spin-1 resonances. To estimate constraints on our model, we employ the limits on leptonic cross sections from [46], a dilepton (dielectron and dimuon)-channel null search using 139 fb⁻¹ of 13 TeV data, and the analogous expected limits from a null result with 3000 fb⁻¹ of LHC data at 14 TeV [47]. In Figure 7, we depict the maximum value of g_4/c_P allowed as a function of m_{Z_P} and the corresponding minimum allowed value of v_P , both for current experimental limits and those expected from a null result at the High-Luminosity LHC, assuming that the Z_P boson only decays



into SM final states (in which case the branching ratios for the dielectron or dimuon channels are simply 1/24).

Figure 7: (Top) The maximum allowed value of g_4/c_P (left) and the minimum allowed v_P (right) for different values of m_{Z_P} , based on limits from dilepton-channel searches with 139 fb⁻¹ of 13 TeV LHC data [46], assuming Z_P only decays into SM final states. For comparison, the electroweak coupling g/c_w (left) and the value of v_P assuming $g_4/c_P = g/c_w$ for a given m_{Z_P} value (right) have been depicted as dashed orange lines. (Bottom) Same as above, but using the expected constraints from a null result with 3 ab⁻¹ of 14 TeV LHC data [47]

Examining Figure 7, we note that for realistic O(1) values of g_4/c_P (or, equivalently, O(10 TeV) values of v_P), the present constraint roughly requires $m_{Z_P} \gtrsim 5$ TeV; if we assume that $g_4/c_P = g/c_w$, where g is the $SU(2)_L$ coupling constant and c_w is the cosine of the Weinberg angle, the limit is, for example, $m_{Z_P} \geq 5.15$ TeV. A null result at the HL-LHC meanwhile suggests that these limits could be increased by approximately 1 TeV: If $g_4/c_P = g/c_w$, the expected limit from a null result of HL-LHC data would be $m_{Z_P} \geq 6.2$ TeV. However, we note that similar to the setup in **II**, this circumstance can be somewhat modified in regions of parameter space in which decays to exotic fermion species are kinematically allowed. In particular, constraints may be somewhat relaxed by assuming that the portal matter fields, $P_{1,2}^{u,d,e,\nu}$ are sufficiently light to allow for Z_P to decay into them (which, assuming $g_4/c_P \sim O(1)$, would require at worst a mild $O(10^{-1})$ tuning of the Yukawa coupling parameters y_{P1} and y_{P2}). While the decay into the top partner can reduce the branching fraction of the Z_P to dilepton channels by approximately 10%, the large number of portal matter states with roughly degenerate masses (if we assume that decays to all of these portal matter states are kinematically accessible, for example, Z_P would have 8 different new particles, some color triplets, to which it can decay) can provide as much as an order of magnitude reduction in the cross section observable in dilepton final state searches. We depict the proportional suppression factor of the branching ratio $\mathcal{B}(Z_P \to l^+ l^-)$ assuming that some or all portal matter states are kinematically accessible in Figure 8, where we note that, as Eq.(88) suggests, the suppression factor is a function of c_P .

In Figure 8, we have specifically considered two sets of simple benchmarks for the possibility that the portal matter states are kinematically accessible in Z_P decays – either that all portal matter states have the same mass (which, we note, would not be an unreasonable approximation if $y_{P1} \approx y_{P2}$, in which case all portal matter states would have degenerate masses up to radiative corrections), or that only half of the portal matter states are light enough for Z_P to decay into them (as we might expect if, for example, $y_{P1} < y_{P2}$). The suppression factors in Figure 8, in turn, correspond to as much as O(1) increases in the maximum allowed g_4/c_P values (or equivalently, O(1) decreases in the minimum v_P values) depicted in Figure 7. We note, however, that even under the best of circumstances (for example, if the suppression factor is ~ 0.2, very near the lowest we can achieve), the g_4/c_P values are only increased by a factor of ~ 2 . A cursory inspection of Figure 7 indicates that this amelioration is enough to perhaps allow $m_{Z_P} \sim 4$ TeV while maintaining a reasonable O(1) value of g_4/c_P (at least based on 13 TeV data), but certainly cannot enable much smaller m_{Z_P} values. Of course, the potential ameliorating effects of these exotic decay channels on dilepton searches may be counteracted by new distinctive signals due to the exotic matter production. In particular, the production of particle-antiparticle pairs of portal matter or top partner quarks via a resonant Z_P might have a significant effect on the overall production cross section of these states, especially in the case of leptonic portal matter, which otherwise may only be produced via electroweak and dark photon interactions. Null results for portal matter and top partner searches, then, will likely apply a non-trivial constraint on the possible mass of the Z_P as well, however treating this situation quantitatively is beyond the scope of this paper.

6 Discussion and Conclusions

The distinctive requirements for fermionic portal matter fields in vector portal DM models provide an intriguing framework for UV extensions of these models. In particular, constraints from SM anomalies and lifetimes of new SM-charged particles suggest that these fields are vector-like, albeit with collider signatures that are quite different from those seen with more "conventional" vector-like fermions. Here, we have seen that highly complex additional physics, not a priori related to DM, can in fact be incorporated into a portal matter model by following simple rules: Extending the SM gauge group to a semisimple one guarantees that KM with a $U(1)_D$ embedded in a separate dark group will be finite and calculable, at which point the only requirements are to ensure that the complete theory is anomaly-free and contains only the SM and heavy vector-like particles.

To demonstrate the potential of following this recipe, we have developed a model with Pati-Salam symmetry extended by $SU(4)_F \times U(1)_F = 4_F \mathbf{1}_F$, in which the $4_F \mathbf{1}_F$ symmetry contains both a dark photon gauge group $U(1)_D$ and an SU(3) flavor symmetry. The Pati-Salam symmetry is assumed to be broken at a high scale ~ 10^{13} GeV, at which point KM between $U(1)_F$ and $U(1)_Y$ occurs. Scalars in the adjoint representation of $SU(4)_F$ then break the $4_F \mathbf{1}_F$ down to $\mathbf{1}_{F'} \mathbf{1}_F$, completely breaking the embedded $SU(3)_F$ flavor symmetry, at multi-TeV scales to avoid flavor constraints and generate the fermion mass hierarchy observed in the SM. The $\mathbf{1}_{F'} \mathbf{1}_F$ symmetry is then broken down to $U(1)_D$ at a scale of ~ 1 - 10 TeV, and $U(1)_D$, the gauge symmetry corresponding to the DM vector portal, is finally broken by small vev terms of O(0.1 - 1 GeV). The presence of diverse scales in the symmetry breaking in turn translates to diverse scales of the new exotic particles proposed. However, we find that among the new fermions, only certain portal matter fields, which we have labelled $P_{1,2}^{u,d,e,\nu}$, and a vector-like new partner to the top quark, which we call T, can be expected to be light enough to be observed at present collider experiments, having masses at the scale of ~ a few TeV. Among the new gauge bosons



Figure 8: (Top) The proportional suppression of the branching ratio of Z_P to leptons, compared to the scenario where Z_P only decays to SM states, as a function of c_P , assuming portal matter states have the same mass m_P , assuming $m_P/m_{Z_P} = 0.2$ (blue), 0.3 (red), and 0.4 (magenta). (Bottom) Same as above, but assuming that only one of the two lighter portal matter states in each sector, rather than both, are kinematically accessible for this decay.

we have anticipate, we find that other than the dark photon A_D , only a heavy TeV-scale boson which has flavor-universal couplings, which we have dubbed the "portal Z", Z_P , and a flavor-changing neutral boson Z_F associated with a combination of generators of the flavor group $SU(3)_F$, may potentially have effects at observable scales.

With the model set up, we proceeded to explore the phenomenological implications of the experimentally observable (that is, TeV-scale or less) sector of the theory, in particular focusing on the phenomenology of quarks and their portal matter partners, which we expect to have more phenomenologically visible signals than the corresponding physics in the lepton sector. We found that present constraints from flavor-changing neutral currents arising from the dark photon, in particular from $K \to \pi A_D$ transitions, placed harsh limits on the form of the vev parameters which might break $U(1)_D$, and in turn can substantially influence the branching fractions of portal matter fields to SM particles. We next noted that the vector-like top partner T and the new vector boson Z_P contributed non-trivially to $\Delta B = 2$ flavorchanging processes, the Z_P at tree-level and the T at the one-loop level. These effects were comparable in magnitude and heavily dependent on the value of the model parameter $y_S v_S$, a dimensionful quantity which set the scale of both T and Z_F 's mass. However, even when combined the sum of these contributions still provided only a limited constraint on our parameter space based on present measurements, although Belle II and LHCb data may constrain these more stringently in the near future [27].

Beyond the rich collider phenomenology already explored in **I** and **II**, much of which is replicated in our model, we find additional model-building flexibility arising from the potential for the portal matter fields P_2^u to decay into T (or vice versa, depending on the particles' relative masses), which can provide distinctive collider signatures appearing in neither the treatments discussed in **I** and **II** nor more conventional models of vector-like quarks [12, 30, 43].

Looking forward, we note that the present model represents a single specific realization of a much broader recipe for developing models with non-minimal portal matter sectors. There are a number of different avenues through which this particular effort can be further explored, for example by incorporating an explanation for the small masses and near-maximal mixing observed in the neutrino sector, addressing the scalar sector rigorously, considering scenarios in which the Pati-Salam symmetry (or some components of it) are broken at a similar or lower scale to the one at which the $SU(4)_F \times U(1)_F$ symmetries are broken, or explicitly incorporating either scalar or fermionic DM in the construction. Perhaps a broader conclusion to be drawn from this work, however, is the potential that portal matter model building possesses to elaborate on other extensions of the SM. In particular, we have seen that incorporating a local SU(3) flavor symmetry with a model of portal matter leads to rich phenomenological signatures, some of which do not appear in more conventional portal matter models or models with local flavor symmetries individually, such as a potential correlation between portal matter lifetime, branching fractions, and the allowed dark photon masses, or decays of portal matter to more conventional vector-like quarks. Presumably, similar paths can be taken to explore potential links between portal matter models and other popular extensions of the SM, predicting entirely different unique signatures that may depart radically from those anticipated by conventional new physics searches.

Acknowledgements

GNW would like to thank T.D. Reuter for discussions relating to this work. This work was supported by the Department of Energy, Contract DE-AC02-76SF00515.

A SU(4) Generator Matrices

Here we list the generators of the SU(4) algebra, t^i . The first eight generators correspond to the embedded group $SU(3)_F$ in $SU(4)_F$, and are given by

The next 6 generators are then

Finally, t^{15} is the generator corresponding to the embedded group $U(1)'_F$ in $SU(4)_F$, which mixes with $U(1)_F$ to form the dark charge group. It is given by

$$t^{15} = \frac{1}{2\sqrt{6}} \begin{pmatrix} 1 & 0 & 0 & 0\\ 0 & 1 & 0 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & -3 \end{pmatrix}.$$
 (92)

References

- M. Kawasaki and K. Nakayama, Ann. Rev. Nucl. Part. Sci. 63, 69-95 (2013) [arXiv:1301.1123 [hep-ph]].
- [2] P. W. Graham, I. G. Irastorza, S. K. Lamoreaux, A. Lindner and K. A. van Bibber, Ann. Rev. Nucl. Part. Sci. 65, 485-514 (2015) [arXiv:1602.00039 [hep-ex]].

- [3] G. Arcadi, M. Dutra, P. Ghosh, M. Lindner, Y. Mambrini, M. Pierre, S. Profumo and F. S. Queiroz, Eur. Phys. J. C 78, no.3, 203 (2018) [arXiv:1703.07364 [hep-ph]].
- [4] M. Battaglieri, A. Belloni, A. Chou, P. Cushman, B. Echenard, R. Essig, J. Estrada, J. L. Feng, B. Flaugher and P. J. Fox, et al. [arXiv:1707.04591 [hep-ph]].
- [5] J. Alexander, M. Battaglieri, B. Echenard, R. Essig, M. Graham, E. Izaguirre, J. Jaros, G. Krnjaic, J. Mardon and D. Morrissey, et al. [arXiv:1608.08632 [hep-ph]].
- [6] There has been a huge amount of work on this subject; see, for example, D. Feldman, B. Kors and P. Nath, Phys. Rev. D 75, 023503 (2007) [arXiv:hep-ph/0610133 [hep-ph]]; D. Feldman, Z. Liu and P. Nath, Phys. Rev. D 75, 115001 (2007) [arXiv:hep-ph/0702123 [hep-ph]]; M. Pospelov, A. Ritz and M. B. Voloshin, Phys. Lett. B 662, 53-61 (2008) [arXiv:0711.4866 [hep-ph]]; M. Pospelov, Phys. Rev. D 80, 095002 (2009) [arXiv:0811.1030 [hep-ph]]; H. Davoudiasl, H. S. Lee and W. J. Marciano, Phys. Rev. Lett. 109, 031802 (2012) [arXiv:1205.2709 [hep-ph]]; H. Davoudiasl, H. S. Lee and W. J. Marciano, Phys. Rev. Lett. 109, 031802 (2012) [arXiv:1203.2947 [hep-ph]]; R. Essig, J. A. Jaros, W. Wester, P. Hansson Adrian, S. Andreas, T. Averett, O. Baker, B. Batell, M. Battaglieri and J. Beacham, et al. [arXiv:1311.0029 [hep-ph]]; E. Izaguirre, G. Krnjaic, P. Schuster and N. Toro, Phys. Rev. Lett. 115, no.25, 251301 (2015) [arXiv:1505.00011 [hep-ph]]; M. Khlopov, Int. J. Mod. Phys. A 28, 1330042 (2013) [arXiv:1311.2468 [astro-ph.CO]]; For a general overview and introduction to this framework, see D. Curtin, R. Essig, S. Gori and J. Shelton, JHEP 02, 157 (2015) doi:10.1007/JHEP02(2015)157 [arXiv:1412.0018 [hep-ph]].
- B. Holdom, Phys. Lett. 166B, 196 (1986) and Phys. Lett. B 178, 65 (1986); K. R. Dienes, C. F. Kolda and J. March-Russell, Nucl. Phys. B 492, 104 (1997) [hep-ph/9610479]; F. Del Aguila, Acta Phys. Polon. B 25, 1317 (1994) [hep-ph/9404323]; K. S. Babu, C. F. Kolda and J. March-Russell, Phys. Rev. D 54, 4635 (1996) [hep-ph/9603212]; T. G. Rizzo, Phys. Rev. D 59, 015020 (1998) [hep-ph/9806397].
- [8] E. Izaguirre, G. Krnjaic, P. Schuster and N. Toro, Phys. Rev. D 88, 114015 (2013) doi:10.1103/PhysRevD.88.114015 [arXiv:1307.6554 [hep-ph]].
- [9] T. Gherghetta, J. Kersten, K. Olive and M. Pospelov, Phys. Rev. D 100, no.9, 095001 (2019) doi:10.1103/PhysRevD.100.095001 [arXiv:1909.00696 [hep-ph]].
- [10] T. G. Rizzo, Phys. Rev. D 99, no.11, 115024 (2019) [arXiv:1810.07531 [hep-ph]].
- [11] T. D. Rueter and T. G. Rizzo, Phys. Rev. D 101, no.1, 015014 (2020) [arXiv:1909.09160 [hep-ph]].
- [12] For a recent overview of vector-like quarks and original references, see C. Y. Chen, S. Dawson and E. Furlan, Phys. Rev. D 96, no.1, 015006 (2017) [arXiv:1703.06134 [hep-ph]]; for a corresponding recent overview of vector-like leptons and original references, see Z. Poh and S. Raby, Phys. Rev. D 96, no.1, 015032 (2017) [arXiv:1705.07007 [hep-ph]]. For a general recent review of VLF, see V. Peralta, [arXiv:1712.06193 [hep-ph]].
- [13] See, however, some earlier general discussions in B. Patt and F. Wilczek, [arXiv:hep-ph/0605188 [hep-ph]]; H. Davoudiasl, R. Kitano, T. Li and H. Murayama, Phys. Lett. B 609, 117-123 (2005) [arXiv:hep-ph/0405097 [hep-ph]]; D. McKeen, M. Pospelov and A. Ritz, Phys. Rev. D 86, 113004 (2012) [arXiv:1208.4597 [hep-ph]]; B. Batell, M. Pospelov, A. Ritz and Y. Shang, Phys. Rev. D 81, 075004 (2010) [arXiv:0910.1567 [hep-ph]]; J. H. Kim, S. D. Lane, H. S. Lee, I. M. Lewis and M. Sullivan, Phys. Rev. D 101, no.3, 035041 (2020) [arXiv:1904.05893 [hep-ph]].
- [14] J. C. Pati and A. Salam, Phys. Rev. D 10, 275-289 (1974) [erratum: Phys. Rev. D 11, 703-703 (1975)] doi:10.1103/PhysRevD.10.275
- [15] S. S. Bao, Z. Liu and Y. L. Wu, Nucl. Phys. B 904, 1-17 (2016) [arXiv:1505.03583 [hep-ph]].
- [16] S. Saad, Nucl. Phys. B 943, 114630 (2019) doi:10.1016/j.nuclphysb.2019.114630 [arXiv:1712.04880 [hep-ph]].

- [17] A. H. Chamseddine, A. Connes and W. D. van Suijlekom, JHEP 11, 011 (2015) doi:10.1007/JHEP11(2015)011 [arXiv:1507.08161 [hep-ph]].
- [18] F. Hartmann, W. Kilian and K. Schnitter, JHEP 05, 064 (2014) [arXiv:1401.7891 [hep-ph]].
- [19] F. Sannino, [arXiv:1511.09022 [hep-ph]].
- [20] A. Maiezza, G. Senjanović and J. C. Vasquez, Phys. Rev. D 95, no.9, 095004 (2017) doi:10.1103/PhysRevD.95.095004 [arXiv:1612.09146 [hep-ph]].
- [21] A. Maiezza, M. Nemevsek, F. Nesti and G. Senjanovic, Phys. Rev. D 82, 055022 (2010) [arXiv:1005.5160 [hep-ph]].
- [22] J. F. Gunion, J. Grifols, A. Mendez, B. Kayser and F. I. Olness, Phys. Rev. D 40, 1546 (1989) doi:10.1103/PhysRevD.40.1546
- [23] M. S. Chanowitz and M. K. Gaillard, Nucl. Phys. B 261, 379 (1985); B. W. Lee, C. Quigg and H. B. Thacker, Phys. Rev. D 16, 1519 (1977); J. M. Cornwall, D. N. Levin and G. Tiktopoulos, Phis. Rev. D 10, 1145 (1974) Erratum: [Phys. Rev. D 11, 972 (1975)]; G. J. Gounaris, R. Kogerler and H. Neufeld, Phys. Rev. D 34, 3257 (1986).
- [24] M. Aaboud *et al.* [ATLAS], Phys. Rev. Lett. **121**, no.21, 211801 (2018) [arXiv:1808.02343 [hep-ex]].
- [25] C. Alpigiani, A. Bevan, M. Bona, M. Ciuchini, D. Derkach, E. Franco, V. Lubicz, G. Martinelli, F. Parodi and M. Pierini, et al. [arXiv:1710.09644 [hep-ph]].
- [26] M. Blanke, CERN Yellow Rep. School Proc. 1705, 71-100 (2017) doi:10.23730/CYRSP-2017-005.71 [arXiv:1704.03753 [hep-ph]].
- [27] J. Charles, S. Descotes-Genon, Z. Ligeti, S. Monteil, M. Papucci, K. Trabelsi and L. Vale Silva, Phys. Rev. D 102, no.5, 056023 (2020) [arXiv:2006.04824 [hep-ph]].
- [28] R. Röntsch and M. Schulze, JHEP 07, 091 (2014) [erratum: JHEP 09, 132 (2015)] [arXiv:1404.1005 [hep-ph]].
- [29] M. Ciuchini, E. Franco, S. Mishima, M. Pierini, L. Reina and L. Silvestrini, Nucl. Part. Phys. Proc. 273-275, 2219-2225 (2016) [arXiv:1410.6940 [hep-ph]].
- [30] A. De Simone, O. Matsedonskyi, R. Rattazzi and A. Wulzer, JHEP 04, 004 (2013) [arXiv:1211.5663 [hep-ph]].
- [31] P.A. Zyla et al. [Particle Data Group], PTEP 2020, no.8, 083C01 (2020) doi:10.1093/ptep/ptaa104
- [32] A. K. Perrevoort [Mu3e], SciPost Phys. Proc. 1, 052 (2019) [arXiv:1812.00741 [hep-ex]].
- [33] A. V. Artamonov et al. [BNL-E949], Phys. Rev. D 79, 092004 (2009) [arXiv:0903.0030 [hep-ex]].
- [34] K. Fuyuto, W. S. Hou and M. Kohda, Phys. Rev. Lett. 114, 171802 (2015) [arXiv:1412.4397 [hep-ph]].
- [35] J. K. Ahn et al. [KOTO], Phys. Rev. Lett. 122, no.2, 021802 (2019) [arXiv:1810.09655 [hep-ex]].
- [36] J. Grygier et al. [Belle], Phys. Rev. D 96, no.9, 091101 (2017) [arXiv:1702.03224 [hep-ex]].
- [37] J. P. Lees et al. [BaBar], Phys. Rev. D 87, no.11, 112005 (2013) [arXiv:1303.7465 [hep-ex]].
- [38] H. Leutwyler and M. Roos, Z. Phys. C. 25, 91-101 (1984)
- [39] P. Ball and R. Zwicky, Phys. Rev. D 71, 014015 (2005) [arXiv:hep-ph/0406232 [hep-ph]].
- [40] J. Charles, S. Descotes-Genon, Z. Ligeti, S. Monteil, M. Papucci and K. Trabelsi, Phys. Rev. D 89, no.3, 033016 (2014) [arXiv:1309.2293 [hep-ph]].

- [41] A. Hocker and Z. Ligeti, Ann. Rev. Nucl. Part. Sci. 56, 501-567 (2006) [arXiv:hep-ph/0605217 [hep-ph]].
- [42] A. J. Buras, W. Slominski and H. Steger, Nucl. Phys. B 245, 369-398 (1984)
- [43] D. Vatsyayan and A. Kundu, Nucl. Phys. B 960, 115208 (2020) [arXiv:2007.02327 [hep-ph]].
- [44] A. Bazavov et al. [Fermilab Lattice and MILC], Phys. Rev. D 93, no.11, 113016 (2016) [arXiv:1602.03560 [hep-lat]].
- [45] J. A. Bagger, K. T. Matchev and R. J. Zhang, Phys. Lett. B 412, 77-85 (1997) [arXiv:hep-ph/9707225 [hep-ph]].
- [46] G. Aad et al. [ATLAS], Phys. Lett. B 796, 68-87 (2019) [arXiv:1903.06248 [hep-ex]].
- $[47]\ [ATLAS],\ ATL-PHYS-PUB-2018-044.$